PREFACE March 26, 2019

In response to a request under the California Public Records Act, the City of Vallejo is exercising its discretion to make public the unfinished draft Final Environmental Impact Report (FEIR) for the VMT/Orcem project. The current version of this document is not ready for certification under the purposes of California Environmental Quality Act (CEQA). As of this date, clarification is needed as to who is the responsible party for certain indemnity and mitigation measures, and who has site control and ownership of the project site. While this clarification is obtained processing of the EIR has been paused.

As of March 26, 2019, the City, as lead agency, has determined that the VMT/Orcem project is not yet ready for approval and that the environmental documents that have been prepared do not yet achieve a compliance with CEQA (Cal. Code Regs, Titl. 14 Section 15090(a)(1)) Thus, the FEIR is not ready to be presented to the City Council for certification and project approval under CEQA (Cal. Code Regs, Titl. 14 Section 15090(a)(2)).

While the processing of the application has been paused, staff will endeavor to work with applicants to obtain an updated environmental justice analysis, and data, as well as commitments from the applicants to perform certain mitigation measure in order to present them as feasible. Feasible in this context means capable of being accomplished in a successful manner within a reasonable period of time, taking into account economic, environmental, legal, social, and technological factors (Cal. Code Regs, Titl.14 Section 15364).

Recognizing that the applicants and the public have requested release of the draft FEIR for public viewing, the City is accommodating these requests by posting the documents here. A progress report on this project is expected to be presented to Council by April 23, 2019.

DRAFT FEIR

FINAL

VALLEJO MARINE TERMINAL AND ORCEM PROJECT ENVIRONMENTAL IMPACT REPORT

Prepared for:

City of Vallejo

555 Santa Clara Street Vallejo, California 94590

Prepared by:

DUDEK

1630 San Pablo Ave, Suite 300 Oakland, CA 94612 Contact: Darcey Rosenblatt

FEBRUARY 2019

Printed on 30% post-consumer recycled material.

TABLE OF CONTENTS

Section

Page No.

ACRO	NYMS	AND ABBREVIATIONSACF	≀-I
EXEC	UTIVE	E SUMMARY ES	5-1
	ES.1	IntroductionES	5-1
	ES.2	Project Location	5-1
	ES.3	Existing Project SiteES	5-2
	ES.4	Project Overview ES	5-2
	ES.5	Project Objectives ES	5-4
	ES.6	Summary of Impacts ES	5-5
	ES.7	Analysis of Alternatives	41
		ES.7.1 Alternatives AnalyzedES-	41
		ES.7.2 Environmentally Superior Alternative	42
	ES.8	Areas of Controversy ES-	42
	ES.9	Issues to be Resolved by Lead Agency ES-	43
1	INTRO	DDUCTION1	1-1
	1.1	Background 1	-1
		1.1.1 The VMT Component of the Project 1	-2
		1.1.2 The Orcem Component of the Project 1	-2
	1.2	Project Purpose and Need 1	-2
	1.3	Purpose of the EIR 1	-2
	1.4	Intended Uses of the EIR 1	-3
	1.5	Scope of the EIR	-4
	1.6	CEQA Process	-5
		1.6.1 Lead and Responsible Agencies	-5
		1.6.2 Notice of Preparation and Responses1	-6
		1.6.3 Draft EIR Public Review	-7
		1.6.4 Draft Final EIR 1	-7
		1.6.5 Final EIR	-7
	1.7	Document Organization	-8
2	PROJ	ECT DESCRIPTION 2	2-1
	2.1	Project Location	2-1
	2.2	Existing Project Site	2-1
	2.3	Project Objectives	2-3
	2.4	Proposed Project	2-4
		2.4.1 Construction	2-7
		2.4.2 Operation2-	11
		2.4.3 Infrastructure	23

		2.4.4	Off-Site Public Access Improvements	
		2.4.5	Optional Development Agreement and/or Community Benefits	s Agreement
3	ENV	IRONM	IENTAL ANALYSIS	
	3.1	Aesthe	etics	
		3.1.1	Regulatory Setting	
		3.1.2	Existing Conditions	
		3.1.3	Thresholds of Significance	
		3.1.4	Impact Discussion	
		3.1.5	Mitigation Measures	
		3.1.6	Level of Significance After Mitigation	
	3.2	Air Qu	uality	
		3.2.1	Regulatory Setting	
		3.2.2	Existing Conditions	3.2-11
		3.2.3	Thresholds of Significance	3.2-14
		3.2.4	Impact Discussion	3.2-16
		3.2.5	Mitigation Measures	3.2-42
		3.2.6	Level of Significance After Mitigation	3.2-45
	3.3	Biolog	gical Resources	
		3.3.1	Regulatory Setting	
		3.3.2	Existing Conditions	3.3-14
		3.3.3	Thresholds of Significance	3.3-38
		3.3.4	Impact Discussion	3.3-39
		3.3.5	Mitigation Measures	3.3-66
		3.3.6	Level of Significance after Mitigation	3.3-75
	3.4	Cultur	ral Resources	3.4-1
		3.4.1	Regulatory Setting	3.4-1
		3.4.2	Existing Conditions	
		3.4.3	Thresholds of Significance	3.4-15
		3.4.4	Impact Discussion	3.4-15
		3.4.5	Mitigation Measures	3.4-19
		3.4.6	Level of Significance After Mitigation	3.4-24
	3.5	Geolo	gy and Soils	
		3.5.1	Regulatory Setting	
		3.5.2	Existing Conditions	
		3.5.3	Thresholds of Significance	3.5-12
		3.5.4	Impact Discussion	3.5-13
		3.5.5	Mitigation Measures	3.5-18
		3.5.6	Level of Significance After Mitigation	3.5-19

3.6	Green	house Gas Emissions	
	3.6.1	Regulatory Setting	
	3.6.2	Existing Conditions	
	3.6.3	Thresholds of Significance	
	3.6.4	Impact Discussion	
	3.6.5	Mitigation Measures	
	3.6.6	Level of Significance After Mitigation	
3.7	Hazar	ds and Hazardous Materials	
	3.7.1	Regulatory Setting	
	3.7.2	Existing Conditions	
	3.7.3	Thresholds of Significance	
	3.7.4	Impact Discussion	
	3.7.5	Mitigation Measures	
	3.7.6	Level of Significance After Mitigation	
3.8	Hydro	logy and Water Quality	
	3.8.1	Regulatory Setting	
	3.8.2	Existing Conditions	
	3.8.3	Thresholds of Significance	
	3.8.4	Impact Discussion	
	3.8.5	Mitigation Measures	
	3.8.6	Level of Significance After Mitigation	
3.9	Land	Use and Planning	
	3.9.1	Regulatory Setting	
	3.9.2	Existing Conditions	
	3.9.3	Thresholds of Significance	
	3.9.4	Impact Discussion	
	3.9.5	Mitigation Measures	
	3.9.6	Level of Significance After Mitigation	
3.10	Noise		
	3.10.1	Regulatory Setting	
	3.10.2	Existing Conditions	
	3.10.3	Thresholds of Significance	
	3.10.4	Impact Discussion	
	3.10.5	Mitigation Measures	
	3.10.6	Level of Significance After Mitigation	
3.11	Public	Services and Recreation	
	3.11.1	Regulatory Setting	
	3.11.2	Existing Conditions	
	3.11.3	Thresholds of Significance	

		3.11.4 Impact Discussion	3.11-7
		3.11.5 Mitigation Measures	3.11-8
		3.11.6 Level of Significance After Mitigation	3.11-8
	3.12	Transportation and Traffic	3.12-1
		3.12.1 Regulatory Setting	3.12-1
		3.12.2 Existing Conditions	3.12-4
		3.12.3 Thresholds of Significance	3.12-16
		3.12.4 Impact Discussion	3.12-18
		3.12.5 Mitigation Measures	3.12-34
		3.12.6 Level of Significance After Mitigation	3.12-38
	3.13	Utilities and Service Systems	3.13-1
		3.13.1 Regulatory Setting	3.13-1
		3.13.2 Existing Conditions	3.13-7
		3.13.3 Thresholds of Significance	3.13-9
		3.13.4 Impact Discussion	3.13-10
		3.13.5 Mitigation Measures	3.13-17
		3.13.6 Level of Significance After Mitigation	3.13-17
4	CUM	ULATIVE IMPACTS	
	4.1	Introduction	
	4.2	Methodology	
		4.2.1 Cumulative Projects List	
	4.3	Cumulative Impact Analysis	
		4.3.1 Aesthetics	
		4.3.2 Air Quality	
		4.3.3 Biological Resources	
		4.3.4 Cultural Resources	
		4.3.5 Geology and Soils	
		4.3.6 Greenhouse Gas Emissions	
		4.3.7 Hazards and Hazardous Materials	
		4.3.8 Hydrology and Water Quality	
		4.3.9 Land Use and Planning	
		4.3.10 Noise	
		4.3.11 Public Services and Recreation	
		4.3.12 Transportation and Traffic	
		4.3.13 Utilities and Service Systems	
5	OTH	ER CEQA CONSIDERATIONS	
	5.1	Effects Not Found to be Significant	
	5.2	Significant and Unavoidable Environmental Impacts	

	5.3	Signifi	cant and Irreversible Environmental Effects	
	5.4	Growt	h Inducement	
6	ALT	ERNAT	IVES	
	6.1	Introdu	action	6-1
	6.2	Projec	t Objectives	
	6.3	Altern	atives Considered But Rejected	
		6.3.1	Alternate Site	
		6.3.2	Preservation Alternative	
		6.3.3	Reduced Truck and Rail Alternative	
		6.3.4	Reduced Scale Alternative	6-6
	6.4	Altern	atives Analysis	6-7
		6.4.1	No Project Alternative	6-7
		6.4.2	Revised Operations Alternative	
		6.4.3	Aesthetics	6-10
		6.4.5	Summary of ROA Air Quality Analysis	6-11
		6.4.7	Existing Conditions	
		6.4.8	Thresholds of Significance	
		6.4.9	Impact Discussion	
		6.4.10	CEQA Appendix G Threshold Questions	
	6.5	Summ	ary Matrix	
	6.6	Enviro	nmentally Superior Alternative	
7	REF	ERENC	ES	
	Exect	utive Sur	nmary	7-1
	Chap	ter 1	Introduction	7-1
	Chap	ter 2	Project Description	7-1
	Chap	ter 3	Environmental Analysis	7-1
		3.1	Aesthetics	7-1
		3.2	Air Quality	
		3.3	Biological Resources	
		3.4	Cultural Resources	
		3.5	Geology and Soils	
		3.6	Greenhouse Gas Emissions	7-14
		3.7	Hazards and Hazardous Materials	
		3.8	Hydrology and Water Quality	
		3.9	Land Use and Planning	
		3.10	Noise	7-18
		3.11	Public Services and Recreation	7-19
		3.12	Transportation and Traffic	

	3.13 Utilities and Service Systems
	Chapter 5 Other CEQA Considerations
8	LIST OF PREPARERS
APP	ENDICES
A-1	Initial Study and Notice of Preparation
A-2	Comments on Initial Study/NOP
B-1	VMT Application
B-2	Orcem Application
С	Draft VMT Lighting Plan
D-1	Air Quality and Greenhouse Gas Emissions Report
D-2	Sea Level Rise Technical Memo
E-1	Biological Resources Assessment
E-2	Tree Survey
E-3	Biological Resources Assessment Peer Review and Update
E-4	Field Report: Intertidal Habitat and Marine Biota Survey
E-5	Technical Memo: Fish Species Inhabiting Lower Napa River and San Pablo Bay
E-6	Benthic Survey of Vallejo Marine Terminal LLC site
E-7	Technical Memo: Intertidal Habitat and Bio Community Survey at Proposed Kayak Launch Site
F	Historical Resources Evaluation
G	NAHC Records Search and Confidential Archaeological Resources Records Search
H-1	Geotechnical and Environmental Consultation
H-2	Preliminary Geotechnical Exploration
I-1	Site Investigation Report
I-2	Phase I Environmental Site Assessment
I-3	Phase II Soil and Groundwater Quality Investigation
I-4	Solano County Remedial Action Completion Certification
I-5	Final Backfill Report
I-6	Environmental Audit Summary
I-7	2007 Groundwater Monitoring Report
I-8	Asbestos Report
I-9	Hazards and Hazardous Materials Report
I-10	2012 Groundwater Monitoring Report
I-11	Covenant and Environmental Restrictions and Revised Site Management Plan
J-1	Stormwater Control Plan for 780 and 790 Derr Street
J-2	Ecocem/Orcem Hydro and Water Quality Narrative
J-3	Orcem Stormwater Management & Treatment Facilities Design Summary
т 4	Our set Standard Country 1 D1 - 1

J-4 Orcem Stormwater Control Plan

K-1	Environmental Noise Impact Assessment of the Proposed VMT Development,
	Vallejo, California
K-2	Environmental Noise Impact Assessment of the Proposed Orcem Development,
	Vallejo, California
K-3	Cumulative Environmental Noise Impact Assessment of the Proposed Orcem and
	VMT Developments
L	Transportation Technical Data
М	Orcem Revised Operations Alternative Air Quality and Health Risk Assessment
Ν	Mitigation and Monitoring Reporting Program
0	Vallejo Marine Terminal/Orcem Mitigation Monitoring and Reporting Program

FIGURES

1-1	Regional Map1-11
1-2	Vicinity Map 1-13
1-3	Aerial View of Project Site
2-1	Former General Mills Structures
2-2	Project Timeline Diagram
2-3	Revised VMT Project Platform Sections
2-4	Revised VMT Project Dredging Plan
2-5	Revised VMT Project Site Plan
2-6	Orcem Site Plan
2-7a	Orcem Site Sections B and C
2-7b	Orcem Site Sections E, F, and G
2-7c	Orcem Site Sections A and D
2-8	Proposed Public Access Improvements
2-9	Proposed Dock Removal
3.1-1	Photo Location Map
3.1-2	Photo Location 1 – Existing View and Visual Simulations
3.1-3	Photo Location 2 – Existing View and Visual Simulation
3.1-4	Photo Location 3 – Existing View and Visual Simulation
3.1-5	Photo Location 4 – Existing View and Visual Simulation
3.1-6	Photo Location 5 – Existing View and Visual Simulation
3.1-7	Photo Location 6 – Existing View and Visual Simulation
3.2-1	Cancer Risk: Unmitigated Full Operations (48 Ships) 3.2-47
3.2-2	Cancer Risk: Mitigated Full Operations (48 Ships)
3.3-1	Vegetative Communities
3.3-2	CNDDB Special-Status Species Occurrences
3.4-1	Historical Resources Survey Map
3.5-1	Site Geology and Topography

3.7-1	Former Chemical Storage and Remediation Areas
3.8-1	Flood Hazard Zones
3.8-2	VMT Preliminary Stormwater Management Plan
3.8-3	Orcem Drainage Plan
3.10-1	Land Use Compatibility Guidelines (Community Noise)
3.10-2	Noise Monitor Locations
3.10-3	Noise Sensitive Land Use Locations in the Project Vicinity
3.10-4	VMT Mobile Plant Operations Layout
3.10-5	VMT On-Site Rail Activity Areas
3.10-6	Orcem Plant Wheeled Loader Operations Area
3.10-7	Orcem On-Site Rail Activity Areas
3.10-8	Extent of Required Continuous Weld Rail (CWR) for Rail Activity
	Noise Mitigation
3.12-1	Project Study Area 3.12-39
3.12-2A	Existing Peak Hour Intersection Traffic Volumes
3.12-2B	Existing Peak Hour Intersection Traffic Volumes
3.12-3	Trip Distribution for ORCEM and VMT
3.12-4A	Vallejo Marine Terminal Project Trip Assignment
3.12-4B	Vallejo Marine Terminal Project Trip Assignment
3.12-5A	Orcem Project Trip Assignment
3.12 - 5B	Orcem Project Trip Assignment
3.12-6A	Combined Projects Project Trip Assignment
3.12-7A	Existing + Vallejo Marine Terminal Peak Hour Intersection Traffic Volumes 3.12-59
3.12 - 7B	Existing + Vallejo Marine Terminal Peak Hour Intersection Traffic Volumes 3.12-6
3.12-8A	Existing + Orcem Peak Hour Intersection Traffic Volumes
3.12-8B	Existing + Orcem Peak Hour Intersection Traffic Volumes
3.12 - 9A	Existing + Combined Project Peak Hour Intersection Traffic Volumes
3.12 - 9B	Existing + Combined Project Peak Hour Intersection Traffic Volumes
3.12-10A	Cumulative (2040) No Project Peak Hour Intersection Traffic Volumes
3.12-10B	Cumulative (2040) No Project Peak Hour Intersection Traffic Volumes
3.12-11A	Cumulative (2040) + Vallejo MarineTerminal Peak Hour Intersection
	Traffic Volumes
3.12-11B	Cumulative (2040) + Vallejo MarineTerminal Peak Hour Intersection
	Traffic Volumes
3.12-12A	Cumulative (2040) + Orcem Peak Hour Intersection Traffic Volumes
3.12-12B	Cumulative (2040) + Orcem Peak Hour Intersection Traffic Volumes
3.12-13A	Cumulative (2040) + Combined Projects Peak Hour Intersection
	Traffic Volumes

3.12-13B Cumulative (2040) + Combined Projects Peak Hour Intersection	1
Traffic Volumes	
6-1 VMT-Orcem ROA Project View 1, ORcem Modes 2-3	
6-2 VMT-Orcem ROA Project View 2. Orcem Modes 2-3	6-47

EXHIBITS

6-3	Source Contribution to Annual NOx Emission	6-14
6-4	Source Contributions Annual PM ₁₀ Emissions	6-14

TABLES

ES-1	Summary of Potentially Significant Environmental Impacts	ES-6
2-1	Existing General Mills Structures	2-1
2-2	Proposed Orcem Buildings, Equipment, and Major Facilities	2-6
2-3	Summary of Maximum Material Volumes and Transport Methods - VMT	
	Volumes (with Orcem Materials Included)	2-9
2-4	Summary of Maximum Material Volumes and Transport Methods – Orcem	
	Phase 1 and Phase 2 Volumes	2-11
3.2-1	Ambient Air Quality Standards	. 3.2-2
3.2-2	BAAQMD Attainment Classification	. 3.2-7
3.2-3	Non-criteria Pollutant Significant Emission Levels	. 3.2-8
3.2-4	Top Ten Toxic Air Contaminants (TACs)	. 3.2-9
3.2-5	Ambient Air Quality Data	3.2-12
3.2-6	Thresholds of Significance	3.2-15
3.2-7	VMT Construction Emissions	3.2-22
3.2-8	Orcem Construction Emissions	3.2-23
3.2-9	Combined VMT and Orcem Average Daily Construction $Emissions - 2017^1 \dots$	3.2-24
3.2-10	VMT and Orcem Operational Throughput	3.2-26
3.2-11	Maximum Annual Emissions of Criteria Pollutants - VMT	3.2-29
3.2-12	Orcem Annual Emissions of Criteria Pollutants (Phase 2)	3.2-31
3.2-13	Maximum Annual Emissions of Criteria Pollutants from the	
	Combined Operations of VMT and Orcem	3.2-32
3.2-14	Sensitive Receptors Within 2.5 Miles of the Project	3.2-35
3.2-15	On-Site and Near-Site Construction DPM and PM _{2.5} Emissions	3.2-36
3.2-16	Local Carbon Monoxide Emissions	3.2-38
3.2-17	Project Health Risks Impacts	3.2-39
3.2-18	Cumulative Health Risks	3.2-41
3.3-1	Special-Status Wildlife Species with Potential to Occur On or	
	Near the Project Site	3.3-18

3.3-2	Special-Status Fish and Marine Mammal Species That May Occur	
	Within the Waters of the Study Area	3.3-30
3.3-3	In-Water Acreage of the Napa River Affected by the VMT	
	Project Component	3.3-45
3.3-4	Environmental Work Windows for Maintenance Dredging Activities	
	Established in the Long-Term Management Strategy for	
	San Francisco Bay	3.3-50
3.3-5	Estimated Near-Source Underwater Noise Levels From Pile Driving	3.3-53
3.3-6	Estimated Vibratory and Impact Hammer Pile Driving Sound Levels	
	and Disturbance to Criteria Levels	3.3-55
3.3-7	Potential Effects of Varying Noise Levels to Fish and Marine Mammals	3.3-56
3.3-8	Summary of NOAA Established Permanent Threshold Shift ¹ and Temporar	y
	Threshold Shift ² Sound Levels ³ from Underwater Noise Levels for	
	Marine Mammals	3.3-56
3.5-1	Soil Types in the Proposed Project Area	3.5-10
3.5-2	Slope Stability and Seismic Slope Displacements	3.5-12
3.6-1	Greenhouse Gas Sources in California	3.6-11
3.6-2	VMT Construction Greenhouse Gas Emissions	3.6-16
3.6-3	Orcem Construction Greenhouse Gas Emissions	3.6-17
3.6-4	Combined VMT and Orcem Construction Greenhouse Gas Emissions	3.6-17
3.6-5	VMT and Orcem Operational Throughput	3.6-19
3.6-6	VMT Operational GHG Emissions	3.6-20
3.6-7	Orcem Plant Operational GHG Emissions	3.6-21
3.6-8	Annual CO ₂ E Reductions Associated with Production of GGBFS	
	by Orcem (MT)	3.6-22
3.6-9	Annual CO ₂ E Emissions from Combined VMT and Orcem Operations	3.6-23
3.6-10	Proposed Project Consistency with City of Vallejo Climate Action Plan	3.6-24
3.6-11	Sea Level Rise Projections for San Francisco, California	
	(NRC 2012 Report)	3.6-30
3.7-1	Subsurface Sediments in Mare Island Strait	3.7-10
3.7-2	Maximum Documented Soil Concentrations – Before and After Cleanup	3.7-14
3.8-1	Existing Beneficial Uses of Relevant Water Bodies	3.8-13
3.8-2	CWA Section 303(d) Impairments in Northern San Francisco Bay-Delta	3.8-14
3.8-3	Mare Island Strait Water Quality Monitoring Results	3.8-14
3.8-4	VMT Pre-Development and Post-Development Impervious Surfaces	3.8-22
3.9-1	Existing General Mills Structures	3.9-9
3.9-2	Consistency of the Proposed Project with Relevant Goals, Objectives,	
	and Policies	3.9-12
3.10-1	EPA Noise Guidelines	3.10-4

3.10-2	Summary of Results for Unattended (Long-Term) Measurement Locations	3.10-10
3.10-3	Summary of Results for Attended (Short-Term) Measurement Locations	3.10-10
3.10-4	Noise-Sensitive Locations	3.10-13
3.10-5	Typical Construction Noise Levels	3.10-15
3.10-6	Predicted Maximum VMT Construction Noise Levels at Closest	
	Sensitive Receptors	3.10-16
3.10-7	Noise Levels due to VMT Operations	3.10-18
3.10-8	Noise Levels Due to Off-Site Truck Trips Associated with	
	VMT Operations	3.10-20
3.10-9	Individual Component Noise Levels Due to VMT Rail Activity	3.10-21
3.10-10	Total Noise Levels due to VMT Rail Activity	3.10-22
3.10-11	Noise Levels from All VMT Operations Activity (Combined)	3.10-23
3.10-12	Significance Determination for Noise Levels from All VMT	
	Operations Activity (Combined)	3.10-24
3.10-13	Predicted Maximum Orcem Construction Noise Levels at Closest	
	Sensitive Receptors	3.10-26
3.10-14	Noise Levels due to Orcem Fixed and Mobile Plant Operations - Phase 1	3.10-28
3.10-15	Noise Levels due to Orcem Fixed and Mobile Plant Operations - Phase 2	3.10-29
3.10-16	Noise Levels due to Orcem Vessel Unloading Activity	3.10-29
3.10-17	Noise Levels due to Truck Movements Associated with Orcem	
	Operations – Phase 1	3.10-30
3.10-18	Noise Levels due to Truck Movements Associated with Orcem	
	Operations – Phase 2	3.10-31
3.10-19	Individual Component Noise Levels due to Orcem Rail Activity	3.10-32
3.10-20	Total Noise Levels Due to Orcem Rail Activity	3.10-33
3.10-21	Noise Levels from All Orcem Operations Activity Plus Truck Movements	
	(Combined) – Scenario A	3.10-35
3.10-22	Significance Determination for Noise Levels from All Orcem	
	Operations Activity Plus Truck Movements (Combined) - Scenario A	3.10-37
3.10-23	Noise Levels from All Orcem Operations Activity Plus Truck	
	Movements Plus Vessel Unloading (Combined) - Scenario B	3.10-38
3.10-24	Significance Determination for Noise Levels from All Orcem	
	Operations Activity Plus Truck Movements Plus Vessel	
	Unloading (Combined) – Scenario B	3.10-40
3.10-25	Noise Levels from All Orcem Operations Activity Plus Truck	
	Movements Plus Vessel Unloading, Plus Rail (Combined) - Scenario C	3.10-41
3.10-26	Significance Determination for Noise Levels from All Orcem	
	Operations Activity Plus Truck Movements Plus Vessel Unloading,	
	Plus Rail (Combined) –Scenario C	3.10-43

3.10-27	Predicted Maximum VMT and Orcem Construction Noise Levels	
	at Closest Sensitive Receptors	3.10-44
3.10-28	Combined Noise Levels from All VMT and Orcem Operations Activity	3.10-46
3.10-29	Significance Determination for Combined Noise Levels from	
	All VMT and Orcem Operations	3.10-48
3.10-30	Typical Construction Ground Vibration Levels	3.10-49
3.10-31	Orcem Plant Exhaust Stack Mitigation Requirements	3.10-58
3.10-32	Mitigated Noise Levels from All VMT Operations Activity (Combined)	3.10-60
3.12-1	Signalized Intersection LOS Criteria	3.12-8
3.12-2	Unsignalized Intersection LOS Criteria	3.12-8
3.12-3	Volume-to-Capacity (V/C) Thresholds for Project Impacts	
	(Signalized Intersections)	3.12-9
3.12-4	Existing Peak Hour Intersection LOS	3.12-10
3.12-5	Freeway LOS Definitions	3.12-12
3.12-6	Existing Freeway Operations	3.12-13
3.12-7	Existing Grade Crossings	3.12-15
3.12-8	Vallejo Marine Terminal Trip Generation	3.12-19
3.12-9	Orcem Trip Generation	3.12-20
3.12-10	Existing Plus Project Peak Hour Intersection Service Levels	3.12-21
3.12-11	Rail Crossing Evaluation	3.12-23
3.12-12	Existing Plus Project Freeway Operations	3.12-24
3.12-13	Year 2040 Peak Hour Intersection LOS ¹	3.12-26
3.12-14	Cumulative (Year 2040) With Project Freeway Operations	3.12-29
6.1	Cancer Risk in a Million	6-19
6.2	Ambient Air Quality Data	
6.3	ROA On-Road Truck Revisions	6-23
6.4	ROA Ship Revisions	6-24
6.5	ROA Rail Revisions	6-25
6.6	ROA On-Site Equipment Revisions	6-25
6.7	ROA Health Risk Assessment Revisions	6-26
6.8	Material Throughput and Activity: Maximum Combined Scenario for	
	Criteria Pollutants	6-27
6.9	Emissions Summary: Maximum Combined Scenario	6-31
6.9	Emissions Summary: Maximum Combined Scenario	6-34
6.11	ROA Health Risks Impacts	6-37
6-12	Summary of Impacts from Alternatives	

ACRONYMS AND ABBREVIATIONS

Acronym/Abbreviation	Definition
°F	degrees Fahrenheit
AB	Assembly Bill
ACM	Asbestos-containing materials
AFY	acre-feet per year
AMSL	above mean sea level
ATS	Active Treatment System
BAAQMD	Bay Area Air Quality Management District
BACT	Best Available Control Technology
BCDC	Bay Conservation and Development Commission
BMPs	best management practices
CAA	Clean Air Act (federal)
CAAQS	California Ambient Air Quality Standards
CalOSHA	California Occupational Safety and Health Administration
CalRecycle	California Department of Resources Recycling and Recovery
Caltrans	California Department of Transportation
CA-MUTCD	California Manual of Uniform Traffic Control Devices
CAP	Clean Air Plan
CARB	California Air Resources Board
CARE	Community Air Risk Evaluation
CBC	California Building Code
CCR	California Code of Regulations
CDFW	California Department of Fish and Wildlife
CEC	California Energy Commission
CEQA	California Environmental Quality Act
CESA	California Endangered Species Act
CFR	Code Federal Regulations
CGS	California Geological Survey
CH ₄	methane
City	City of Vallejo
CMP	Congestion Management Plan
CNDDB	California Natural Diversity Database
CNEL	Community Noise Equivalent Level
CNG	compressed natural gas
CO ₂	carbon dioxide
CO ₂ E	carbon dioxide equivalent
CO-CAT	Coastal and Ocean Working Group of the California Climate Action Team
CPUC	California Public Utilities Commission
CREATE	Chicago Rail Efficiency and Transportation Efficiency
CRHR	California Register of Historical Resources
CUPA	Certified Unified Program Agency
CWA	Clean Water Act
CWR	Continuous Welded Rail

Acronym/Abbreviation	Definition
cyd	cubic yards
CZMA	Coastal Zone Management Act
dB	decibel
DFEIR	Draft Final Environmental Impact Report
DHS	California Department of Health Services
DPM	diesel particulate matter
DPS	distinct population segment
dscf	dry standard cubic foot
EIR	Environmental Impact Report
EPA	U.S. Environmental Protection Agency
ESA	Endangered Species Act
FEMA	Federal Emergency Management Agency
FHWA	Federal Highway Administration
FMP	fishery management plan
FOS	factor of safety
FTA	Federal Transit Administration
GBFS	granulated blast furnace slag
GGBFS	ground granulated blast furnace slag
GHG	greenhouse gas
GVRD	Greater Vallejo Recreation District
GWP	global warming potential
HAG	hot air generator
HARP	Hotspots Analysis Reporting Program
HCM	Highway Capacity Manual
HFC	hydroflourocarbon
HI	Hazard Index
HMBP	Hazardous Materials Business Plan
Hz	hertz
I-780	Interstate Highway 780
1-80	Interstate Highway 80
IEP	Interagency Ecological Program
IGP	Industrial General Permit
kHz	kilohertz
LAFCO	Solano County Local Agency Formation Commission
lbs/year	pounds per year
L _{dn}	day-night sound level
L _{eq}	equivalent sound level
L _{max}	maximum sound level
Lmin	minimum sound level
LID	Low Impact Development
LOS	level of service
LTMS	Long-Term Management Strategy
LUST	Leaking Underground Storage Tank

Acronym/Abbreviation	Definition
MBTA	Migratory Bird Treaty Act
mgd	million gallons per day
MLLW	mean lower low water
MMPA	Marine Mammal Protection Act
MMT	million metric tons
mph	miles per hour
MMRP	Mitigation and Monitoring Reporting Program
MRP	Municipal Regional Permit
MMscf	million standard cubic feet
MSDS	materials safety data sheets
MT	metric tons
MTSA	Maritime Transportation Security Act
N ₂ O	nitrous oxide
NAAQS	National Ambient Air Quality Standards
NAHC	Native American Heritage Commission
NEPA	National Environmental Policy Act
nm	nautical mile
NOAA	National Oceanic and Atmospheric Administration
NOAA Fisheries	National Oceanic and Atmospheric Administration Marine Fisheries Service
NOP	Notice of Preparation
NPDES	National Pollutant Discharge Elimination System
NRHP	National Register of Historic Places
NSL	noise-sensitive location
O ₃	ozone
Orcem	Orcem California Inc.
OSHA	Occupational Safety and Health Administration
PAH	polycyclic aromatic hydrocarbons
PCB	polychlorinated biphenyls
PCE	passenger car equivalents
PFC	perfluorocarbon
PGA	peak ground acceleration
PG&E	Pacific Gas and Electric
ppt	parts per trillion
PPV	perturbation projection vector
PSD	Prevention of Significant Deterioration
PSHA	probabilistic seismic hazard assessment
QSD/QSP	Qualified SWPPP Developer/Qualified SWPPP Practitioner
RCNM	Roadway Construction Noise Model
RCRA	Resource Conservation and Recovery Act
REL	reference exposure level
ROA	Revised Operations Alternative
ROG	reactive organic gas
RPS	Renewable Portfolio Standard

Acronym/Abbreviation	Definition
RWQCB	Regional Water Quality Control Board
SAFE Port Act	Security and Accountability for Every Port Act
SAV	submerged aquatic vegetation
SB	Senate Bill
SF6	sulfur hexafluoride
SFBAAB	San Francisco Bay Area Air Basin
SHPO	State Historic Preservation Office
SLR	sea level rise
<u>SMP</u>	Site Management Plan
SR	State Route
SRI	solar reflectance index
SSMP	Sanitary Sewer Management Plan
STA	Solano Transportation Authority
SWPPP	Stormwater Pollution Prevention Plan
SWRCB	State Water Resources Control Board
TACs	Toxic Air Contaminants
TMDLs	total maximum daily loads
TOG	total organic gas
tpy	tons per year
USACE	U.S. Army Corps of Engineers
USDA	U.S. Department of Agriculture
USFWS	U.S. Fish and Wildlife Service
<u>USGS</u>	U.S. Geological Survey
UWMP	Urban Water Management Plan
V/C	volume-to-capacity
VFD	Vallejo Fire Department
VMT	Vallejo Marine Terminal LLC
VPD	Vallejo Police Department
VRM	vertical roller mill
VCUSD	Vallejo City Unified School District
VSFCD	Vallejo Sanitation and Flood Control District
WDRs	Waste Discharge Requirements
WTP	water treatment plant
WWTP	Wastewater Treatment Plant

4.1 INTRODUCTION

Although the environmental effects of an individual project may not be significant when that project is considered independently, the combined effects of several projects may be significant when considered collectively. Such impacts are "cumulative impacts." Section 15355 of the California Environmental Quality Act (CEQA) Guidelines defines cumulative impacts as "two or more individual effects which, when considered together, are considerable or which compound or increase other environmental impacts" (14 CCR 15000 et seq.). Section 15130 of the CEQA Guidelines provides guidance for analyzing significant cumulative impacts in an Environmental Impact Report (EIR). According to this section of the CEQA Guidelines, the discussion of cumulative impacts "need not provide as great detail as is provided for the effects attributable to the project alone. The discussion should be guided by standards of practicality and reasonableness." The discussion should also focus only on significant effects resulting from the project's incremental effects and the effects of other projects. According to Section 15130(a)(1), "An EIR should not discuss impacts which do not result in part from the project evaluated in the EIR."

However, substantial cumulative impacts more often result from the combined effect of past, present, and future projects located in proximity to the project under review. Therefore, it is important for a cumulative impacts analysis to be viewed over time and in conjunction with other related past, present, and reasonably foreseeable future developments whose impacts might compound or interrelate with those of the project under review.

4.2 METHODOLOGY

According to Section 15130(b) of the CEQA Guidelines, cumulative impact analysis may be conducted and presented by either of two methods: (1) a list of past, present, and probable activities producing related or cumulative impacts; or (2) a summary of projections contained in an adopted general plan or related planning document, or in a prior environmental document that has been adopted or certified, which described or evaluated regional or area-wide conditions contributing to the cumulative impact. Other than for transportation and traffic, the cumulative list approach has been utilized in the cumulative analysis presented in this chapter, as discussed below. Cumulative traffic and transportation impacts have been analyzed utilizing the summary of projections method.

4.2.1 Cumulative Projects List

The cumulative impacts analysis is based on a list of projects within the proposed project's study area that either have applications submitted or approved, are under construction, or have recently been completed. Based on information provided by the City of Vallejo staff <u>under the Draft EIR</u>,

three cumulative projects were considered in this analysis. One of these projects, the quick service restaurant, did not move forward and thus has been removed from this list. The other two projects are still considered current as planned. One project in the planning stages at California State University Maritime Academy has been added to this list.

- 1. Proposed 2,580-square-foot quick service restaurant and 1,300-square-foot convenience store with gasoline sales located at 1217 Fifth Street/Sonoma Boulevard.
- 2.1.Anchor Storage 925-unit self-storage facility with an on-site manager's unit on 3.9 acres, located at 501 Solano Avenue.
- 2. Former Vallejo Manufactured Gas Plant (MGP) Site Cleanup Remediation of 26-acre former MGP site located at the southwest corner of Curtola Parkway and Sonoma Boulevard. Remediation is expected to occur in phases between 2017 and 2019 and would be under the oversight of the California Department of Toxic Substances.
- 3. <u>The California State University Maritime Academy Master Plan includes one project that is</u> planned for completion within the same timeframe as the proposed project. This project would remodel and replace a 250-bed residence hall to a facility that would accommodate 550 beds.

4.3 CUMULATIVE IMPACT ANALYSIS

The discussion below evaluates the potential for the proposed project to contribute to an adverse cumulative impact on the environment. For issues addressed in this EIR, the thresholds used to determine significance are those presented in each of the sections of Chapter 3, Environmental Analysis. For issues in which project impacts were determined to be less than significant during the preliminary environmental review process, the thresholds consist of the questions posed for that respective issue in Appendix G of the CEQA Guidelines. For each resource area, an introductory statement is made regarding what would amount to a significant cumulative impact in that resource area. Discussion is then presented regarding the potential for the identified cumulative projects to result in such a cumulative impact, followed by discussion of whether the project's contribution to any cumulative impact would be cumulatively considerable.

4.3.1 Aesthetics

As described in Section 3.1, Aesthetics, the proposed project would not have any significant impacts to aesthetics aside from a potentially significant impact due to proposed lighting; however, this impact would be reduced to a less-than-significant level with mitigation. The cumulative projects are not located within sight of the proposed project and would therefore not impact the aesthetics of the proposed project site. Although the cumulative projects may introduce new sources of lighting, the lighting would not be visible from the proposed project site. Cumulative impacts to aesthetics would therefore be less than significant.

4.3.2 Air Quality

As described in Section 3.2, Air Quality, by its nature air pollution is largely a cumulative impact; no single project is sufficient in size to, by itself, result in nonattainment of ambient air quality standards. In developing thresholds of significance for air pollutants, the Bay Area Air Quality Management District (BAAQMD) considered the emission levels for which a project's individual emissions would be cumulatively considerable. If a project exceeds the identified significance thresholds, its emissions would be considered cumulatively considerable, resulting in significant adverse air quality impacts to the region's existing air quality conditions. The proposed project would conflict with the Bay Area 2010 Clean Air Plan due to the proposed rezoning of the 5.25-acre portion of the project site. In addition, the proposed project would exceed the BAAQMD threshold for cancer risk. Mitigation as described in Section 3.2.5 would help reduce project impacts, including reducing the potential for cancer risk to below a level of significance. However, impacts related to the conflict with the 2010 Clean Air Plan and NO_x emissions during operations would remain significant and unavoidable. Project operations would therefore result in significant cumulative air quality impacts.

As described in Section 3.2, Air Quality, construction of the cumulative projects would be short term and temporary in nature. Conversion of the California Maritime housing facilitystruction of the quick-service restaurant and gas station convenience store, and selfstorage facility would contribute minimal emissions during construction, and would not be anticipated to result in substantial emissions when considered in combination with the proposed project. Construction of the Pacific Gas & Electric (PG&E) Southern Waterfront site would consist of demolition of on-site structures, site preparation, and remediation activities. Pollutants generated as a result of these activities would consist primarily of fugitive dust as a result of demolition and site preparation/ remediation activities. The PG&E remediation project would include on-site fugitive dust monitoring as part of its demolition work plan and health and safety plan. On-site monitoring would ensure adequate implementation of fugitive dust control measures during dust-generating activities, and would mitigate visible dust plumes and related fugitive dust impacts to a level below significance. As fugitive dust impacts are generally localized to individual project sites, and on-site emissions would be sufficiently mitigated through demolition and dust control measures, coupled with implementation of BAAQMD best management practices for all cumulative projects, cumulative impacts related to fugitive dust would be considered less than significant. Construction of the proposed project would not exceed BAAQMD construction thresholds for any criteria pollutants; therefore, cumulative impacts would be considered less than significant during the temporary construction period.

4.3.3 Biological Resources

As described in Section 3.3, Biological Resources, the proposed project would have potentially significant impacts on both marine and terrestrial species due to construction activities (noise and structure demolition) and alteration of habitat from facility operation; however, these impacts would be reduced to less-than-significant levels with mitigation. The cumulative projects are proposed for sites that are in highly impacted areas with limited biological resources. Thus, significant biological resources are not expected from these projects. Therefore, the proposed project in combination with the cumulative projects would not result in significant cumulative effects related to biological resources.

4.3.4 Cultural Resources

The proposed project would contribute to a cumulative impact on cultural resources if its incremental effects coincided and potentially compounded with effects from other reasonably foreseeable future projects to result in a significant impact on local cultural resources. As described in Section 3.4, Cultural Resources, the proposed project would result in a significant and unavoidable impact to historic resources due to demolition of the existing flour mill, grain silos, and dock. In addition, construction of the proposed project could result in significant impacts to buildings not proposed to be demolished as well as significant impacts to archaeological resources, fossils, or human remains, if discovered on site. However, with implementation of mitigation measures, these impacts would be reduced to less-than-significant levels.

The cumulative projects are located on vacant sites or <u>with the exception of California Maritime</u>, sites that do not contain any historic resources and would therefore not result in the demolition of any historic structures. <u>The California Maritime campus may contain historic structures</u>; however, mitigation included in that project calls for a qualified architectural historian that meets or exceeds the Secretary of the Interior's Standards for History and/or Architectural History to record project design based upon professional standards and assess its significance under CEQA Guidelines. The campuses distance from the project site and the fact that the two facilities did not overlap in purpose or landscapes, would result in less-than-significant cumulative impacts related to historic resources.

Although there could be potential for the discovery of unknown archaeological or paleontological resources, it is anticipated that standard measures would be in place to ensure impacts are less than significant. Therefore, although the proposed project would result in a significant and unavoidable impact to historic resources, the cumulative impact in combination with the cumulative projects would not be significant.

4.3.5 Geology and Soils

As described in Section 3.5, the proposed project would not result in any significant impacts related to geology and soils aside from the potential for landslides; however, this impact would be reduced to less than significant with mitigation. Both the proposed project and the cumulative projects would be required to comply with the California Building Code to ensure impacts due to seismic activity are minimized. In addition, the cumulative projects are located on generally flat sites that are not at risk for landslide. Therefore, the proposed project in combination with the cumulative projects would not result in a significant cumulative impact to geology and soils.

4.3.6 Greenhouse Gas Emissions

As described in Section 3.6, Greenhouse Gas Emissions, operational emissions of both the Orcem and VMT components of the proposed project would exceed the BAAQMD threshold for operational greenhouse gas (GHG) emissions. In addition, while the proposed project would comply with applicable implementation measures of the City's 2012 Climate Action Plan, it cannot be guaranteed that the project would be consistent with the objectives of the City's Climate Action Plan to achieve reduction targets established for 2020 and 2035. This is because the City's adopted CAP does not cover marine and rail operations which are an important part of the proposed project. The proposed project would also be exposed to impacts due to sea level rise that would be reduced to below a level of significance with mitigation. The cumulative projects include small commercial <u>and educational</u> operations and a temporary remediation project that are not expected to generate significant GHG emissions. However, the BAAQMD considers any project that would generate GHG emissions above the BAAQMD threshold, to contribute substantially to a cumulative impact. Therefore, a significant cumulative impact to GHG emissions would occur as a result of the project, and this impact would be significant and unavoidable.

4.3.7 Hazards and Hazardous Materials

As described in Section 3.7, Hazards and Hazardous Materials, construction of the proposed project would require the temporary use of hazardous materials, such as diesel fuels, lubricants, solvents, and asphalt. Without mitigation to ensure proper handling, storage, disposal, and emergency response planning, impacts could be significant but are reduced to less-than-significant levels with mitigation as outlined in Section 3.8. The cumulative projects would likely involve similar temporary use of hazardous materials during the construction phase, particularly the former MGP site cleanup. However, the cumulative projects would be required to manage hazardous materials in compliance with both state and federal regulations on hazardous materials such that their individual effects would be mitigated to less-than-significant levels. In addition, the former MGP site cleanup would be overseen by the California Department of Toxic Substances Control to ensure that impacts related to hazardous and hazardous

materials are minimized. Given that mitigation would be required, the proposed project would not contribute a cumulatively considerable effect.

The proposed project would also involve the dredging of sediment in Mare Island Strait which may contain elevated concentrations of metal contaminants. Transport and disposal of the dredged material could result in a significant impact. With mitigation, this impact is reduced to less-than-significant levels as outlined in Section 3.7. The cumulative projects do not involve dredging or the transport or disposal of dredged material; therefore, the proposed and cumulative projects would not result in cumulative effects related to dredging.

The proposed project involves the demolition of buildings that were found to contain asbestoscontaining materials. Disposal, transport or use of these materials as engineered fill could result in a significant impact. These impacts are reduced to less-than-significant levels with the mitigation outlined in Section 3.7. The cumulative projects are located on vacant sites or <u>with the exception of</u> <u>California Maritime</u>, sites with minimal buildings and therefore would not require demolition and subsequent transport or disposal of any asbestos-containing materials. <u>The renovation of the existing</u> <u>student housing building may require disposal of asbestos-containing materials</u>, but mitigation would <u>be applied to disposal for this project that would remove the</u> Therefore, there is no risk of a cumulative effect due to the handling of asbestos-containing materials.

4.3.8 Hydrology and Water Quality

Cumulative Impacts to hydrology and water quality would result if the proposed project and the cumulative projects contributed incrementally to a net effect on water quality and hydrology in the project vicinity, or any downstream body of water. As described in Section 3.8, Hydrology and Water Quality, the proposed project would result in a potentially significant impact from the risk of mobilizing pollutants currently sequestered in bay sediments and the pilings of the former General Mills wharf during dredge and fill, and piling removal as part of the VMT project component. These impacts are reduced to less-than-significant levels as outlined in Section 3.8, with mitigation measures MM-3.8-1, MM-3.8-2, MM-3.3-3, and MM-3.3-4. Two of the cumulative projects—the convenience storerenovation of student housing and storage unit facility—would be located away from the shoreline and would not involve any proposed marine construction activities. The former MGP site cleanup would be located adjacent to the shoreline; however, no in-water work would occur that would contribute to a potential cumulative impact. Therefore, there is no risk of a cumulative effect due to those dredge and fill or piling removal activities, and the proposed project's contribution to a cumulative effect would not be cumulatively considerable.

4.3.9 Land Use and Planning

A cumulative impact to land use and planning could occur if the proposed and cumulative projects contributed incrementally to a land use impact that is inconsistent with local plans and policies, including those set by the Bay Conservation and Development Commission, the City of Vallejo General Plan, and the Solano County General Plan. As described in Section 3.9, Land Use and Planning, the proposed project does not result in any significant impacts. However, the proposed project would involve the annexation and re-designation of 5.25 acres of land currently designated as "Park and Recreation" use in the Solano County General Plan, into "Employment" use by the City of Vallejo. This impact is considered to be less than significant as described in Section 3.9. Similarly, t<u>T</u>he other cumulative projects do not involve any changes in land use designation under the Solano County General Plan and are not anticipated to result in any significant impacts since the City would ensure consistency with applicable plans and policies. Therefore, cumulative impacts to land use and planning would be less than significant.

4.3.10 Noise

A cumulative impact on the noise environment in the project vicinity would result if the proposed project and the cumulative projects in combination resulted in a noise impact greater than either project generates independently. At least one of the areas designated as a potentially noise-sensitive location for the proposed project (the Seawitch Drive Apartments) is located within 0.5-mile of the cumulative project located at 1217 5th Street, meaning that there is potential for noise from the proposed project to have a cumulative effect in that area. The operational noise impact of the proposed project is considered to be less than significant based on the railroad's policy that does not allow trains with worn wheels to operate on tracks (thus removing the former potential noise impact) and unavoidable after mitigation since the mitigation cannot be guaranteed. The project located at 1217 5th Street would include a quick-service restaurant and convenience store with gasoline sales, which are not expected to generate long-term permanent noise increases. In addition, the long-term permanent noise impact of the other cumulative projects is not likely to be significant given the location and nature of these projects. Therefore, a cumulative effect due to operational impacts is not anticipated.

If the timing of construction of the cumulative projects coincides with the proposed project, the projects could contribute to a temporary cumulative impact on noise in the area. However, given the location of the proposed project in relation to the cumulative projects, the potential for cumulative noise effects during construction is unlikely. As described in Section 3.10, the construction noise impacts of the proposed project would be mitigated to less-than-significant levels. Therefore, cumulative noise impacts would be less than significant.

4.3.11 Public Services and Recreation

As described in Section 3.11, Public Services and Recreation, the proposed project would not result in any significant impacts to public services and recreation. Although the project would slightly increase demands for police and fire services, the impact would not be significant. Similarly, the cumulative projects may cause a slight increase in demands for police and fire services; however, the projects are located in developed areas of the City that are currently served by the City's police and fire departments or served by California Maritime service providers. Neither the proposed project nor the cumulative projects would cause an increase in demands for recreation facilities. Therefore, cumulative impacts to public services and recreation would not be significant.

4.3.12 Transportation and Traffic

As described in Section 3.12, Transportation and Traffic, the proposed project would result in increased truck traffic to and from the project site that could temporarily inhibit vehicular and non-vehicular travel. This impact is reduced to less-than-significant levels with mitigation described in Section 3.12. However, two of the cumulative projects are located on near Sonoma Boulevard along one of the truck routes that connects the project site to Interstate 80 (I-80) West. Further increases in traffic due to added truck trips, or construction equipment for the cumulative projects on Sonoma Boulevard, could result in a temporary cumulative effect on local traffic congestion during the construction phase, but due to the minor short-term increase, this impact is not expected to be significant.

The proposed project is projected to generate train trips that would cause increased delays at train crossings. As described in Section 3.12, this impact would remain significant and unavoidable with mitigation since the mitigation cannot be guaranteed. The cumulative projects would not utilize railways. Therefore, although the project's impacts due to delays at train crossings would be significant, there would not be a significant cumulative effect on transportation and traffic as a result of train traffic from the cumulative projects.

The proposed project would also require improvements to roads in order to safely handle the increased truck traffic associated with daily operation of the proposed project which constitutes a significant impact but is reduced to less-than-significant levels with mitigation. Two of the cumulative projects—the <u>convenience_storestudent housing replacement</u> and storage unit facility—are likely to involve a small number of truck trips <u>due to operational increases</u>, <u>for construction</u>, restocking and delivery. However, the volume of truck traffic for those purposes is unlikely to warrant any capital improvements to roadways. Therefore, there would not be a significant cumulative effect as a result of increased demand for road maintenance and improvements.

The increase in train, automobile, and truck trips resulting from operation of the proposed project is likely to result in a significant impact on pedestrian and bicycle transit by making those modes of transportation less safe and convenient. This effect would be reduced to less-than-significant levels with mitigation. The cumulative projects would have minimal operational impacts on automobile and truck traffic. Therefore, there would be no cumulative impact on pedestrian and bicycle traffic as a result of the proposed project in combination with the cumulative projects. Cumulative traffic impacts would be less than significant.

4.3.13 Utilities and Service Systems

A significant cumulative impact would result if the proposed project and other nearby projects contributed to a net impact on local utilities and service systems such as overburdening municipal waste management services or depleting available municipal water. As described in Section 3.13, Utilities and Service Systems, the proposed project would have less-than-significant impacts on wastewater treatment and water consumption. These impacts would be reduced to less-than-significant levels with mitigations outlined in Section 3.13. The cumulative projects would also contribute an incremental increase on demand for water and wastewater treatment. However, the cumulative projects are small in scale and therefore are unlikely to result in a cumulative effect when added to the demands of the proposed project.

The proposed project would result in the generation of 170 tons of debris during construction and a projected 10 cubic yards/week during operation that would need to be disposed at Keller Canyon Landfill. As described in Section 3.13, this impact is considered to be less than significant. The cumulative projects would likely be served by the same municipal waste management service and disposed at the same location at Keller Canyon. Keller Canyon currently receives 3,500 tons of solid waste per day and has a remaining capacity of 63,408,410 cubic yards. Given those capacities, the combined impacts of the proposed project and cumulative projects would not result in a cumulative effect.

The proposed project is likely to be a large consumer of natural gas and electricity to power milling equipment and to dry the slag used in the production of the cement product. PG&E performed a feasibility study for the proposed project and concluded that existing circuits in Vallejo have capacity to accommodate the projects demands. This impact is also considered to be less than significant. The cumulative projects would have a small impact on electricity and natural gas relative to the proposed project. Therefore the cumulative effect from the combined impacts of the cumulative projects and proposed project would not be significant.

INTENTIONALLY LEFT BLANK

This chapter includes the following other considerations that are required to be discussed in an Environmental Impact Report (EIR):

- Effects Not Found to be Significant (Section 5.1)
- Significant and Unavoidable Environmental Impacts (Section 5.2)
- Significant and Irreversible Environmental Effects (Section 5.3)
- Growth Inducement (Section 5.4)

5.1 EFFECTS NOT FOUND TO BE SIGNIFICANT

Based on the analysis provided in the Initial Study, the proposed project would not result in significant impacts related to the following topics, which are not further evaluated in the EIR:

- Agricultural and Forest Resources
- Mineral Resources
- Population and Housing

Additional information and discussion regarding the effects found not to be significant can be found in the Initial Study, which is included as Appendix A of this EIR.

5.2 SIGNIFICANT AND UNAVOIDABLE ENVIRONMENTAL IMPACTS

Implementation of the project-specific mitigation measures identified in the Chapter 3 analysis would reduce all significant impacts to below a level of significance, with the exception of the following impacts:

Section 3.2, Air Quality

Impact 3.2-1

The proposed rezoning of the 5.25-acre portion of the project site has the potential to introduce a more intensive land use to the property, and this potential change was not taken into account in the most recent state ozone plan the Bay Area 2010 Clean Air Plan, adopted by the Board of Directors in September 2010. As described in Section 3.2, there is no feasible mitigation to reduce or avoid this impact; therefore, the impact would be **significant and unavoidable**.

Impact 3.2-2

The proposed project would result in an exceedance of the Bay Area Air Quality Management District (BAAQMD) NO_x threshold, which would conflict with the Clean Air Plan's goal of bringing the San Francisco Bay Area Air Basin into attainment for ozone since NO_x is a precursor to the development of ozone. Although implementation of MM-3.2-1 would reduce NO_x emission levels, it cannot be quantitatively determined whether emissions levels would be reduced to a level that is less than significant. As such, Impact 3.2-2 would remain **significant and unavoidable**.

Impact 3.2-4

The proposed project would result in a considerable contribution to a significant cumulative impact because it would exceed the BAAQMD threshold for NO_x emissions during project operations. Although implementation of MM-3.2-1 would reduce emission levels, it cannot be quantitatively determined whether emissions levels would be reduced to a level that is less than significant. As such, Impact 3.2-4 would remain **significant and unavoidable**.

Impact 3.2-5

The proposed rezoning of the 5.25-acre portion of the project site has the potential to introduce a more intensive land use to the property, and this potential change was not taken into account in the most recent state ozone plan — the Bay Area 2010 Clean Air Plan, which would result in a cumulatively considerable impact. As described in Section 3.2, there is no feasible mitigation to reduce or avoid this cumulative impact; therefore, the impact would be **significant and unavoidable**.

Section 3.4, Cultural Resources

Impact 3.4-2

The proposed demolition of the flour mill, grain silos, and dock, and extensive new construction and site work would have a significant adverse effect on documented historic resources. Implementation of MM-3.4-2a and MM-3.4-2b would reduce the impact, but not to a less-than-significant level. Thus, the impact would remain **significant and unavoidable**.

Section 3.6, Greenhouse Gas Emissions

Impact 3.6-1

The proposed project would exceed the BAAQMD threshold for operational GHG emissions of 10,000 metric tons of carbon dioxide equivalent (MT CO₂E) per year. Implementation of MM-3.6-1 would require fuel supply measures to reduce GHG emissions associated with operation of the proposed

project; however, emissions would not be reduced to below a level of significance. Impact 3.6-1 would remain **significant and unavoidable**.

Impact 3.6-2

Although the proposed project would not directly conflict with or obstruct implementation of the City of Vallejo Climate Action Plan (CAP), because the City's adopted CAP does not cover marine and rail operations, it cannot be guaranteed that the proposed project would be consistent with the overarching objective of the City's CAP to achieve the reduction targets as established for 2020 and 2035. Implementation of MM-3.6-2a through 3.6-2d would require the applicants to encourage employee commute alternatives and reduce the amount of energy used for landscaping maintenance and irrigation. However, emissions would not be reduced to a level that would ensure the project would be consistent with the overarching objective of the City's CAP to achieve the reduction targets as established for 2020 and 2035.

Impact 3.6-3

Although the proposed project would not directly conflict with or obstruct implementation of the City of Vallejo CAP, it cannot be guaranteed that the proposed project would be consistent with the overarching objective of the City's CAP to achieve the reduction targets as established for 2020 and 2035, or the state's target reduction goals in 2030 and 2050. This is because the City's adopted CAP does not cover marine and rail operations, and therefore emissions cannot be assured of being consistent with the CAP. Implementation of MM-3.6-2a through 3.6-2d would require the applicants to encourage employee commute alternatives and reduce the amount of energy used for landscaping maintenance and irrigation. However, emissions would not be reduced to a level that would ensure the project would be consistent with the overarching objective of the City's CAP to achieve the reduction targets as established for 2020 and 2035, or the state's target reduction goals in 2030 and 2050. Impact 3.6-3 would remain **significant and unavoidable**.

Section 3.10, Noise

Impact 3.10-1

The increase in noise levels due to operation of the VMT project component would exceed established policies and standards at the following two locations:

- NSL5 (Colt Court Residences)
- NSL10 (3rd Street Residence)

Mitigation measure MM-3.10-1a would, if implemented, reduce these impacts; however, implementation is dependent on the California Northern Railroad since the City does not have

jurisdiction over the railroad. While the City can require the applicants to work with the California Northern Railroad to make these improvements, the City cannot ensure that the California Northern Railroad will agree to make the improvements. Therefore, Impact 3.10-1 would remain significant and unavoidable.

Impact 3.10-3

The VMT project component would not generate any significant groundborne vibrations as a result of its operations aside from vibration caused by rail operations as described previously under Threshold A. For rail operations, one of the major sources of noise and vibration would be rolling stock on the existing jointed track. Mitigation measure MM 3.10-1a would, if implemented, reduce these impacts; however, implementation is dependent on the California Northern Railroad since the City does not have jurisdiction over the railroad. While the City can require the applicants to work with the California Northern Railroad will agree to make these improvements. Therefore, Impact 3.10-3 would remain significant and unavoidable.

Section 3.12, Transportation and Traffic

Impact 3.12-2

The proposed project would cause substantial delays and queues at rail crossings (delays of over 1 minute during peak hours, or queues that block upstream intersections during the day and early evening when traffic volumes are at or near their peak hour levels) relative to delays and queues without the project. Mitigation measure MM-3.120-24 would, if implemented, reduce these impacts; however, implementation is dependent on the California Northern Railroad since the City does not have jurisdiction over the railroad. While the City can require the applicants to work with the California Northern Railroad to make these improvements, the City cannot ensure that the California Northern Railroad will agree to make the improvements. Therefore, Impact 3.12-2 would remain **significant and unavoidable**.

Impact 3.12-3

The proposed project would cause substantial delays and queues at rail crossings (delays of over 1 minute during peak hours, or queues that block upstream intersections during the day and early evening when traffic volumes are at or near their peak hour levels) relative to delays and queues in the Cumulative No Project condition. Mitigation measure MM-3.10-1a would, if implemented, reduce these impacts; however, implementation is dependent on the California Northern Railroad since the City does not have jurisdiction over the railroad. While the City can require the applicants to work with the California Northern Railroad to make these improvements, the City cannot ensure that the California Northern Railroad will agree to make the improvements. <u>MM-3.12-2b would</u>

be implemented to provide emergency service providers with the opportunity to plan alternative routing during emergencies; however, delays due to rail operations could still impact emergency evacuation routes. Therefore, Impact 3.12-3 would remain significant and unavoidable.

Impact 3.12-5

The proposed project would have a significant impact on emergency access, based on the potential delays generated by train crossings at the grade crossings in Vallejo, American Canyon, and crossings further north. Mitigation measure MM-3.120-21-a would, if implemented, reduce these impacts; however, implementation is dependent on the California Northern Railroad since the City does not have jurisdiction over the railroad. While the City can require the applicants to work with the California Northern Railroad to make these improvements, the City cannot ensure that the California Northern Railroad will agree to make the improvements. <u>MM-3.12-2b would be implemented to provide emergency service providers with the opportunity to plan alternative routing during emergencies; however, delays due to rail operations could still impact emergency evacuation routes.</u> Therefore, Impact 3.12-5 would remain **significant and unavoidable**.

5.3 SIGNIFICANT AND IRREVERSIBLE ENVIRONMENTAL EFFECTS

California Environmental Quality Act (CEQA) Guidelines mandate that the EIR must address any significant irreversible environmental changes that would be involved in the proposed action should it be implemented (CEQA Guidelines, Section 15126(c)). An impact would fall into this category if:

- The project would involve a large commitment of nonrenewable resources;
- The primary and secondary impacts of the project would generally commit future generations to similar uses;
- The project involves uses in which irreversible damage could result from any potential environmental incidents associated with the project; and/or
- The proposed consumption of resources is not justified (e.g., the project results in wasteful use of energy).

Determining whether the proposed project may result in significant irreversible effects requires a determination of whether key resources would be degraded or destroyed in such a way that there would be little possibility of restoring them.

Implementation of the proposed project would reestablish industrial uses on the project site, including construction and operation of a modern deep-water terminal that would be capable of handling a wide range of commodities, including construction materials and break-bulk items. In addition, the proposed project would result in the construction and operation of an industrial

facility for the production and export of ground granulated blast furnace slag (GGBFS) cement, a product which is intended to meet the needs of construction projects for cement with a substantially reduced associated carbon footprint compared to traditional portland cement products, and the import of the raw material precursors of that product. This process necessarily consumes limited, slowly renewable, and nonrenewable resources. Resources consumed in this process include fossil fuels burned for the production of electricity that would power the main milling equipment and natural gas burned in the process of drying GGBFS materials. As an industrial process, the operation of this facility would by nature be resource-intensive; however, as an alternative to conventional cement production, the finished product could result in potential carbon savings and emissions reductions.

The construction of this facility would require the demolition of existing structures and the subsequent use of construction supplies including certain types of lumber and other forest products; aggregate materials used in concrete and asphalt such as sand, gravel and stone; metals such as steel, copper, and lead; petrochemical construction materials such as plastics; water; and fossil fuels such as gasoline and oil. All of these resources are frequently used in most general construction processes and are potentially nonrenewable.

The implementation of this project would commit future generations to the use of this site for the industrial production of GGBFS and or cement for the foreseeable future. As it stands, the site is already developed for industrial purposes and is in a state of vacancy and disrepair. The proposed project would not alter the use of the terrestrial site other than to utilize the currently unoccupied industrial space and erect one small building on a currently undeveloped portion of the property. The re-construction of the marine terminal would commit current and future generations to the use of the site as a terminal for bulk carrier ships delivering raw materials for the production of GGBFS and or cement. So long as the facility continues to operate, the area would experience increased traffic from bulk carrier ships delivering raw materials and transporting finished products from the facility. As described in Section 3.3, the benthic marine environment in the vicinity of the proposed marine terminal is not considered to be high value habitat for any sensitive or special-status aquatic species and fits predominantly within the footprint of the current decomposing General Mills wharf.

The project is not expected to result in any wasteful use of energy, as discussed in greater detail in Section 3.13, Utilities and Service Systems. The proposed project would be dependent on optimizing production and thus would have a vested interest in maximizing the efficiency of its use of resources.

5.4 GROWTH INDUCEMENT

CEQA requires a discussion of ways in which the proposed project could induce growth. The CEQA Guidelines identify a project as growth inducing if it fosters economic or population growth, or the construction of additional housing, either directly or indirectly in the surrounding

environment (CEQA Guidelines, Section 15126.2[d]). New employees from commercial or industrial development and new population from residential development represent direct forms of growth. These direct forms of growth have a secondary effect of expanding the size of local markets and inducing additional economic activity in the area. A project could indirectly induce growth by reducing or removing barriers to growth or by creating a condition that attracts additional population or new economic activity. However, a project's potential to induce growth does not automatically result in growth. Growth can only happen through capital investment in new economic opportunities by the private or public sectors. Under CEQA, growth inducement is not considered necessarily detrimental, beneficial, or of little significance to the environment.

The proposed project does not include any residential development nor would it expand infrastructure in a way that facilitates future growth. The project would utilize an existing but currently non-operational site, already zoned and developed for industrial use, using the existing public utilities and infrastructure. The project is therefore not expected to directly induce growth by creating new housing, commercial, or industrial developments.

The proposed project is expected to generate jobs; the Orcem Plant estimates 100 jobs during the 15-month construction phase and up to 40 full-time jobs during operation. The VMT component of the project estimates 25 full-time jobs during regular daily operation, and up to 40 jobs during vessel loading and unloading periods. The generation of these new jobs could be considered indirectly growth inducing; however, a high demand for those skilled jobs exists within the City of Vallejo. As of 2010, approximately 3,184 Vallejo residents commuted out of the City to work in a manufacturing industry. About of one-third of these workers are in production occupations. Additionally, approximately 2,700 Vallejo residents commuted outside the City to work in the transportation and warehousing industry, including 61% in materials moving occupations such as truck drivers and ship packers (City of Vallejo 2012). Given the high number of Vallejo residents commuting outside the City for manufacturing and transportation/warehousing jobs, it is anticipated that the jobs generated as a result of the proposed project could be filled by existing Vallejo residents without resulting in growth from an influx of labor. The temporary spike in jobs during the construction phase is not expected to be growth inducing because of the short duration and temporary nature of those jobs.

This project could be considered to promote economic growth as it is likely to expand local markets and induce additional economic activity in the area through the import of raw materials for the production and export of "green cement." This effect is considered an indirect effect on growth.
INTENTIONALLY LEFT BLANK

6.1 INTRODUCTION

Pursuant to the California Environmental Quality Act (CEQA) Guidelines, Environmental Impact Reports (EIRs) are required to "describe a range of reasonable alternatives to the project, or to the location of the project, which would feasibly attain most of the basic objectives of the project but would avoid or substantially lessen any of the significant effects of the project, and evaluate the comparative merits of the alternatives" (14 CCR 15126.6(a)). An EIR "must consider a reasonable range of potentially feasible alternatives that will foster informed decision making and public participation" (14 CCR 15126.6(a)). The alternatives discussion is required even if these alternatives "would impede to some degree the attainment of the project objectives or would be more costly" (14 CCR 15126.6(b)).

The inclusion of an alternative in an EIR does not constitute definitive evidence that the alternative is in fact "feasible." The final decision regarding the feasibility of alternatives lies with the decision maker for a given project who must make the necessary findings addressing the feasibility of alternatives for avoiding or substantially reducing a project's significant environmental effects (California Public Resources Code Section 21081; see also 14 CCR 15091).

Following is a list and summary of the alternatives to the VMT and Orcem Project which have been considered in this chapter of the EIR:

- **1.** Alternate Site: A possible alternative location to implement the same project. This alternative was found to be infeasible and was therefore not further evaluated.
- 2. **Preservation Alternative:** A possible alternative in which the historic resources on the project site are structurally repaired and reused. This alternative was found to be infeasible and was therefore not further evaluated.
- **3. Reduced Truck and Rail Alternative:** A possible alternative in which 40% of the project's truck and rail volumes are shifted to barges. This alternative was found to be infeasible and was therefore not further evaluated.
- **4. Reduced Scale Alternative:** A possible alternative in which the volume of goods and materials processed through the Terminal and produced by Orcem are reduced by 25%. This alternative was found to be infeasible and was therefore not further evaluated.
- 5. No Project Alternative: An alternative in which the project site would remain in its current condition and no construction would occur. As required under CEQA, this alternative has been analyzed below.
- 6. Revised Operations Alternative: An alternative in which a series of plan refinements and operational changes to both the Orcem and VMT components are implemented focusing on retaining project feasibility while reducing environmental impacts.

6.2 **PROJECT OBJECTIVES**

The primary objectives of the proposed project are set forth in Chapter 2, Project Description, of the EIR and consist of the following:

- Establishment of the VMT Terminal as a key site of multi-modal and intermodal transportation and logistics, thereby enhancing Vallejo's role in the regional and international trade economy and providing a means for locally manufactured products to be transported and distributed, increasing the viability of and the potential for attracting further manufacturing operations to Vallejo.
- Maximize the potential for the manufacture of ground granulated blast furnace slag (GGBFS), a product that helps to meet the needs of the construction industry for high-performance, environmentally favorable concrete and sustainable building materials, by providing for an efficient scale of production at a plant which would operate around the clock as a multi-modal receiving, storage, processing, and distribution facility.
- To provide management and skilled labor employment opportunities for local and regional residents in the construction phases, as well as the long-term operations of commercial and industrial uses on the project site.
- To generate various tax revenues including property taxes and assessments, possessory interest tax, and utility user fees.
- To reestablish and optimize the industrial use of this centrally located marine industrial property through removal of those remaining components of the severely damaged timber wharf and construction of a modern deep-water terminal.
- To maximize accommodations for shipping and receiving of a wide range of products through the VMT Terminal, including loading and unloading of vessels, including deep draft vessels and barges, of up to 70,000 metric tons (MT) in size with draft of up to 38 feet through the restructured project. The improvements would help to further develop Vallejo's capabilities for water-based shipping.
- To maximize throughput capacity through the implementation of intermodal upgrades designed to optimize cargo handling operations as well as modern design initiatives enabling the most efficient use of the ground area and taking advantage of existing truck, rail, and shipping access for import and export of raw materials and finished products.
- To reliably provide competitively priced and environmentally preferable cement products and offer GGBFS and non-GGBFS cementing products, in order to provide a complete line of competitive products that meet long-term client and project needs, and to have the ability to respond to potential worldwide shortages of GGBFS supplies, thereby assuring sustainability of Orcem's operation over time.

• To follow the federal Short Sea Shipping Highway Initiative where possible by focusing on short sea shipping opportunities that move cargo by coastal and inland waterway barges, reducing both truck and rail environmental impacts.

6.3 ALTERNATIVES CONSIDERED BUT REJECTED

An EIR must briefly describe the rationale for selection and rejection of alternatives. The lead agency may make an initial determination as to which alternatives are potentially feasible, and therefore merit in-depth consideration, and which are not feasible. Alternatives whose implementation is remote or speculative, or the effects of which cannot be reasonably predicted, need not be considered (CEQA Guidelines, Section 15126.6(f)(3)). Factors that may be considered when addressing the feasibility of an alternative include site suitability, availability of infrastructure, general plan consistency, other plans or regulatory limitations, jurisdictional boundaries, economic viability, and whether the proponent can reasonably acquire, control, or otherwise have access to the alternative site.

6.3.1 Alternate Site

Alternate locations for the project site were considered; however, the applicants do not own any waterfront property in the area that would be suitable for the proposed project. The project site was selected by VMT on the basis of its unique capability to accommodate deep-water berthing for vessels of up to 70,000 tons, in combination with rail and truck access, with sufficient space for transloading of goods and materials as described in Chapter 2.0, Project Description. The project site was previously used for water-related industrial uses, with access to deep-water shipping, interstate highways, and rail infrastructure. The proposed project would reestablish this water-related industrial use.

This combination of functional amenities suitable for operation of both the VMT and Orcem project components is not easily accommodated in other Bay Area sites. Since VMT currently owns a majority of the 31.4-acre project site and leases a portion of the project site from the City of Vallejo (City), and Orcem is sub-leasing their 4.88 acre portion of the site VMT, it is not feasible for the applicants to reasonably acquire another site with comparable amenities for the proposed project. An alternate site alternative is therefore infeasible, and is not evaluated further in this EIR.

6.3.2 **Preservation Alternative**

The Preservation Alternative would protect the historic characteristics of the project site by complying with the Secretary of the Interior's Standards for Rehabilitation. Section 15064.5(b)(3) of the CEQA Guidelines states, "Generally, a project that follows the Secretary of the Interior's Standards for the Treatment of Historic Properties with Guidelines for Preserving, Rehabilitating, Restoring, and Reconstructing Historic Buildings or the Secretary of the Interior's Standards for

Rehabilitation and Guidelines for Rehabilitating Historic Buildings ... shall be considered as mitigated to a level of less than a significant impact on the historical resource."

The grain silo is a concrete structure that would require extensive structural and infrastructure improvements for either reuse as a storage facility or other adaptive use. The flour mill is a reinforced concrete building with brick infill panels and veneer that would have to be brought up to current code standards. The State Historical Building Code could be used to address some of the codes issues, but there would likely be a need for extensive seismic strengthening of the buildings. Structural work, accompanied by mechanical, electrical, and plumbing upgrades for the Mill and Silo buildings, would be needed prior to or in conjunction with any kind of tenant improvements. Orcem would not reuse any portions of the grain silos and flour mill for their proposed mill building, filter building, workshop control room, and storage silos because the functional and structural requirements associated with the production of GGBFS would make reuse infeasible. The existing buildings occupy the majority of the Orcem Site and would preclude use of these areas for accommodation of necessary equipment, finished material storage, and raw material storage. The proposed raw material storage areas would be necessary for project operations. There is not sufficient space within the Orcem Site to accommodate the mill, outload silos, raw material storage areas, and other proposed GGBFS Mill improvements without demolishing the grain silos and flour mill. Therefore, the Preservation Alternative would not be feasible for the Orcem project component.

The remnants of the existing dock would not lend themselves to rehabilitation due to their deteriorated condition. To serve as a functioning wharf, removal of the 440 decaying creosote pilings and construction of a modern reinforced concrete wharf is necessary. In order to accommodate deep water shipping and a water-related industrial use, as called for in the project description, reconstruction would require a larger berthing area and a reinforced concrete wharf structure to accommodate modern vessels that would transport materials to and from the site. Although it may be possible for VMT to construct new wharves on either side of the existing dock, this option would: (1) Be problematic because of the relationship to the Mare Island Strait, designated shipping lanes, water currents, and insufficient water depth; (2) Conflict with the functional operation of a modern deep water marine terminal, including the need to accommodate vessels of up to 70,000 MT; and (3) Require a substantial additional amount of bay fill, which would increase impacts to marine biological resources and may conflict with policies in the San Francisco Bay Plan.

It would not be feasible for the VMT project component to move forward under the Preservation Alternative, because VMT is dependent on removal and replacement of the severely damaged wooden piles and deck remnants with a modern wharf capable of accommodating deep-water vessels of up to 70,000 MT capacity. In addition, the need for site grading and reuse of materials recycled from the mill and silo structures to complete the terminal operations portion of the VMT

Site would not be feasible under this alternative. While the VMT component could move forward with a different tenant for the 4.88-acre portion of the site proposed for use by Orcem, elimination of this key project component would eliminate a substantial portion of the shipping volume currently relied upon in determining the feasibility of the VMT operations. Additionally, it is unlikely that the existing buildings proposed for demolition could be used by another manufacturer. As a result, the Preservation Alternative is therefore infeasible, and is not evaluated further in this EIR.

6.3.3 Reduced Truck and Rail Alternative

The objective of the Reduced Truck and Rail Alternative is to decrease potential project impacts related to noise, traffic, and greenhouse gas (GHG) emissions and air quality. The Reduced Truck and Rail Alternative would develop the project site in a similar manner as the proposed project; however, the operations would be altered by shifting the mode of import and export of materials away from truck and rail and toward barges in all phases of the project. This alternative would replace approximately 40% of the truck and rail transport with barge transport. If this alternative were implemented, it would reduce truck and rail volumes, but would not eliminate the significant noise and vibration impacts identified in connection with the proposed project. Reduction in truck and rail volumes would further reduce the identified less-than-significant traffic impacts associated with truck transport; however, it would not eliminate the significant traffic impacts associated with temporary roadway closures at rail crossings identified in connection with the project. Finally, a Reduced Truck and Rail Alternative would further reduce the less-than-significant air quality impacts associated with the emissions of criteria pollutants resulting from trucks and rail traffic with the proposed project, but it would not result in a reduction of NO_x emissions to a less-than-significant level.

The Reduced Truck and Rail Alternative would not meet the basic project objectives because: (a) It would preclude accommodation of the same level of throughput as the proposed project, (b) It would interfere with the critical market-driven operations of both the VMT and Orcem components of the project, thereby precluding feasibility; and (c) The Reduced Truck and Rail Alternative would preclude the necessary shipping of Orcem's finished products via truck and rail, as it is in the proposed project. Because the majority of Orcem's primary markets are in the inland areas and are only accessible via truck and rail, this alternative would not be feasible, as it would prevent the Orcem component from operating competitively. While the VMT operations may be able to incentivize shipment of goods via barge over time, a 40% reduction in rail and truck volumes would interfere with market contracts that are only accessible by rail and truck. As a result, the Reduced Truck and Rail Alternative would prevent development and operation of the Terminal. This, in turn, would preclude both VMT and Orcem components. As a result, the Reduced Rail and Truck Alternative is therefore infeasible, and is not evaluated further in this EIR.

6.3.4 Reduced Scale Alternative

The objective of the Reduced Scale Alternative is to decrease potential project impacts related to noise, traffic, and GHGs and air quality. If it were feasible, the Reduced Scale Alternative would develop the project site in a similar manner as the proposed project; however, the volume of materials moved through the VMT Terminal and Orcem Plant would both be reduced by 25% compared to the proposed project. This reduction in maximum operating capacity would affect the VMT component by reducing the maximum monthly shipping cargo volume at the terminal from 160,000 to 120,000 MT. It would also reduce the maximum annual production volume for the Orcem Plant from 900,000 to 675,000 MT. This alternative would reduce the total average monthly number of vessels expected to arrive at the VMT Terminal (as shown in Table 2-3 in the Project Description) from 7.5 to 6 vessels. The ratios of distribution for finished products and goods by truck, rail, and barge would be expected to remain unchanged, with proportionate volume reductions in all modes of transportation. Similar to the Reduced Truck and Rail Alternative, were the Reduced Scale Alternative implemented, it would lead to across-the-board reductions in operational volumes, and would further reduce noise and traffic impacts associated with truck and rail transport for the proposed project. Traffic impacts associated with temporary roadway closures at rail crossings under a 25% reduction in rail traffic would continue to be significant and unavoidable, as the duration of roadway closures would remain unchanged from the proposed project. The Reduced Scale Alternative would also reduce air quality impacts associated with the emissions of criteria pollutants resulting from trucks and rail traffic, and would proportionately reduce GHG emissions; however, GHG and certain criteria pollutant emissions levels would remain significant and unavoidable, as in the case of the proposed project.

The Reduced Scale Alternative would preclude Orcem's ability to ensure the revenue required to justify: (1) The high level of fixed capital cost associated with construction of the plant; (2) The high fixed acquisition costs of equipment and operating systems involved in the state-of-the-art production and handling of GGBFS which also satisfies the Best Available Control Technology mandates of the Bay Area Air Quality Management District (BAAQMD); and (3) The high costs of operating the plant which are relatively inelastic with respect to scaling back of output volumes, making this component of the project infeasible. The Reduced Scale Alternative is therefore infeasible for the Orcem component of the project.

In addition, the Reduced Scale Alternative would also preclude the feasibility of VMT to construct and operate the Terminal because of: (1) Very high initial fixed capital costs associated with demolition of the existing flour mill, silos, and wharf; initial dredging for deepwater accessibility to accommodate larger vessels; and overall site preparation and construction of the wharf; (2) On-going fixed costs involving maintenance dredging, and terminal and equipment maintenance; and (3) The need to achieve a level of throughput scale required to support operation of barges as well as larger ocean-going deep-water vessels with access to international and local markets. The 25% reduction in

production and throughput volumes, and therefore efficiency, as reflected in the Reduced Scale Alternative therefore makes this alternative to the proposed project infeasible for construction and operation of the VMT component, and consequently also for the Orcem component. As a result, the Reduced Scale Alternative is infeasible, and is not evaluated further in this EIR.

6.4 ALTERNATIVES ANALYSIS

This section discusses two alternatives to the proposed project, including the No Project Alternative. The No Project Alternative is a required element of an EIR pursuant to Section 15126.6(e) of the CEQA Guidelines that examines the environmental effects if the project were not to proceed. The Revised Operations Alternative (ROA) is also discussed, as part of the "range of reasonable alternatives" as the only other meaningful alternative to the proposed project which could result in substantial reductions in project environmental impacts, while achieving most of the basic objectives of the project, including achieving a level of economic feasibility.

The consideration of a range of potentially feasible alternatives in this EIR was limited based on the following facts: (a) A substantial portion of this site is within the Public Trust Tidelands Area, and pursuant to State Lands Commission requirements imposed on the City. Since 1913, several grants of state tide and submerged land have been made to the City of Vallejo to administer in compliance with the Public Trust Doctrine and for the benefit of the statewide public. The Commission is statutorily required to oversee the management of sovereign public trust lands and assets by legislative grantees who manage these lands on behalf of the state. These lands are therefore limited to use for maritime industrial activities, commercial activities, and recreational, and open space; and (b) The site also contains a large area which was subject to clean-up of hazardous materials from the former mill operations use, as well as a closure plan, which preclude use of the site for most types of residential, playgrounds, child care, or other uses where exposure would otherwise create safety concerns. Additional alternatives, including the Reduced Scale Alternative, capable of substantially reducing certain environmental impacts of the proposed project, are identified and discussed in Chapter 6.3; however, these alternatives have been dismissed because they would preclude project feasibility, and would therefore have the same outcome as the No Project Alternative.

6.4.1 No Project Alternative

Under the No Project Alternative, the project site would remain in its current condition. No buildings or structures would be demolished, and no construction of new buildings or structures would occur. The existing wharf structures would also remain, and no dredging or filling of Mare Island Strait would occur. The project site would remain vacant and no new operations would occur. The No Project Alternative would not meet any of the project objectives since the site would remain unchanged. The site would not generate new employment opportunities or increased tax

revenues. The site would not become a marine terminal and would not provide for the production of GGBFS; therefore, the objectives related to maximizing the capabilities of the site for shipping and GGBFS production would not be achieved under this alternative.

6.4.2 Revised Operations Alternative

Under the Revised Operations Alternative, the overall operations of the proposed project would be modified to decrease potential project impacts related to air quality, GHG emissions, noise, transportation and traffic. The Revised Operations Alternative would develop the project site in an identical manner as the proposed project; however, the operation of some project components would be altered with the resulting reductions in impacts, as outlined below. It should be noted that should this alternative be approved, each of these components would be included in the MMRP and/or be included as a condition of approval. The alterations to the project include:

- Revised Orcem Truck Loading & Weight Confirmation System: Through the use of vehicles that are capable of carrying slightly larger loads, this ROA element improves the efficiency of Orcem tanker trucks leaving the site by 4% from 22.68 to 23.59 MT, thereby reducing the number of project trucks trips by less than eight trips and reducing NO_x, CO₂, and PM_{2.5}/PM₁₀ emissions by a comparable percentage. This change reduces the maximum average number of daily truck trips from 189 to 181 trucks per day.
- 2. Enhanced Orcem Truck Scheduling Efficiency and Operations and Increased Trucking Days: Shipping of finished products by Orcem would be more efficiently spread out on a monthly basis, utilizing an average of 26.0 trucking days per month (six days per week) in comparison to the 17.5 trucking days per month (five days per week) utilized in the Original Project. This component of the ROA project would further reduce the number of average daily trucks leaving the Orcem Mill, in comparison to the proposed project. The revised scheduling reduces daily truck trips under Orcem Mode 2 / Milestone 5 operations (when clinker is ground to produce cement) from 181 identified above to 122. However, the number of days the trucks run would increase from five days per week to six days per week. A 10% variation in Orcem's daily truck trips under Mode 2 / Milestone 5 would range be between 110 and 134 trips. With the Loading & Weight Confirmation System and the Scheduling Efficiency the maximum number of daily truck trips for VMT and Orcem would be 21 round trips (87 trips + 134 trips = 221 trips) or 442 one-way daily trips.¹

¹ These truck trip counts represent the highest impact scenario and were therefore used for purposes of impact analysis. It is possible that lower trip counts could occur under Orcem operating Modes 2 and 3, both with and without rail, translating to the following combined Orcem plus VMT figures: (1) 202 total trucks in Mode 2 without rail (115 Orcem + 87 VMT=202 trips or 404 one-way daily trips); (2) 119 total trucks in Mode 2 with rail (115 Orcem + 4 VMT=119 trips or 238 one-way daily trips); (3) 183 total trucks in Mode 3 without rail (96

- 3. **Reduced Train Lengths:** To reduce impacts on emergency services and traffic at railroad/road intersections from long trains, the maximum length of trains serving the project site is reduced from 77 cars in the proposed project, to 50 cars in this alternative for both Orcem and VMT.
- 4. **Tier 4 Equipment:** VMT and Orcem will use Tier 4 construction equipment (new technology which reduces exhaust gases from diesel powered equipment) for all land-based construction activities (excluding wharf pile drivers and diesel hammer).
- 5. Elimination of Late Night Orcem Operations: As part of the ROA, Orcem will eliminate late night (between 12:00 midnight and 6:00 a.m.) operations within 300 feet of the nearest residential boundary. This restriction's effect on noise levels has not been separately quantified in this report; the Noise Chapter analyzes the maximum noise effects of the project without the additional benefit of this ROA component.
- 6. **Reduced VMT Trucking:** The original project envisioned a large dike in addition to the wharf which is included in both the project and the ROA. This dike would have enabled additional whipping and would have created a large laydown area for the management of bulk and break bulk goods. This dike was removed from the project to lessen several significant impacts associated with the project. However, due to removal of the dike, space constraints would reduce VMT's ability to load material by truck and rail simultaneously for material that exceeds 12 VMT ships (480,000 tones/yr). Therefore, in order to reach the terminal's maximum capacity of 29 VMT ships per year, rail will need to be operational. Once rail is operational VMT will switch to using rail solely (no trucks) to export goods from the site, thereby eliminating 83 one-way truck trips per day. There will continue to be up to four delivery trucks serving the site on a daily basis.

Supplemental Landscape Screening for Orcem Mill Operations: Stockpiles and equipment within the southerly portion of the Orcem site were partially visible under the project, even with installation of the planned 6-foot perimeter fence. This was not considered a significant visual impact in the DEIR. Under the ROA the landscape plan for this portion of the site has been modified. Additional trees and shrubbery will run along the west and south sides of the Material Storage Area. The additional landscaping will serve to further screen the open material stockpile and equipment in the background, from the south and west.

A Fleet & Equipment Management Plan to commit to the use of newer technologies (than currently required pursuant to BAAQMD standards) on trucks, front-loaders and other powered equipment was proposed by the applicant but not produced in time for analysis in this document. Similarly the applicant discussed development of a Barge Preference Implementation Strategy that would be used by VMT to actively market to and select users of barges over users exclusively relying on

Orcem + 87 VMT=183 daily trips or 366 one-way daily trips); and (4) 80 total trucks in Mode 3 with rail (76 Orcem + 4 VMT=80 daily trips or 160 one-way trips).

trucks and rail, during periods when the Terminal is not operating at maximum capacity. This Strategy was also not produced in time for analysis in this document. While these elements could be produced in the future, their implementation has not been quantified and therefore has no effect on the project's expected impacts as analyzed in this report.

The Revised Operations Alternative would meet all of the basic objectives of the proposed project, since it accommodates the same level of maximum throughput and, unlike the Reduced Truck and Rail Alternative ensures access to critical markets. Like the proposed project, the Revised Operations Alternative increases employment opportunities and tax revenues for the City. The Revised Operations Alternative was designed to accomplish the following:

- 1. Maintain economic feasibility by not substantially altering the volume of production or throughput for either the Orcem Mill or the VMT Terminal, including the objective of maximizing the potential for the milling of ground granulated blast furnace slag (GGBFS);
- 2. Fulfill other basic objectives of the project, including: (a) establishment of the VMT Terminal as a key site of multi-modal and intermodal transportation and logistics, thereby enhancing Vallejo's role in the regional and international trade economy; and (b) providing a means for locally manufactured products to be transported and distributed, increasing the viability of and the potential for attracting further manufacturing operations to Vallejo.
- 3. Reduce some of the potentially significant environmental effects of both proposed project components by implementing a coordinated set of related and interdependent alterations to the manner in which the project as a whole would operate.

Community Benefits Package: The project applicants have submitted a Community Benefits package that includes a one-time \$1 million Community Grant Program where \$200,000 of grant funds would be available for five years. The funds are intended to support job training, education, youth services and environmental sustainability.

6.4.3 Aesthetics

With the exception of additional landscape screening (trees and shrubbery) along the west and south sides of the Material Storage Area (see Figures 6-1 and 6-2) within the Orcem Site, the ROA would involve identical physical changes to the project site as the proposed project, including planned demolition of existing on-site structures and construction of identical facilities for VMT and Orcem. Views of and from the project site would be similar to the proposed project with similar potential for impacts due to lighting (reduced through mitigation). The ROA would also provide more landscape screening of the Material Storage Area compared to the proposed project.

The ROA could result in an increase in ship traffic and the number of days vessels would be docked at the VMT Terminal due to the potential shift away from trucks and trains for material transport.

However, the presence of a small number of additional ships and corresponding trucks on additional days would not be considered a significant impact to aesthetics since ships, boats, trucks, and trains are frequent users of the waters, roadways, and rail that surround the project site, and therefore would not detract from the existing visual character or quality of the site. For these reasons, aesthetic impacts from the ROA would be similar to aesthetic impacts from the proposed project.

6.4.4 Air Quality

This section evaluates the potential construction and operational impacts of the Vallejo Marine Terminal (VMT) and Orcem California Inc. (Orcem) ROA with respect to air quality impacts, and includes mitigation measures required to mitigate the original proposed project (in Chapter 3 of this document) where necessary to reduce or avoid significant impacts.

Because reductions in air quality impacts are the most noteworthy change when comparing the ROA to the original proposed project, this section describes the ROA analysis in much greater detail than other areas of potential impact. A comprehensive summary of the analysis is found below, followed by a more detailed analysis that corresponds in format and elements analyzed to the impact discussion found in Section 3.2 of this document. Information provided in this section is based on analysis done for the original proposed project and a technical study prepared for the ROA found in the following appendices:

Appendix D-1: Ramboll Environ. 2015. Orcem/VMT Project – Air Quality and Greenhouse Gas Evaluation.

Appendix M: Ramboll Environ. 2018. VMT/Orcem Revised Operations Alternative Air Quality and Health Risk Assessment, Vallejo, CA.

Details regarding methodology, emissions calculations and model outputs can be found in Appendix D-1.

6.4.5 Summary of ROA Air Quality Analysis

In June 2018, Ramboll, VMT/Orcem's air quality consultant, analyzed the Revised Operations Alternative (ROA) for the VMT/Orcem project. Whereas the Project Alternative (Project) had been quantitatively analyzed in the 2017 Draft Final Environmental Impact Report (DFEIR), the ROA had only been assessed qualitatively.² Ramboll's 2018 ROA analysis quantifies ROA impacts by building on the calculation methodology used in DFEIR analysis and revises the methodology, assumptions, and emission factors to better reflect current industry practices.

² The DFEIR was based on analyses prepared by AWN and Ramboll, and submitted by the applicant.

This following summarizes the revised ROA assumptions, revised calculation methodology, revised emission factors, and results of the 2018 ROA analysis and compares the 2018 ROA analysis results to the project impacts described in the DFEIR.

The 2018 ROA is different from the DFEIR Project Alternative as follows:

ROA Construction Components

• The 2018 ROA proposes to use equipment meeting Environmental Protection Agency (EPA) Tier 4 off-road engine emission standards for all land-based construction activities (excluding wharf pile drivers and diesel hammers). This measure reduces criteria pollutant emissions as compared to the DFEIR Project Alternative.

ROA Operational Components

- The 2018 ROA proposes a Revised Orcem Truck Loading & Weight Confirmation System, which increases truck capacity, thereby reducing the number of trucks associated with Orcem operation by 4%.
- The 2018 ROA proposes an Enhanced Orcem Truck Scheduling Efficiency System, which reduces the average daily number of Orcem trucks from 189 to 122 by increasing the average monthly trucking days from 17.5 to 26. This measure, on average, increases trucking days from five days per week to six days per week.
- The 2018 ROA proposes to reduce train lengths from 77 rail cars per train to 50. This ROA component reduces the time required for trains to cross local city streets but increases the number of trains required to transport product.
- Ramboll proposed Revised Project-Sponsored Technology Upgrades to ensure the ROA avoids significant air quality and Health Risk Assessment (HRA) impacts. The specific technologies are presented in Table 6 of the Orcem-VMT ROA Summary Report for the HRA (see Appendix M). HRA mitigation measures that are included in the 2018 ROA are based on the Revised Project-Sponsored Technology Upgrades.

ROA Maximum Combined Scenario

The 2018 ROA analyzes various operating scenarios and concludes that maximum impacts, except cancer risk, would occur when (1) Orcem operates at Mode 3 / Milestone 5 (producing maximum Ground Granulated Blast Furnace Slag (GGBFS) and blended cement products), using 19 (average 40,000 MT) annual vessel shipments of raw materials, where processed materials are loaded into trucks and rail for distribution; and (2) VMT operates using 29 annual vessel shipments, where all imported goods are loaded onto rail for distribution. This is referred to as the Maximum Combined Scenario in the 2018 ROA Analysis. In the 2018 ROA maximum cancer risk occurs when the Orcem Mill operates

at Mode 2 / Milestone 5 (producing cement products only) and VMT operates as described above. The 2018 ROA analysis concludes that air emissions and health impacts would not exceed BAAQMD California Environmental Quality Act (CEQA) thresholds following mitigation.

ROA Revisions and Impacts: Construction

The 2018 ROA quantifies diesel particulate matter (DPM) emissions and associated cancer risk due to construction. In comparison to the DFEIR, the 2018 ROA construction DPM emissions are reduced by approximately 89% and the corresponding construction cancer risk is reduced by approximately 80%. These reductions are due to the following revisions in the ROA:

- The ROA assumes the use of Tier 4 diesel equipment for all construction equipment, except pile drivers and diesel hammer, as a component of construction. Tier 4 engines are significantly cleaner than lower tier engines. In addition, the elimination of activities associated with VMT Phase 2 construction (i.e., construction of an expanded laydown area that was eliminated from the Project Description) from the ROA also reduces construction emissions. These revisions result in a DPM decrease from 0.301 tons to 0.032 tons, a decrease of 0.27 tons of DPM compared to the DFEIR.
- Although the 2018 ROA does not quantify criteria pollutants associated with construction, it is unlikely that these emissions would exceed BAAQMD significance thresholds because they would be lower than the DFEIR emissions, which did not exceed BAAQMD significance thresholds. Criteria pollutant emissions would be reduced in the ROA compared to the DFEIR for the reasons identified above.
- The 2018 ROA analysis uses the 2015 Office of Environmental Health Hazard Assessment (OEHHA) Guidance to quantify health impacts, which is more stringent than the methodology used in the DFEIR. Specifically, the 2018 ROA incorporates increased daily breathing rates per the 2015 OEHHA guidance. Although this revision increases cancer risk associated with construction activities, the increase is offset by reduction in DPM emissions described above.

ROA Revisions and Impacts: Operation

The following discusses 2018 ROA changes in activity, assumptions and the resulting air quality impacts compared to the DFEIR. Exhibits 6-3 and 6-4 show the contribution of each source category to annual operational NO_x and PM_{10} emissions, respectively, for the 2018 ROA and DFEIR.³ Figure 6-3 shows that NO_x emissions are approximately 34% lower in the 2018 ROA compared to the DFEIR. Figure 6-4 shows that PM_{10} exhaust emissions do not change appreciably.

³ Numbers may not add exactly due to round off.

Although not shown in the figures below, the 2018 ROA analysis shows that VOC emissions are approximately 14% lower in the 2018 ROA compared to the DFEIR.





Exhibit 6-4 Source Contributions Annual PM₁₀ Emissions



On-Road Truck Impacts

In comparison to the DFEIR, the 2018 ROA reduces 1-way truck trips from approximately 81,058 to 49,152 per year. The reduction in truck trips in combination with the items discussed below reduce NO_x and PM_{10} truck emissions by approximately 73% and 17%, respectively, for the Maximum Combined Scenario. These reductions are due to the following:

- Orcem Revisions
 - The 2018 ROA analysis subtracts material that would be transported via rail from total truck trips. This had not been accounted for in the AWN report submitted by the applicants and was therefore not included in the DFEIR. This revision reduces truck trips.
 - The 2018 ROA proposes an ROA condition, Orcem Truck Loading & Weight Confirmation System, which increases the capacity of Orcem trucks from 20 to 23.58 tones. This revision decreases truck trips because fewer trucks with larger capacity would be needed to transport the same amount of product.
 - The 2018 ROA proposes an ROA condition, The Enhanced Orcem Truck Scheduling Efficiency System, which reduces the average daily number of trucks from 189 to 122 by increasing the average monthly trucking days from 17.5 to 26. Per the transportation analysis, this increase in monthly trucking days would correspond to an increase from 5 to 6 days per week.⁴
- VMT Revisions
 - The 2018 ROA eliminates construction of VMT Phase 2. Without construction of VMT Phase 2, which had expanded the laydown area for the bulk and break bulk materials delivered to the site, space constraints preclude VMT from loading material by truck and rail simultaneously, for material throughput above 12 VMT vessels (480,000 MT per year). Although, rail and truck loading could occur sequentially, rail and trucks could not be loaded simultaneously, in the same month. The reason for this is that during the truck transloading process, the terminal would be fully occupied for one full month and another vessel would be unable to unload to rail until the terminal is cleared due to lack of transloading space. Transloading to trucks one month and rail the next month could occur during years of lower throughput, but this practice would not be possible at the maximum scenario of 29 VMT vessels, which is when maximum ROA impacts are calculated to occur. Therefore, the 2018 ROA assumes that all material, for operation above 12 ships per year is transported by rail.

⁴ DFEIR, Appendix L, Fehr and Peers. 2016. *Transportation Technical Data*.

- The only trucks associated with VMT ROA activity in the Maximum Combined Scenario are 4 supply delivery trucks per day of terminal operation, operating for a maximum of 26 days per month.
- Revisions Common to Orcem and VMT
 - The 2018 ROA calculates the one-way truck transit distance of 41.9 miles, as described in the Orcem VMT one-way trip distance, included in Appendix M. The AWN report submitted by the applicants and consequently relied upon in the DFEIR incorrectly calculated a round-trip transit distance of 50 miles (25 miles one-way). Although this revision increases truck transit, the increase in emissions associated with longer transit is offset by other revisions discussed below.
 - The 2018 ROA revises the operational year to 2025 compared to the DFEIR 2020 operational year. This revision decreases truck engine emission factors because trucks in future years are expected, by CARB, to be cleaner per existing regulatory requirements.
 - The 2018 ROA uses speed-specific emission factors and aggregated emission factors. Speed-specific emission factors reflect a speed of 10 miles per hour (mph) for on-site transit and 20 and 40 mph for transit along surface streets. For freeway transit, the 2018 ROA uses emission factors that are based on aggregated speeds in the County.⁵ In comparison, the FDEIR used emission factors for 40 mph for all transit.

Shipping Impacts

The 2018 ROA does not reduce ship calls. However, in comparison to the DFEIR, the 2018 ROA results in a 24% and 7% reduction in NO_x and PM_{10} shipping emissions, respectively, for the Maximum Combined Scenario. These reductions are due to the following revisions:

 Revisions Common to Orcem and VMT - The 2018 ROA revises emission factors to reflect the "lower emission factors" and "low load adjustment factors" in the anticipated 2018 CARB shipping inventory database. In 2018, CARB plans to release a shipping inventory database that incorporates newer ship engine information than what was available at the time of the DFEIR, including lower emission factors and lower adjustment factors.

Rail Impacts

In comparison to the DFEIR, the 2018 ROA increases annual train trips from 114 (77-rail car trains) to 288 (50-rail car trains) and increases annual railcars from 11,421 to 14,393 (14,400 was assumed in the DFEIR Project Description). In combination with the items discussed below, these

⁵ Aggregated speed refers to a combined speed distribution that is developed by the CARB as part of the EMFAC database.

revisions contribute to an increase of over 100% in NO_x and PM₁₀ rail emissions for the Maximum Combined Scenario.⁶ These increases are due to the following:

- Orcem Revisions The 2018 ROA assumes that Orcem trains arrive full, bringing raw material to Orcem, offload the raw material, are loaded with finished product, and then leave full, transporting finished product from Orcem. This had not been accounted for in the AWN report, submitted by the applicants, which was used in the DFEIR and which assumed that (1) trains bringing raw material arrived full, offloaded the raw material, and left empty; and (2) trains transporting finished product arrived empty, were loaded with finished product, and left full. Because this information had not been included in the AWN report submitted by the applicants, it was not included in the DFEIR. Although the 2018 ROA assumption reduces the number of Orcem trains serving the facility because it eliminates empty train trips, the corresponding emissions reduction is offset by revisions described below.
- VMT Revisions As discussed under truck impacts, due to removal of VMT Phase 2, space constraints preclude VMT from loading material by truck and rail simultaneously, for material throughput above 12 VMT ships. Therefore, the 2018 ROA assumes that all material, for operation above 12 ships per year is transported by rail. This revision increases VMT rail trips.
- Revisions Common to Orcem and VMT
 - The 2018 ROA assumes shorter trains for both Orcem and VMT. Train length is reduced from 77 rail cars per to train in the DFEIR to 50 rail cars per train in the 2018 ROA to accommodate traffic considerations. This reduction increases the number of trains because twice as many shorter trains are needed to transport the same amount of material.
 - Based on information from CalNorthern, the rail line operator, the 2018 ROA assumes that 50 rail-car trains require the use of two locomotives per train for line-haul trains. This is a revision from the DFEIR, which assumed one locomotive for an empty train and three for a full train.
 - The DFEIR states that the National Railway Equipment Company would provide "ultra-low emission" equipment for switching activity, which at the time of the DFEIR was equivalent to EPA Tier 2 engines. Since the DFEIR was published, the NREC has begun to offer switcher locomotives that meet EPA Tier 4 engine standards. For this reason, the 2018 ROA assumes Tier 4 engines for switcher locomotives. This revision decreases emission factors and emissions associated with switching operations.
 - The DFEIR states that NREC would provide "ultra-low emission" equipment for line haul activity, which the DFEIR assumed to be primarily Tier 4 locomotives. The 2018 ROA

⁶ Please note that the DFEIR Project Description assumed that an average of 2.6 77-car trains would run per week.

revises the DFEIR assumption and instead, conservatively assumes a 2025 CARB fleet mix for line haul trains, which consists of Tier 2 through Tier 4 locomotives. This ROA revision increases emission factors and associated emissions from line haul locomotives.

On-Site Operations

On-site emissions would be generated by Orcem and VMT equipment. Orcem equipment includes stack, hopper and conveyors, bag filters, excavators, and front end loaders. VMT equipment includes front end loaders and forklifts. Both Orcem and VMT would generate fugitive dust from material handling. On-site emissions would primarily be driven by the natural gas-fueled Orcem stack emissions, which comprise more than 85% of on-site equipment NO_x emissions; other on-site equipment would be diesel-fueled. In comparison to the DFEIR, the 2018 ROA results in a 7% decrease in NO_x and a negligible decrease in PM₁₀ emissions for the Maximum Combined Scenario. This decrease is due to the following revisions:

- Revisions Common to Orcem and VMT
 - The 2018 ROA revises diesel emission factors to reflect the 2025 operational year and equipment model year 2020. In comparison, the DFEIR was based on the 2020 operational year and 2015 model year equipment.
 - The 2018 ROA assumes that diesel-fueled equipment (i.e., hoppers, conveyors, front end loaders, excavators, and forklifts) meet CARB/EPA Tier 4 Final engine standards. In comparison, the DFEIR assumed Tier 4 engines for Orcem front end loaders and excavators, but not for Orcem hoppers and conveyors or for VMT front end loaders or forklifts.
 - The 2018 ROA assumes that diesel-fueled equipment (i.e., hoppers, conveyors, front end loaders, excavators, and forklifts) use fuel with a 20% biodiesel content (B20). In comparison, the DFEIR assumed that Orcem and VMT front end loaders, and Orcem excavator used B20 fuel, VMT forklifts used conventional diesel fuel and, Orcem hoppers and conveyors used conventional diesel fuel.

NO_x Offsets

The 2018 ROA results in NO_x emissions greater than the BAAQMD significance threshold. The BAAQMD, as part of their permitting process, requires that emissions from the combination of ocean going vessels, rail, and stationary sources be offset if those emissions are greater than 10 tons per year. Each facility with emissions greater than 10 tons per year, but less than 35 tons per year can obtain offsets from the BAAQMD's Small Facility Offset Banking Account. Shipping,

rail and stationary source emissions are eligible for offsets from BAAQMD's Small Facility Offset Banking Account.⁷ Emissions greater than 35 tons per year must be offset by purchasing offsets.

2018 ROA NO_x emissions eligible for offsets, as shown in Table 5 of Appendix M, are 15.5 tons per year for Orcem and 18.9 tons per year for VMT, for a total of 34.5 tons per year. Following the application of BAAQMD offsets, total NO_x emissions are 7 tons per year, and as such are below the BAAQMD significance threshold for NO_x.

Cancer Risk

In comparison to the DFEIR, the 2018 ROA results in a negligible change in operational cancer risk. Cancer risk remains above the BAAQMD threshold of 10 in a million under the ROA, triggering the need for mitigation. In comparison to the DFEIR, mitigation is triggered sooner under the 2018 ROA for the following reason:

• The 2018 ROA implements a self-imposed threshold of 9 in a million cancer risk, which is below the BAAQMD threshold of 10 in a million. This revised threshold is implemented in part to allow for potential minor overlap between construction and operation, and in part to allow sufficient time for implementation of mitigation. Table 6.1 presents a comparison of DFEIR and 2018 ROA cancer risk and shows that mitigation (discussed under the proposed project in Section 3.0) is triggered above 14 vessels (i.e., sooner) under the 2018 ROA in comparison to 16 vessels under the DFEIR.

Table 6.1			
Cancer	Risk in	a	Million

	DFEIR	ROA
Construction Cancer Risk	5.57	1.14
Operation Cancer Risk at 48 Vessels	18.33	18.25
Operation Cancer Risk at 16 vessels	9.94	
Operation Cancer Risk at 14 vessels		8.96
Mitigation Required	Mitigation is required above 16 vessels	mitigation is required above 14 vessels

⁷ The BAAQMD funds the Small Facility Banking Account by deposit of unclaimed emission reductions from source or facility closures, and by a small facility growth allowance established in the District's Clean Air Plan. In general, BAAQMD's Emissions Banking Program allows for the creation of air pollutant emission reduction credits (ERCs). ERCs are created when companies introduce new emissions controls, such as upgrading or replacing old equipment, shutting down equipment, upgrading processes and materials, adopting stricter operating guidelines and adding control equipment to existing sources. These new controls must go beyond the requirements of current regulations and must be real, permanent, quantifiable, and enforceable. ERCs can then be used to offset emissions increases from new, permitted projects and traded or sold to other companies for their use. Over time, the offset program drives down the overall burden of pollution in the region because ERCs can only be generated by closures or introduction of control technologies that exceed regulatory requirements.

6.4.6 ROA Detailed Air Quality Analysis

The following information repeats some of the material presented in the above summary, but is presented to correspond in format and elements analyzed to the impact discussion found in Section 3.2 of this document

Regulatory Setting

Federal

Federal regulations applicable to the ROA are the same as those applicable to the proposed project in DFEIR Section 3.2.1.

State

State regulations applicable to the ROA are the same as those applicable to the proposed project in DFEIR Section 3.2.1.

Local

Bay Area Air Quality Management District

BAAQMD regulations applicable to the ROA are the same as those applicable to the proposed project in DFEIR Section 3.2.1 with the exception of the 2010 Clean Air Plan (CAP), which was revised in 2017. The 2017 CAP addresses nonattainment of the national 1-hour ozone standard in the San Francisco Bay Area Air Basin (SFBAAB) and defines a vision to achieve California's GHG reduction targets for 2030 and 2050. The purpose of the 2017 CAP is to:

- Update the Bay Area 2010 CAP in accordance with the requirements of the California Clean Air Act to implement "all feasible measures" to reduce ozone;
- Reduce transport of ozone and its precursors to neighboring air basins;
- Reduce emissions of fine particulate matter and toxic air contaminants;
- Present a long-range vision of how the Bay Area could function in a year 2050 post-carbon economy; and
- Review progress in improving air quality in recent years.

The 2017 CAP contains 85 control measures aimed at reducing air pollution in the SFBAAB including stationary, transportation, building and energy, agriculture, waste, water, and super-GHG control measures.

6.4.7 Existing Conditions

Background Concentrations

BAAQMD operates a regional 32-station monitoring network that measures the ambient concentrations of criteria pollutants.

Representative background concentrations for ozone, NO₂, and PM_{2.5} are based on the ambient monitoring station located on Tuolumne Street, Vallejo, California (Station No. 06-095-0004) and covers the three most recent complete years (2015–2017). The Tuolumne Street station does not collect data for CO, SO₂ or PM₁₀. The station is designated a neighborhood scale station (with a range of 500 meters to 4 kilometers) and is suitable for assigning a background concentration for determining project impacts. The monitoring station is located 2.5 kilometers northeast of the proposed project. The monitoring station is also located approximately downwind of the project, based on the wind data for both Vallejo and Conoco-Phillips Rodeo meteorological stations, and thus should be broadly representative of the location at which the maximum emissions from the facilities will occur. In relation to fugitive emissions from the project, the use of the Tuolumne Street station is likely to overestimate the background levels of PM_{2.5} due to the remote nature of the project site relative to the ambient monitoring station. The background data for the relevant pollutants is outlined in Table 6.2 for the last 3 years for which data is available.

			Highest Monitored Concentration a / Number of Days Above Standard		
Pollutant	Averaging Period	Standard	2015	2016	2017
Ozone (ppm)	1-hour State ^b	0.09	0.08/0	0.08 / 1	0.08 / 1
	8-hour National c	0.07	0.061 /0	0.063 / 1	0.061/2
	8-hour State ^b	0.07	0.061 /0	0.063/0	0.061/2
CO (ppm)	1-hour National	35	_	_	_
	1-hour State	20	_	_	_
	8-hour National and State	9	_	_	_
NO ₂ (ppm)	1-hour National ^d	0.100	0.038/0	0.036 / 0	0.042/0
	1-hour State ^d	0.18	0.050/0	0.050/0	0.050/0
	Annual National	0.053	0.008	0.007	0.008
	Annual State	0.030	0.009	0.008	0.008
SO ₂ (ppm)	1-hour National	0.075	_	_	_
	1-hour State	0.25	_	—	_
	24-hour State	0.04	_	_	_

Table 6.2Ambient Air Quality Data

Table 6.2Ambient Air Quality Data

			Highest Monitored Concentration ^a / Number of Days Above Standard		
Pollutant	Averaging Period	Standard	2015	2016	2017
PM ₁₀ (µg/m ³)	24-hour National	150	_/_	_/_	_/_
	24-hour State	50	_/_	_/_	_/_
	Annual State	20	—	—	—
PM _{2.5} (µg/m³)	24-hour National ^e	35	30.7 /3	19.0/0	41.5 / 9
	Annual National f	12	9.8	9.0	9.6
	Annual State g	12	11	10	12

Notes:

^a Exceedances of the standards are shown in bold. All reported values represent the highest recorded concentration unless otherwise noted.

^b The concentrations reported for the state 1-hour and 8-hour O3 standards represent the California designation values.

c The concentrations reported for the national 8-hour O3 represent the national standard design value.

^d The concentrations reported for the national 1-hour NO2 standard represent the 98th percentile national standard design value. The concentrations reported for the national annual NO2 standard represent the annual national standard design value. The concentrations reported for the state 1-hour and annual NO2 standard represent the CA designation values.

e The concentrations reported for the national 24-hour PM2.5 standard represent the 98th percentile national standard. High values are likely due to area wildfires.

^f The concentrations reported for the national annual PM2.5 standard represent the annual standard design value.

⁹ The concentrations reported for the state annual PM2.5 standard represent the CA annual standard designation value. Source: iADAM Database (CARB, 2015-2017).

6.4.8 Thresholds of Significance

Thresholds of significance are the same for the ROA as for the proposed project, in Section DFEIR 3.2.3. The analysis questions (originating from CEQA Appendix G) are repeated in this Section to facilitate ROA analysis.

6.4.9 Impact Discussion

This section presents a summary of the ROA activities and discusses potential impacts to air quality. The goal of the ROA is to provide an alternative to the proposed project that would accomplish the applicant's objectives but would reduce the potential environmental impacts associated with the project.

ROA Construction

The ROA would differ from the proposed project as follows:

• The ROA proposes to use construction equipment meeting EPA Tier 4 Final off-road engine emission standards. These engine emission standards are significantly cleaner than lower tier engines standards analyzed in the proposed project. Tier 4 Final engines would

be used for land-based construction activities, excluding wharf pile drivers and diesel hammers because the availability of Tier 4 Final engines for these types of equipment is uncertain. Although this component would reduce criteria pollutant emissions as compared to the proposed project, the ROA analysis does not recalculate criteria pollutant construction emissions and so does not take credit for this component in evaluating construction emissions.

- The ROA analysis does calculate and take credit for the use of Tier 4 Final engines in quantifying cancer risk associated with ROA construction. The discussion under Impact Threshold Questions B and D (discussed below) address this in greater detail.
- The ROA also calculates and takes credit for emission reductions associated with the elimination of VMT Phase 2 construction, described in the DFEIR, in quantifying cancer risk associated with ROA construction. The discussion under Impact Threshold Questions D (discussed below) addresses this in greater detail.

In addition to the above, the same Best Management Practices (BMPs) identified for the proposed project in DFEIR Section 3.2.4 would be used for the ROA. These BMPs, recommended by BAAQMD, would be required during all ROA construction activities.

ROA Operation

Tables 6.3, 6.4, 6.5, and 6.6 provide further detail regarding how ROA components and ROA analysis differ from the proposed project for on-road trucks, ships, rail, and on-site equipment, respectively. Table 6.7 identifies revisions to the health risk assessment. These Tables also identify where these changed elements originated: Either as a new "component of the ROA," a change due to updated analysis methodology, or a change in the project description of the original project that transferred to the ROA.

Orcem Revisions	
The ROA proposes an Orcem Truck Loading & Weight Confirmation System, which would increase the capacity of Orcem trucks from approximately 20 to 23.58 MT. This ROA component would decrease truck trips compared to the proposed project because fewer trucks with larger capacity would be needed to transport the same amount of product.	Component of the ROA
The ROA proposes The Enhanced Orcem Truck Scheduling Efficiency System, which would reduce the average daily number of trucks from approximately 189 to 122 by increasing the average monthly trucking days from approximately 17.5 to 26. Per the transportation analysis, this increase in monthly trucking days would correspond to an increase from 5 to 6 days per week. ⁸	Component of the ROA

Table 6.3ROA On-Road Truck Revisions

⁸ DFEIR, Appendix L, Fehr and Peers. 2016. *Transportation Technical Data*.

Table 6.3ROA On-Road Truck Revisions

The ROA analysis subtracts material that would be transported via rail from total truck trips. This had not been accounted for in the proposed project analysis. This revision would	Change in analysis methodology
reduce truck trips compared to the proposed project.	
VMT Revisions	
VMT Phase 2 was removed from the project based on comments received on the Draft EIR. Without construction of VMT Phase 2, which would have expanded the laydown area for the bulk and break-bulk materials delivered to the site, space constraints preclude VMT from loading material by truck and rail simultaneously, for material throughput above 12 VMT vessels. Although, rail and truck loading could occur sequentially, rail and trucks could not be loaded simultaneously in the same month. The reason for this is that during the truck transloading process, the terminal would be fully occupied for one full month and another vessel would be unable to unload to rail until the terminal is cleared due to lack of transloading space. Transloading to trucks one month and rail the next month could occur during years of lower throughput, but this practice would not be possible at the maximum scenario of 29 VMT vessels, which is when maximum ROA impacts are calculated to occur. Therefore, the ROA assumes that all VMT material, for operation above 12 ships per year would be transported by rail.	Change in the project description
Revisions Common to Orcem and VMT	
The ROA analysis increases the one-way truck transit distance to 41.9 miles, from the 25 miles one-way distance analyzed in the proposed project.	Change in analysis methodology
The ROA revises the operational year from 2020, which was analyzed for the proposed project, to 2025. This revision would decrease truck engine emission factors compared to the proposed project because trucks in future years are expected to be cleaner per CARB's existing regulatory requirements.	Change in analysis methodology
The ROA analysis uses speed-specific emission factors and aggregated emission factors for truck transit. Speed-specific emission factors reflect a speed of 10 miles per hour (mph) for on- site transit and 20 and 40 mph for transit along surface streets. For freeway transit, the ROA analysis uses emission factors that are based on aggregated speeds in the County. ⁹ This is a more conservative assumption than the proposed project, which assumed emission factors based on 40 mph for all transit. Since emission factors increase at higher speed, this revision	Change in analysis methodology

Table 6.4ROA Ship Revisions

Revisions Common to Orcem and VMT	
The ROA analysis revises emission factors to reflect the "lower emission factors" and "low load	Change in
adjustment factors" in the anticipated 2018 CARB shipping inventory database. In 2018, CARB plans to	analysis
release a shipping inventory database that incorporates newer ship engine information than what was	methodology
available at the time of the proposed project analysis in the DEIR, including lower emission factors and	
lower adjustment factors. A CARB documentation e-mail is included in Appendix M. This revision would	
reduce emission factors compared to the proposed project.	

⁹ Aggregated speed refers to a combined speed distribution that is developed by the CARB as part of the EMFAC database.

Table 6.5ROA Rail Revisions

Orcem Revisions	
The ROA analysis reduces the number of Orcem trains serving the project by eliminating empty train trips. The ROA assumes that Orcem trains would arrive full bringing raw material to Orcem, offload the raw material, be loaded with finished product, and would then leave full, transporting finished product from Orcem. This had not been accounted for in the proposed project analysis because it was only recently addressed by the applicants. In comparison, the proposed project assumed that (1) trains bringing raw material would arrive full, would offload raw material, and would then leave empty; and (2) trains transporting finished product would arrive empty, be loaded with finished product, and would leave full.	Component of the ROA
VMT Revisions	
As discussed in Table 6.3, due to removal of VMT Phase 2, space constraints would preclude VMT from loading material by truck and rail simultaneously, for material throughput above 12 VMT ships. Therefore, the ROA analysis assumes that all material, for operation above 12 ships per year would be transported by rail. This revision would increase VMT rail trips (up to a maximum of 4 rail trips a year).	Component of the ROA
Revisions Common to Orcem and VMT	
The ROA assumes shorter trains consisting of 50 rail cars per train for both Orcem and VMT, compared to the proposed project, which assumed 77 rail cars per train. Shorter trains reduce (but do not eliminate) impacts from intersection delay and the delay of emergency vehicles. This revision would increase the number of trains, compared to the proposed project, because a greater number of shorter trains would be needed to transport the same amount of material.	Component of the ROA
Based on recent information from CalNorthern, the rail line operator, the ROA assumes that 50 rail-car trains require the use of two locomotives per train for line-haul trains. This is a revision from the proposed project, which assumed one locomotive for an empty train and three for a full train.	Component of the ROA
The ROA incorporates the use of switcher locomotives that meet EPA Tier 4 engine standards. These locomotives have only recently become available from The National Railway Equipment Company and were not available at the time of the proposed project analysis, which assumed EPA Tier 2 engines. This revision would decrease emission factors associated with switching operations.	Component of the ROA
The ROA conservatively assumes a 2025 CARB fleet mix for line haul locomotives, which consists of Tier 2 through Tier 4 locomotives. In comparison, the proposed project assumed the use of all Tier 4, line haul locomotives. This ROA revision would increase emission factors associated with line haul locomotives.	Component of the ROA

Table 6.6ROA On-Site Equipment Revisions

Revisions Common to Orcem and VMT	
The ROA revises diesel emission factors to reflect the 2025 operational year and equipment model year 2020. In comparison, the proposed project analysis was based on the 2020 operational year and 2015 model year equipment. This revision would decrease emission factors because equipment in future years are expected to be cleaner per CARB's existing regulatory requirements.	Change in analysis methodology
The ROA assumes that diesel-fueled equipment (i.e., hoppers, conveyors, front end loaders, excavators, and forklifts) would meet EPA Tier 4 Final engine standards. In comparison, the proposed project assumed Tier 4 engines for Orcem front end loaders and excavators, but not for Orcem hoppers and conveyors or for VMT front end loaders or forklifts. This revision would reduce emission factors because more equipment would comply with cleaner Tier 4 Final engine standards.	Component of the ROA
The ROA assumes that all diesel-fueled equipment (i.e., hoppers, conveyors, front end loaders, excavators, and forklifts) would use fuel with a 20% biodiesel content (B20). In comparison, the proposed project assumed that Orcem and VMT front end loaders, and Orcem excavator used B20 fuel, VMT	Component of the ROA

Vallejo Marine Terminal and Orcem Project Final EIR

Table 6.6ROA On-Site Equipment Revisions

Revisions Common to Orcem and VMT	
forklifts used conventional diesel fuel and, Orcem hoppers and conveyors used conventional diesel fuel.	
i his revision would reduce particulate and diesel particulate emissions.	

Table 6.7ROA Health Risk Assessment Revisions

Revisions Common to Orcem and VMT	
The ROA analysis implements the 2015 Office of Environmental Health Hazard Assessment (OEHHA) Guidance to quantify health impacts. This methodology was not available at the time of the proposed project air quality analysis and is more stringent.	Change in analysis methodology
The ROA proposes a self-imposed threshold of 9 in a million cancer risk, which is below the BAAQMD threshold of 10 in a million. This revised threshold is proposed in part to allow for potential minor overlap between construction and operation, and in part to allow sufficient time for implementation of mitigation.	Component of the ROA

In addition to the above, the same Project Design Features (PDFs) identified for the proposed project in Section 3.2 would be used in the ROA to reduce on-site emissions during operations.

As with the proposed project, the ROA anticipates that material throughput for both VMT and Orcem would ramp up over time. Operational throughput would depend on market demand and is difficult to predict. However, for the purposes of the ROA analysis, it was assumed that maximum material throughput would not be reached until 2025 (as detailed in the operating scenarios [modes] described above).

It is important to note that given the nature of VMT and Orcem facilities, certain operations could only be undertaken by one of the operators at any given time. For example, ship unloading could only occur from one ship at a time, either a VMT ship or an Orcem ship. Thus, when Orcem would be unloading product from a ship, VMT would not be able to unload from a ship. Likewise, when VMT would be loading export material into railcars, Orcem would not load railcars at the same time. These exclusions would result in a Combined Maximum Scenario that is lower than individual VMT or Orcem throughput, if facilities were to operate independently.

Table 6.8 presents material throughput and annual activity associated with the ROA Maximum Combined Scenario for criteria pollutants. Individual facility maximums are presented in Appendix M.

Table 6.8Material Throughput and Activity:Maximum Combined Scenario for Criteria Pollutants

	VMT	Orcem	
Ma	aterial Throughput (metric tons per year)		
Material Imported by Ship	1,160,000	760,000	
Material Imported by Truck	0	22,306	
Material Imported by Rail	0	120,000	
Material Exported by Truck	0	557,196	
Material Exported by Rail	1,160,000	145,732	
Annual Activity			
Ships	29	19	
Trucks	0	49,152	
Delivery Trucks	2,496	0	
Trains	512	65	
Rail Cars	12,790	1,607	

Notes:

Values are rounded.

^{2.} In the Maximum Combined Scenario for criteria pollutants is when Orcem would operate in Mode 3 / Milestone 5, producing GGBFS and cement products.

³ Trucks reflect 1-way trips. In the Maximum Combined Scenario for criteria pollutants, all VMT product would be moved via rail. Rail loading is more efficient than truck loading, allowing the greatest material throughput and greatest number of ships per year. Since ships have a greater emissions profile than trucks, the analysis is maximized with the maximum number of ships. An analysis that includes VMT trucks would result in greater truck emissions, but lower ship emissions, thereby resulting in lower overall emissions. Table 6.1 provides additional explanation as does Appendix M.

4. Delivery trucks would be used to deliver miscellaneous supplies to the VMT facility. These trucks would be smaller than trucks used to export and import product.

5. Trains represent line haul 1-way trips. Each line haul train would be comprised of 50 rail cars.

6. During production of GGBFS the moisture content is greatly reduced, thus less material is exported than imported.

6.4.10 CEQA Appendix G Threshold Questions

The following questions were presented in Section 3.2.3 of the EIR and are repeated here to facilitate ROA analysis.

A. Would the ROA conflict with or obstruct implementation of the applicable air quality plan?

The ROA would have similar activities as the proposed project. As described in Tables 6.3 through 6.7 several ROA components were added to the ROA and calculation methodologies were revised to reflect current industry practice.

The most recent Bay Area air quality plan is the 2017 CAP. The CAP provides a regional strategy to protect public health and protect the climate. The 2017 CAP updated the 2010 Clean Air Plan, which was used for significance determination of the proposed project because that was the current plan at the time of proposed project analysis. The 2017 CAP sets goals to fulfill state ozone

planning requirements via reduction of ozone precursors, meet the state's 2030 and 2050 GHG targets, and builds upon the BAAQMD's efforts to reduce emissions of fine particulate matter and toxic air contaminants.

Projects are considered consistent with, and would not conflict with or obstruct implementation of, the local air quality management plan if the growth in socioeconomic factors (e.g., population, employment) is consistent with the underlying regional plans used to develop local air quality management plans. Demographic growth forecasts for various socioeconomic categories, developed by the Metropolitan Transportation Commission, the Association of Bay Area Governments, and local and regional agencies were used to estimate future emissions in the 2017 CAP.

The 2017 BAAQMD CEQA Guidelines recommend consideration of the following three questions to determine consistency with the relevant air quality plan:

- 1. Does the project support the primary goals of the air quality plan?
- 2. Does the project include applicable control measures from the air quality plan?
- 3. Does the project disrupt or hinder implementation of any CAP control measures?

Regarding question number 1, the primary goals of the 2017 CAP are to attain air quality standards under the NAAQS and CAAQS, protect public health, and reduce regionally generated GHG emissions. The 2017 CAP includes strategies designed to bring the SFBAAB into attainment of the CAAQS and NAAQS. Strategies include more stringent standards for new engines and cleanup of existing fleets, new measures for port trucks, statewide truck fleets, ships traveling and in port, locomotives, and harbor craft that are enforced at the state and federal level on engine manufacturers and petroleum refiners and retailers, ROA operation would comply with such strategies. The BAAQMD also adopts CAP control measures into its rules and regulations, which are then used to regulate sources of air pollution in the SFBAAB. Therefore, compliance with these requirements would ensure that the ROA, like the proposed project, would not obstruct implementation of the CAP.

As explained in the discussion under Impact Threshold Questions B below, there would be no significant impacts for criteria pollutant emissions during ROA operations. Therefore, impacts associated with the ROA would be **less than significant**.

Regarding question number 2, the CAP includes control measures related to four primary categories: Stationary Source Measures, Transportation Control Measures, Buildings and Energy Control, and Air District's Tools. Many of the control measures in the CAP would not apply to the ROA. However, the ROA would implement several measures that would also promote CAP control measures. Specifically, ROA construction would promote CAP Control Measure TR22 by requiring the use of Tier 4 engines on construction equipment. ROA operations would also

promote CAP Control Measure TR18, through the use of diversified material transport and distribution, through a combined use of truck, rail, and vessel transportation modes.

In addition, implementation of mitigation measures identified in Section 3.2, would include applicable control measures from the CAP. Specifically, MM-3.2-1 would require the use of 2010 trucks upon facility start-up; this mitigation measure is consistent with CAP Control Measure TR19. MM-3.2-2 would require an increase in or replacement of diesel-powered terminal equipment with either biodiesel, natural gas, or electric-powered equipment; MM-3.2-2 would be consistent with CAP Control Measure SS20. MM-3.2-2 would also require the use of a CARB-approved capture and control system to treat emissions from auxiliary engines on ocean-going vessels once the annual ship calls reaches 34; this would support BAAQMD measure SS20.

The ROA would also implement BAAQMD BMPs related to fugitive dust control and project design features PDF-AQ-1 through PDF-AQ-4 as described previously.

However, without mitigation presented in Section 3.2, this impact (regarding CAP control measures) would be **significant**.

Regarding question number 3, the ROA would have similar activities as the proposed project and would therefore result in similar impacts. The ROA would not disrupt or hinder implementation of control measures delineated in the CAP. Impacts, with regard to question number 3, would be **less than significant**.

B. Would the ROA violate any air quality standard or contribute substantially to an existing or projected air quality violation?

Construction Impacts

Construction emissions were calculated using the year the proposed project (VMT and Orcem) was originally proposed and were not re-calculated in the ROA analysis. As described above, the ROA proposes to use Tier 4 Final engines, which are significantly cleaner than lower tier engines. Tier 4 Final engines would be used for land-based construction activities, excluding wharf pile drivers and diesel hammers because the availability of Tier 4 engines for these types of equipment is uncertain. Although this component would reduce criteria pollutant emissions as compared to the proposed project, the ROA analysis did not quantify these reductions because the criteria pollutant emissions associated with ROA construction were not recalculated. Since the proposed project construction emissions were below the CEQA thresholds for all criteria pollutants, the ROA construction emissions would also be below the CEQA thresholds and as such would be **less than significant**.

Operational Impacts

ROA operations would use trucks, ships, rail, and on-site equipment similarly to the proposed project. ROA revisions to operation of these sources, as compared to the proposed project, are

identified in Tables 6.2 through 6.5. ROA operational emissions were calculated for individual emission sources using EPA, CARB, and AP-42 methodologies, updated since the time of the proposed project analysis, and ROA activity as described in Table 6.9.¹⁰

Table 6.10 presents criteria pollutant emissions for the ROA Maximum Combined Scenario. The table shows that exhaust emissions would be driven by shipping, trucks, rail, and stack emissions. The table also shows that fugitive emissions would be driven by Orcem truck trips and both VMT and Orcem material handling emissions.

VMT and Orcem are subject to the New Source Review program and BAAQMD permitting. Given that neither VMT nor Orcem emissions would exceed the PSD threshold of 250 tpy per pollutant, neither VMT nor Orcem would be subject to PSD review (Appendix D-1).

The BAAQMD, as part of their permitting process, requires that emissions from the combination of ocean-going vessels, rail, and stationary sources be offset if those emissions are greater than 10 tons per year. Each facility with emissions greater than 10 tons per year, but less than 35 tons per year can obtain offsets from the BAAQMD's Small Facility Offset Banking Account.¹¹ ROA shipping, rail and stationary source emissions would reach 41.39 tons per year and would therefore be eligible for offsets from BAAQMD's Small Facility Offset Banking Account. Following the application of BAAQMD offsets, total NO_x emissions would be just below 7 tons per year, which would be below the BAAQMD significance threshold for NO_x.

Table 6.9 shows that the ROA emissions would be similar to and, in the case of NO_x , lower than the proposed project (DFEIR Table 3.2-13) due to the revisions described in Tables 6.3 through 6.6.

Following application of BAAQMD-required NO_x offsets, the Maximum Combined Scenario would not exceed the BAAQMD threshold for any criteria pollutant. Impacts would be considered **less than significant**.

¹⁰ It is not anticipated that portable diesel generators will be used during routine activities. Portable diesel generators may be used during the initial phase of construction until Pacific Gas & Electric (PG&E) completes new electric service installation. Portable diesel generators may be used during unanticipated events or repairs. If such events arise, diesel generators shall be registered under the California Air Resources Board's (CARB's) Portable Equipment Registration Program (PERP).

¹¹ The BAAQMD funds the Small Facility Banking Account by deposit of unclaimed emission reductions from source or facility closures, and by a small facility growth allowance established in the District's Clean Air Plan. In general, BAAQMD's Emissions Banking Program allows for the creation of air pollutant emission reduction credits (ERCs). ERCs are created when companies introduce new emissions controls, such as upgrading or replacing old equipment, shutting down equipment, upgrading processes and materials, adopting stricter operating guidelines and adding control equipment to existing sources. These new controls must go beyond the requirements of current regulations and must be real, permanent, quantifiable, and enforceable. ERCs can then be used to offset emissions increases from new, permitted projects and traded or sold to other companies for their use. Over time, the offset program drives down the overall burden of pollution in the region because ERCs can only be generated by closures or introduction of control technologies that exceed regulatory requirements.

			Emissions (tons/year)						
						PM10	PM10	PM _{2.5}	PM _{2.5}
Facility	Source	ROG	СО	NOx	SOx	exhaust	fugitive	exhaust	fugitive
VMT	Shipping	1.3	2.4	13.8	1.2	0.39	—	0.38	_
	Rail	0.18	2.3	5.1	0.009	0.08	—	0.07	—
	Trucks	0.00	0.03	0.10	0.00	0.00	0.02	0.00	0.00
	Workers	0.02	0.22	0.02	0.00	0.00	0.23	0.00	0.06
	Storage Piles	—	—	—	—	—	0.00	—	0.00
	Material Handling	_	_			—	1.0	—	0.13
	Front End Loader	0.13	1.0	0.12	0.00	0.09	0.058	0.09	0.001
	Forklift	0.00	0.09	0.02	0.00	0.00	0.04	0.00	0.04
	VMT Total	1.61	6.03	19.15	1.2	0.48	1.32	0.46	0.20
	BAAQMD Offsets	_	—	18.92		—	—	_	_
	VMT Total After Offsets	1.61	6.03	0.22	1.2	0.48	1.32	0.46	0.20
Orcem	Shipping	0.84	1.5	9.1	0.78	0.26	—	0.25	_
	Rail	0.03	0.41	0.88	0.00	0.01	—	0.01	_
	Trucks	0.08	0.94	6.0	0.03	0.06	1.1	0.06	0.3
	Workers	0.03	0.35	0.025	0.00	0.00	0.38	0.00	0.09
	Storage Piles	_	—	_	_	—	0.00	—	0.00
	Material Handling	_	—	_	_	—	0.68	—	0.10
	Front-End Loader	0.22	1.2	0.13	0.00	0.01	0.10	0.01	0.01
	Excavator	0.03	0.52	0.09	0.00	0.01	0.02	0.01	0.00
	Stack	0.69	11	5.6	0.18	0.25	—	0.25	_
	Hopper/conveyor	0.06	2.4	0.50	0.00	0.01	_	0.01	_
	Bag Filters	_	_	_	_	0.18	_	0.16	_
	Orcem Total	1.98	18.68	22.24	1.00	0.60	2.43	0.59	0.66
	BAAQMD Offsets	_	—	15.53		—	—	_	_
	Orcem Total After Offsets	1.98	18.68	6.72	1.00	0.60	2.43	0.59	0.66
Total VMT + Orcem After Offsets		3.59	24.71	6.94	2.20	1.08	3.75	1.05	0.86
BAAQMD CEQA Thresholds		_	_	10	_	15	—	10	
CEQA Determination		_	_	No	_	No	—	No	—

Table 6.9 **Emissions Summary: Maximum Combined Scenario**

Notes:

1

Emissions are rounded and may not add exactly. Detailed emissions are presented in Appendix M. Annual emissions and BAAQMD annual thresholds are equivalent to average daily emissions and daily thresholds, assuming 365 days/year 2 of operation.

Shipping emissions include ship transit, ship hoteling, and tugboat emissions. Rail emissions include line haul and switcher emissions. 3

4

C. Would the ROA result in a cumulatively considerable net increase of any criteria pollutant for which the project region is non-attainment under an applicable federal or state ambient air quality standard (including releasing emissions which exceed quantitative thresholds for ozone precursors)?

Past, present and future development projects may contribute to the region's adverse air quality impacts on a cumulative basis. Per BAAQMD's CEQA Guidelines, by its nature air pollution is largely a cumulative impact: no single project is sufficient in size to, by itself, result in nonattainment of ambient air quality standards. In developing thresholds of significance for air pollutants, BAAQMD considered the emission levels for which a project's individual emissions would be cumulatively considerable. BAAQMD holds that if a project exceeds the identified significance thresholds, its emissions would be considered cumulatively considerable, resulting in significant adverse air quality impacts to the region's existing air quality conditions.

Construction Impacts

ROA construction activities and potential projects within the ROA's vicinity would be similar to those discussed under the proposed project evaluation in Section 3.2. Construction of cumulative projects would be short term and temporary in nature. As fugitive dust impacts are generally localized to individual project sites, and on-site emissions would be sufficiently mitigated through demolition and dust control measures, coupled with implementation of BAAQMD BMPs as listed in Section 3.2, cumulative impacts related to fugitive dust would be considered **less than significant**.

ROA construction emissions were not calculated but are expected to be lower than proposed project construction emissions because ROA construction would include the use of EPA Tier 4 Final standard engines for construction equipment, as described in DFEIR Section 3.2-4. As is the case with the proposed project, the ROA would implement BAAQMD's BMPs to control fugitive dust. Moreover, once construction activities are completed, construction-related emissions would cease.

Thresholds established by the BAAQMD as shown in DFEIR Table 3.2-6 are used to evaluate air quality impacts, including cumulative impacts. Thresholds established by the BAAQMD reflect the attainment status of the project area and provide for the consideration of project impacts in light of the region's nonattainment status for certain criteria pollutants. As such, these thresholds also provide a basis to evaluate the ROA's contribution to air pollutant emissions and concentrations under the cumulative criterion.

DFEIR Table 3.2-9 shows that construction of the proposed project would not exceed BAAQMD construction thresholds for any criteria pollutants. Since ROA construction emissions are expected to be less than the proposed project, as discussed above, ROA construction activities would also not result in cumulatively considerable impacts. Cumulative project impacts would be considered **less than significant** during the temporary construction period.

Operational Impacts

Table 6.9 shows that the ROA would generate operational emissions that would not exceed the significance threshold for any of the criteria pollutants. The ROA would therefore, not result in cumulatively considerable impacts. Cumulative project impacts would be considered **less than significant**.

D. Would the project expose sensitive receptors to substantial pollutant concentrations?

The BAAQMD has adopted project and cumulative thresholds for three risk-related air quality indicators to sensitive receptors: cancer risks, non-cancer health effects, and increases in ambient air concentrations of PM_{2.5}. These impacts are addressed on a localized rather than regional basis, in relation to sensitive receptors within 2.5 miles of the project identified in DFEIR Table 3.2-14. Cancer risk is the probability or chance of contracting cancer over a human life span. Carcinogens are assumed to have no threshold below which there would be no human health impact. In other words, any exposure to a carcinogen is assumed to have some probability of causing cancer. Cancer risk is expressed as excess cancer cases per one million exposed individuals, typically over a lifetime of exposure. Non-carcinogenic substances differ in that there is assumed to be a safe level of exposure below which no negative health impact is believed to occur. These levels are determined on a pollutant-by-pollutant basis. Acute and chronic exposure to non-carcinogens is expressed as a hazard index (HI), which is the ratio of exposure levels to an acceptable reference exposure levels.

As with the proposed project, many of the ROA emission sources would be diesel-powered. Diesel particulate matter, classified as a toxic air contaminant by CARB, is a key pollutant evaluated in assessing health risk. As in the proposed project analysis, PM_{2.5} emissions from diesel engine combustion were used as a surrogate for DPM in the ROA, and ship boiler emissions were speciated into their individual TAC components using speciation data in Appendix D-1. Fugitive TAC emissions, associated with the storage, handling, and processing of GBFS and gypsum, were also speciated into their individual TAC components using speciation data in Appendix D-1.

The ROA analysis used the same dispersion model results that were calculated for the proposed project analysis (See Appendix D-1 for details regarding model input parameters). In other words, dispersion modeling was not re-run for the ROA. Cancer risk, however, was recalculated to incorporate the 2015 Office of Environmental Health Hazard Assessment Guidelines for both construction and operation cancer risk, as well as the revisions noted in Tables 6.3 through 6.7. Construction and operational impact discussions below provide greater detail regarding cancer risk methodology and specifics used in the ROA.

Construction Impacts

Cancer risk associated with ROA construction was calculated by scaling the proposed project construction cancer risk by the DPM emission reductions specific to construction of the ROA.

ROA DPM reductions would be due to 1) the removal of VMT Phase 2 construction and 2) the use of Tier 4 Final construction equipment.¹²

In addition, the ROA analysis incorporates OEHHA's 2015 Guidance, which has been adopted by the BAAQMD since the proposed project analysis was conducted. OEHHA's 2015 Guidelines revised breathing rates, used to calculate inhalation cancer risk for different years of exposure, to be more conservative. This guidance was incorporated into the ROA construction cancer risk and is described in detail in Appendix M.

The ROA construction cancer risk was then scaled from the proposed project to reflect the reduction in DPM emissions and the change in 2015 OEHHA Guidance. Table 6.10 shows that ROA DPM emissions would be reduced compared to the proposed project and shows that the corresponding cancer risk would also be reduced. This reduction in construction cancer risk indicates that although the use of OEHHA's 2015 Guidance tends to conservatively increase cancer risk, this increase would be offset by elimination of Phase 2 VMT construction and, in particular, by the use of Tier 4 Final construction equipment.

Non-cancer health effects due to chronic and acute exposure were not recalculated for the ROA but would be less than the proposed project due to the reductions described above. Therefore, construction impacts would be **less than significant**.

Construction Phase	DPM Emissions without Tier 4 (tons)	DPM Emissions with Tier 4 (tons)	DPM Emission Reduction (tons)	
Orcem 0.16333		0.01579	0.14754	
VMT	0.05017	0.01579	0.03536	
	0.086			
	0.2689			
	0.301			
	0.032			
	89.34%			
	5.7 in a million			
	1.14 in a million			

 Table 6.9

 Emissions Summary: Maximum Combined Scenario

Notes:

¹ Orcem-VMT ROA Summary Report FINAL

² VMT Phase 2 and total DFEIR emissions taken from DFEIR Appendix D-1 Table 4

¹² Although the ROA did not recalculate criteria pollutant emissions and so did not take credit for the use of Tier 4 Final construction equipment in determining impacts associated with criteria pollutants, the ROA analysis of construction cancer risk did take into account DPM reductions associated with the use of Tier 4 Final construction equipment.

Operational Impacts

Local Carbon Monoxide Concentrations

The thresholds of Significance for local CO emissions are the 1-hour and 8-hour CAAQS of 20.0 parts per million (ppm) and 9.0 ppm, respectively. By definition, these represent levels that are protective of public health. The BAAQMD has developed screening conditions below which projects are considered to meet the thresholds without having to conduct dispersion modeling for CO. BAAQMD's screening conditions are discussed in DFEIR Section 3.2.4. ROA sources of CO emissions are the same as the proposed project: stationary source, rail traffic, truck traffic, on-site mobile equipment, and ship traffic. As with the proposed project, the ROA Transportation and Traffic section shows that there are no intersections or grade crossings affected by the ROA with a maximum hourly traffic volume of 24,000 vehicles per hour, the BAAQMD screening threshold.

The CO impact from rail traffic is expected to be low because rail emissions are stringently controlled (Appendix M). In addition, although the ROA would increase rail traffic with the use of shorter trains and greater reliance on VMT rail for the Maximum Combined Scenario, ROA rail emissions would not increase compared to the proposed project for the reasons noted in Table 6.5. Specifically, the ROA would eliminate empty Orcem trains trips and incorporate the use of NREC Tier 4 switcher locomotives.

As with the proposed project, ROA impacts from hoteling vessels, and the Orcem stack (stationary source equipment) would have the greatest potential to result in off-site impacts of CO (Appendix D-1). ROA CO vessel emissions and Orcem stack emissions would be comparable to the proposed project as shown in ROA Table 6.9 and proposed project DFEIR Tables 3.2-11 and 3.2-12. CO dispersion modeling was conducted for the proposed project and showed CO off-site concentrations below BAAQMD thresholds in DFEIR Table 3.2-16. Therefore, since ROA CO emissions would be lower than the proposed project and since the source and receptor locations would not change, ROA CO off-site concentrations would also be below BAAQMD thresholds. For the reasons discussed above, ROA CO impacts would be **less than significant**.

Cancer Risks and Hazards

Operational cancer risk was calculated with the methodology revisions and components identified in Table 6.7. Specifically, the ROA analysis implements the 2015 OEHHA Guidelines and imposes a threshold of 9 in a million, which is more stringent that the standard BAAQMD threshold of 10 in a million. In addition, per Table 6.6, the ROA analysis, assumed the use of Tier 4 engines and the use of B20 (fuel with 20% biodiesel and 80% regular diesel) in all diesel-fueled on-site equipment.
The cancer risk was calculated by combining ROA DPM emissions with the dispersion modeling conducted for the proposed project. This is an appropriate methodology because the ROA source locations, source parameters, and receptor locations would be the same as for the proposed project. Noncancer chronic health impacts, acute health impacts, and the PM_{2.5} concentration were not recalculated because the methodology for these impacts has not changed since the proposed project analysis and the proposed project analysis showed that these impacts would be well below BAAQMD thresholds.

The ROA analyzed various operating scenarios and determined that maximum impacts for cancer risk would occur during the Maximum Combined Scenario when:

- Orcem would operate in Mode 2 / Milestone 5. This production mode reflects Orcem production of blended cement only and would use 19 annual vessel shipments of raw materials, where processed materials would be loaded into trucks for distribution.
- VMT would operate using 29 annual vessel shipments of raw materials, where imported goods would be loaded onto rail for distribution.

As described previously, the Maximum Combined Scenario would be different for cancer risk than for other air quality impacts. The Maximum Combined Scenario for cancer risk would occur when Orcem would operate in Mode 2, when only cement would be produced. During this production Mode, all processed material would be loaded onto trucks for distribution. Other air quality impacts would be maximum when Orcem would operate in Mode 3 during production of GGBFS and cement products, when processed material would be loaded onto both truck and rail. Since trucks would be the closest sources to sensitive receptors and since trucks would result in greater emissions than rail, the maximum cancer risk was determined to occur during Orcem Mode 2, when all processed materials would be loaded onto trucks.

Table 6.11 shows that the ROA cancer risk for the Maximum Combined Scenario would exceed a cancer risk threshold of 9.0 in a million. The 9.0 threshold was established by the applicant and is lower than the 10.0 threshold established by BAAQMD in order to allow for potential minor overlap between construction and operation, and in part to allow sufficient time for implementation of mitigations.

The ROA cancer risk would be approximately the same as the proposed project cancer risk of 18.33 in a million. It should be noted that the ROA cancer risk would not reach the level of significance of 9.0 in one million until the annual number of ship calls exceeds 14 ships per year. A detailed explanation of ROA cancer risk is included in the Orcem-VMT ROA Summary Report Final, Appendix M.

Table 6.11ROA Health Risks Impacts

			Estimated Value	Threshold
	Threshold	Units	(unmitigated)	Exceeded?
ROA Cancer Risk	9.0	In one million	18.25	Yes (unmitigated)

Source: Orcem-VMT ROA Summary Report Final, Appendix M

As shown in Table 6.11, ROA operations would exceed the threshold for cancer risk. Impacts would therefore be **significant**. Mitigation Measures MM-3.2-2 and MM-3.2-3 described in Section 3.2.5, would be implemented to reduce cancer risk to **less-than-significant** levels.

Cumulative Risks and Hazards

The BAAQMD considers a project to have a cumulatively considerable impact if the aggregate total of past, present and foreseeable future sources within a 1,000-foot radius from the fence line plus the contribution from the project exceeds the significance thresholds identified in DFEIR Table 3.2-6. The ROA did not recalculate aggregate cumulative impacts because permitted stationary sources of TACs near the project site, identified using BAAQMD's Stationary Source Risk and Hazard Analysis Tool during the proposed project analysis, have not changed. Table 3.2-18 of the DFEIR shows that aggregate projects would not exceed BAAQMD thresholds. The ROA would also be in compliance with the BAAQMD's adopted Thresholds for Single Source and Cumulative community risks, as well as hazard index risks. The ROA would therefore have a **less-than-significant** cumulative health risk impact.

Would the project create objectionable odors affecting a substantial number of people?

Construction Impacts

ROA construction would increase air pollutants due to the combustion of diesel fuel. ROA construction would be identical in activity and duration to the proposed project and as such would have the same odor impacts as the proposed project. Since odor impacts associated with the proposed project were found to be less than significant, impacts associated with odors during ROA construction would also be considered *less than significant*.

Operational Impacts

ROA operation would increase air pollutants due to the combustion of diesel fuel and processing of GBFS. ROA operation would be similar in activity to the proposed project and as such would have the same odor impacts as the proposed project. Since odor impacts associated with the proposed project were found to be less than significant, impacts associated with odors during ROA operation would also be considered *less than significant*.

Biological Resources

The ROA would involve similar construction activities as the proposed project and would therefore result in similar impacts to terrestrial and marine biological resources during construction (potential impacts to special-status species and their habitats would be reduced through mitigation).

The ROA would involve similar operational activities as the proposed project that are intended to reduce potential impacts related to aesthetics, air quality, noise, and transportation. These activities include more efficient and spread out operation of ship, rail car, truck, and heavy equipment within the area of the existing developed site and developed off-site areas (six days per week instead of five days per week, with fewer truck trips per day). The number of trucks used by the VMT and length of trains would be reduced (from 77 rail cars to 50 rail cars), and preference would be given to increased use of barges over trucks and rail, thereby increasing the likely number of days that vessels would be docked at the VMT Terminal. Finally, late night operations (within 300 feet of the nearest residential boundary) at the Orcem site would be eliminated. These minor variations in operational activities are anticipated to have similar impacts to terrestrial and marine biological resources as the proposed project. For these reasons, biological resources impacts from the ROA would be similar to biological resources impacts from the proposed project.

Cultural Resources

The ROA would involve demolition of the same buildings and the similar construction activities as the proposed project. Therefore, cultural resources impacts from the ROA would be similar to cultural resources impacts from the proposed project, including the significant and unavoidable impact to historic architectural resources due to the loss of integrity of a potential Sperry Flour Mill Historic District associated with demolition of the flour mill, grain silos, and dock.

Geology and Soils

Under the ROA, the project site would be developed in a similar manner as the proposed project, and potential impacts due to geology and soils would therefore be similar. Geology and soils impacts, including slope instability, would also be similar to the proposed project (reduced through mitigation).

Greenhouse Gas Emissions

The ROA would result in similar construction emissions as the proposed project since the ROA includes change to the same facilities on the site. Construction GHG impacts would therefore be similar to the proposed project (reduced through mitigation).

Once operational, both components of the project would be subjected to the requirements of a BAAQMD permit (to regulate stationary on-site equipment), as in the proposed project. As part

of the ROA, Orcem would implement the Revised Truck Loading & Weight Confirmation System to improve the efficiency of tanker trucks leaving the site, reducing the overall number of project truck trips, and therefore resulting in reductions in daily NO_x, CO₂, and PM_{2.5}/PM₁₀. As required by regulation, VMT and Orcem would offset any remaining operational emissions through purchase of credits in a BAAQMD-certified emission bank program for each criteria pollutant. Purchase of these offsets would reduce impacts, but GHG emissions would remain significant and unavoidable because the City's adopted Climate Action Plan does not cover marine and rail operations, and there is no assurance that emissions would be reduced to a level that would ensure the project would be consistent with the overarching objective of the City's Climate Action Plan to achieve the reduction targets as established for 2020 and 2035, or the state's target reduction goals in 2030 and 2050. Overall impacts related to GHG under the ROA would be reduced compared to the proposed project, but would still result in significant and unavoidable impacts to GHG during operations.

Hazards and Hazardous Materials

The ROA would involve similar construction activities as the proposed project and would therefore result in similar impacts related to hazards and hazardous materials during construction (reduced through mitigation).

The ROA would involve similar operational activities as the proposed project. Under the ROA, potential operational impacts related to hazards and hazardous materials would be similar to the proposed project.

Hydrology and Water Quality

The ROA, would involve similar construction activities as the proposed project and would therefore result in similar impacts to site drainage and hydrology. Potential impacts on marine water quality from material dredging, removal of creosote pilings, reuse of materials from on-site demolition activities, and use of Class II aggregate for riprap would be similar to the proposed project with similar potential for impacts (reduced through mitigation). For these reasons, hydrology and water quality impacts from the ROA would be similar to hydrology and water quality impacts from the proposed project.

Land Use and Planning

The ROA would involve similar land uses as the proposed project. The use of the site by VMT and Orcem is consistent with the City's existing General Plan and zoning designations, and most of the applicable land use plans, policies and regulations. However, similar to the proposed project, the ROA conflicts with the City's Public Access Policies. The City's policies rely on San Francisco Bay Conservation and Development Commission (BCDC) Design Guidelines and the proposed project and ROA are potentially significant. Therefore the City has proposed a mitigation measure

which the applicant has accepted to mitigate the potentially significant impacts to provide in-lieu access via the provision of monetary assistance to close the funding gap for the design phase of the Bay/Vine Trail project. This would reduce this impact to less-than-significant levels. Consistency of the ROA with regard to BCDC policies would be similar to land use impacts from the proposed project, including a potentially significant impact related to inconsistency with BCDC plans and policies that will be resolved during the permit issuance process.

Noise

With the exception of additional landscaping screening (trees and shrubbery) along the west and south sides of the Material Storage Area within the Orcem Site, the ROA would involve identical physical changes to the project site as the proposed project, including demolition of the existing structures on the site and construction of identical facilities for VMT and Orcem. Therefore, the ROA would have similar construction-related noise impacts.

The noise associated with operations of the project is primarily due to the transport of materials in trucks and by railcar to and from the site. The noise associated with transfer of incoming material by ship to export via barge would occur within the facility; whereas noise from truck operations and rail activity would affect surrounding neighborhoods. Under the ROA, the number of Orcem daily trucks trips would be reduced, shipping of finished Orcem product would occur over six days instead of five, the length of trains would be reduced from 77 cars to 50, operation of barges over trucks/rail for VMT would be prioritized, late night operations at the Orcem site would be eliminated, and limits on VMT trucking would be enforced. These changes would not result in a noticeable effect on the 24-hour average noise of the project (day-night average sound level (L_{dn}) or community noise equivalent level (CNEL). If VMT is able to incentivize an increase in the use of barges, total truck trips would be reduced, and the related noise impacts would be reduced. Because there is not a guarantee that VMT would be able to increase the use of barges, the noise impacts would be the same or similar to the proposed project. However, the ROA also would reduce late night operations at the Orcem site, so in the vicinity of that project component, noise impacts are anticipated to be reduced.

Public Services and Recreation

With the exception of additional landscape screening, the ROA would involve identical physical changes to the project site as the proposed project. Demands for police, fire, and recreation services and facilities would similar to the proposed project.

The ROA could result in an increase in ship traffic and the number of days vessels would be docked at the VMT Terminal due to the shift away from trucks and trains for material transport. However, the presence of additional ships and trucks on additional days would not be considered a significant impact to pubic services and recreation since ships, boats, and trucks are frequent users of the waters and roadways that surround the project site and are not anticipated to result in a measureable change in demand for police, fire, and recreation services and facilities compared to the proposed project. For these reasons, public services and recreation impacts from the ROA would be similar to public services and recreation impacts from the proposed project.

Transportation and Traffic

With the exception of shorter train lengths (50 cars versus 77 cars) and additional landscape screening, the ROA would involve identical physical changes to the project site as the proposed project, including demolition of the existing structures on the site and construction of identical facilities for VMT and Orcem. Construction-period impacts including increased traffic from trucks and construction workers (reducing roadway capacity, worsening LOS, longer delays at study intersections), temporary closures of sidewalks, prohibition of on-street parking, and impacts to bus stops would all be the same or similar to the proposed project (reduced through mitigation).

The ROA would involve similar operational activities as the proposed project. Under the ROA, ship, rail car, truck, and heavy equipment operations within the area of the existing developed site and developed off-site areas would occur more efficiently, spread out on a monthly basis, six days per week instead of five days per week, with fewer truck trips per day (up to 442 trucks one way). The ROA would reduce the degree of impact on peak hour intersection operations from project truck traffic, but would not eliminate significant impacts because no peak hour intersection operation impacts are identified for the proposed project. Mitigations required under the proposed project would be required under the ROA as well. The reduced daily truck traffic associated with the ROA would reduce the degree of impact on Lemon Street related to the need for safe and efficient vehicle operations (Impact 3.12-4) and safe and convenient vehicle, pedestrian and bicycle movements (Impact 3.12-6), but not to a less than significant level based on the reduction in daily trucks: from 552 daily one-way truck trips (174 for VMT and 378 for Orcem) to 442 daily one-way truck trips (for 174 for VMT and 268 for Orcem). Thus, impacts would all be the same or similar to the proposed project (reduced through mitigation).

Under the ROA, the length of trains would be reduced, from 77 rail cars to 50 rail cars. This would reduce the potential for delays or queues at rail crossings and on emergency access compared to the proposed project (Impacts 3.12-2, 3 and 5), but not to a less than significant level, based on an analysis of the shorter "gate down" time with the ROA (refer to Appendix L). California Northern Railroad provided updated "gate down" times based on the new train lengths, ranging from 4.06 minutes (with one engine) and 4.16 minutes (with two engines). The analysis included an update with these shorter gate-down times. The new analysis found that, at most of the grade crossings in Vallejo, the 50-car trains would result in blockages of upstream intersections. In addition, delays would continue to be longer than 1 minute, which is part of significance criteria A.4 established in the EIR. These impacts would remain significant and unavoidable under the ROA, as the City cannot ensure that the California Northern Railroad will agree to modified hours of operation and

similar blockages may occur during non-peak hours because traffic levels remain at or above 70% of the peak hour traffic volumes.

Utilities and Service Systems

With the exception of additional landscape screening, the ROA would involve identical physical changes to the project site as the proposed project. Demands for utilities and service systems would be similar to the proposed project. Although there would be a need for minor increase in need for irrigation to water the additional landscaping, this demand is anticipated to be negligible.

The ROA could result in an increase in ship traffic and the number of days vessels would be docked at the VMT Terminal due to the shift away from trucks and trains for material transport. However, the presence of additional ships and trucks on additional days would not be considered a significant impact to utilities and service systems since ships, boats, and trucks are frequent users of the waters and roadways that surround the project site and are not anticipated to result in a measureable change in demand for utilities and service systems compared to the proposed project. For these reasons, utilities and service systems impacts from the ROA would be similar to utilities and service system impacts from the proposed project.

6.5 SUMMARY MATRIX

A matrix displaying the major characteristics and significant environmental effects of each alternative considered is provided in Table 6-12 to summarize the comparison with the proposed project. The matrix also indicates whether the alternative meets the proposed project objectives as defined in Chapter 2.0, Project Description.

Environmental Issue	Proposed Project Impacts Prior to Mitigation	Proposed Project Impacts with Mitigation	No Project Alternative	Revised Operations Alternative
Aesthetics	S	LTS	▼	—
Air Quality	S	SU	▼	•
Biological Resources	S	LTS	▼	—
Cultural Resources	S	SU	▼	—
Geology and Soils	S	LTS	▼	—
Greenhouse Gas Emissions	S	SU	▼	—
Hazards and Hazardous Materials	S	LTS	•	-
Hydrology and Water Quality	S	LTS	▼	—
Land Use and Planning	LTS	LTS	▼	_
Noise	S	SU	\checkmark	_

Table 6-12Summary of Impacts from Alternatives

Table 6-12				
Summary of Impacts from Alternatives				

Environmental Issue	Proposed Project Impacts Prior to Mitigation	Proposed Project Impacts with Mitigation	No Project Alternative	Revised Operations Alternative
Public Services and Recreation	LTS	LTS	—	—
Transportation and Traffic	S	SU	▼	▼
Utilities and Service Systems	LTS	LTS	▼	_
Meets Most Project Objectives?	Yes	Yes	No	Yes

Alternative is likely to result in greater impacts to issue when compared to proposed project.

- Alternative is likely to result in similar impacts to issue when compared to proposed project.

▼ Alternative is likely to result in reduced impacts to issue when compared to proposed project.

LTS = Less-than-significant impact.

S = Significant impact.

SU = Significant and unavoidable impact.

6.6 ENVIRONMENTALLY SUPERIOR ALTERNATIVE

As indicated in Table 6-12, the No Project Alternative would result in the least environmental impacts and would be the environmentally superior alternative. However, Section 15126.6(e)(2) of the CEQA Guidelines states that if the environmentally superior alternative is the No Project Alternative, the EIR shall also identify an environmentally superior alternative among the other alternatives considered. In this case, the environmentally superior alternative is the ROA, since it would reduce some aesthetics, air quality, noise, and traffic impacts. The ROA would also meet all of the proposed project objectives. However, impacts that are significant and unavoidable under the proposed project (cultural resources, GHG, and traffic) would remain significant and unavoidable under the ROA.

INTENTIONALLY LEFT BLANK



INTENTIONALLY LEFT BLANK



VALLEJO MARINE TERMINAL AND ORCEM PROJECT

INTENTIONALLY LEFT BLANK

EXECUTIVE SUMMARY

SCAQMD (South Coast Air Quality Management District). 2007. Particulate Matter (PM)Emission Factors For Processes/Equipment at Asphalt, Cement, Concrete and Aggregate Product Plants. June 2007.

CHAPTER 1 INTRODUCTION

- 14 CCR 15000–15387 and Appendix A–L. Guidelines for Implementation of the California Environmental Quality Act, as amended.
- California Public Resources Code, Sections 21000–21177. California Environmental Quality Act (CEQA), as amended.

CHAPTER 2 PROJECT DESCRIPTION

- City of Vallejo. 1999. *Vallejo General Plan*. July 1999. http://www.ci.vallejo.ca.us/common/pages/DisplayFile.aspx?itemId=31253.
- City of Vallejo. 2017. Vallejo General Plan 2040. Adopted August 29, 2017. Accessed December 12, 2018. http://www.ci.vallejo.ca.us/cms/one.aspx?pageId=25644.
- City of Vallejo. 2015. Vallejo, California Municipal Code, as amended. March 10, 2015. http://www.ci.vallejo.ca.us/cms/one.aspx?portalId=13506&pageId=22432.
- County of Solano. 2008. Solano County General Plan. November 2008. Accessed July 9, 2015. http://www.solanocounty.com/depts/rm/planning/general_plan.asp.

County of Solano. 2014. "Solano County Zoning Map." 1 inch:1,000 feet. Map No. 21-S.

CHAPTER 3 ENVIRONMENTAL ANALYSIS

14 CCR 15000–15387 and Appendix A–L. Guidelines for Implementation of the California Environmental Quality Act, as amended.

3.1 Aesthetics

14 CCR 15000–15387 and Appendix A–L. Guidelines for Implementation of the California Environmental Quality Act, as amended.

- BCDC (San Francisco Bay Conservation and Development Commission). 2012. San Francisco Bay Plan. Adopted 1965, as amended through 2012.
- BCDC. 2005. Shoreline Spaces: Public Access Design Guidelines for the San Francisco Bay. April 2005.
- BCDC. 2014. "History of the San Francisco Bay Conservation and Development Commission." San Francisco Bay and Conservation Development Commission. Accessed May 8, 2014. http://www.bcdc.ca.gov/history.shtml.
- City of Vallejo. 1999. Vallejo General Plan. July 1999. http://www.ci.vallejo.ca.us/ common/pages/DisplayFile.aspx?itemId=31253.
- City of Vallejo. 2014. Municipal Code, as amended May 13, 2014. https://www.municode.com/ library/ca/vallejo/codes/code_of_ordinances?nodeId=VACAMUCO.
- <u>City of Vallejo. 2017. Vallejo General Plan 2040. Adopted August 29, 2017. Accessed</u> December 12, 2018. http://www.ci.vallejo.ca.us/cms/one.aspx?pageId=25644.

3.2 Air Quality

- BAAQMD (Bay Area Air Quality Management District). 2001. *Revised San Francisco Bay Area* 2001 Ozone Attainment Plan for the 1-Hour National Ozone Standard. Adopted October 24, 2001. http://www.baaqmd.gov/~/media/Files/Planning%20and%20Research/Plans/ 2001%20Ozone%20Attainment%20Plan/oap_2001.ashx?la=en.
- BAAQMD (Bay Area Air Quality Management District). 2009. California Environmental Quality Act Guidelines Update, Proposed Thresholds of Significance. December 7, 2009. Accessed July 2015. http://www.baaqmd.gov/~/media/files/planning-and-research/ceqa/ proposed-thresholds-of-significance-dec-7-09.pdf.
- BAAQMD. 2010a. *California Environmental Quality Act Air Quality Guidelines*. May 2010. Accessed July 2015. http://www.baaqmd.gov/~/media/files/planning-and-research/ ceqa/draft_baaqmd_ceqa_guidelines_may_2010_final.pdf?la=en.
- BAAQMD. 2010b. *Bay Area 2010 Clean Air Plan.* Adopted September 15, 2010. http://www.baaqmd.gov/~/media/files/planning-and-research/plans/2010-clean-airplan/cap-volume-i-appendices.pdf?la=en.
- BAAQMD. 2012. California Environmental Quality Act Air Quality Guidelines. Updated May 2012. http://www.baaqmd.gov/~/media/Files/Planning%20and%20Research/CEQA/ BAAQMD%20CEQA%20Guidelines_Final_May%202012.ashx?la=en.

BAAQMD. 2014. 2013 Air Monitoring Network Plan. July 1, 2014.

BAAQMD. 2017. California Environmental Quality Act Air Quality Guidelines.

CARB (California Air Resources Board). 2000. *Risk Reduction Plan to Reduce Particulate Matter Emissions from Diesel-Fueled Engines and Vehicles*. October 2000. http://www.arb.ca.gov/diesel/documents/rrpfinal.pdf.

CARB. 2005. Air Quality and Land Use Handbook: A Community Health Perspective. April 2005.

CARB. 2014. Truck and Bus Regulation – On-Road Heavy-Duty Diesel Vehicles (In-Use) Regulation. Amended April 2014. http://www.arb.ca.gov/regact/2014/ truckbus14/tbfrooal.pdf.

CARB. 2018. iADAM Air Quality Data Statistics 2015-2017.

- City of Vallejo. 2017a. *Vallejo General Plan 2040*. Adopted August 29, 2017. Accessed December 12, 2018. http://www.ci.vallejo.ca.us/cms/one.aspx?pageId=25644.
- <u>City of Vallejo. 2017b. "City of Vallejo General Plan 2040 Land Use Map." As amended</u> <u>November 7, 2017. Accessed December 12, 2018.</u> <u>http://www.ci.vallejo.ca.us/cms/one.aspx?pageId=25644.</u>
- Melitta, R. 2015. Personal correspondence between R. Melitta (Senior Project Manager, PG&E Environmental Remediation) and I. Khalsa (City of Vallejo). June 18, 2015.
- OEHHA (Office of Environmental Health Hazard Assessment). 2018. CalEnviroScreen 3.0. Updated June 2018. https://oehha.ca.gov/calenviroscreen/maps-data.
- SCAQMD (South Coast Air Quality Management District). 2007. Particulate Matter (PM)Emission Factors For Processes/Equipment at Asphalt, Cement, Concrete and Aggregate Product Plants. June 2007.

3.3 Biological Resources

- 14 CCR 15000–15387 and Appendix A–L. Guidelines for Implementation of the California Environmental Quality Act, as amended.
- 16 U.S.C. Ch. 31. Marine Mammal Protection Act.
- 16 U.S.C. 668–668d. Bald and Golden Eagle Protection Act (BGEPA), as amended.

16 U.S.C. 703–712. Migratory Bird Treaty Act, as amended.

- 16 U.S.C. 1221–1226. Estuary Protection Act.
- 16 U.S.C. 1451–1464. Coastal Management Zone Act of 1972.
- 16 U.S.C. 1801–1884. Magnuson–Stevens Fishery Conservation and Management Act, as amended.
- 16 U.S.C. 4710. National Invasive Species Act.
- 33 U.S.C. 401, 403, 407. Rivers and Harbors Appropriations Act of 1899.
- 65 FR 7764–7787. Final rule: "Designated Critical Habitat: Critical Habitat for 19 Evolutionary Significant Units of Salmon and Steelhead in Washington, Oregon, Idaho, and California." February 16, 2000.
- 66 FR 3853–3856. Executive Order 13186 of January 10, 2011: "Responsibilities of Federal Agencies to Protect Migratory Birds."
- 70 FR 52488–52585. Final rule: "Endangered and Threatened Species; Designation of Critical Habitat for Seven Evolutionarily Significant Units of Pacific Salmon and Steelhead in California." September 3, 2005.
- 74 FR 52300–52351. Final rule: "Endangered and Threatened Wildlife and Plants: Final Rulemaking to Designate Critical Habitat for the Threatened Southern District Population Segment of North American Green Sturgeon," April 7, 2007.
- AECOM. 2013. Native Fish Utilization and Residency Within Restored Tidal Marsh Habitat in North and South San Francisco Bay. Prepared by AECOM and ECORP. Accessed June 24, 2014. http://www. deltacouncil.ca.gov/get-document/6710.
- AFS (American Fisheries Society). 2007. "The Collapse of Pelagic Fishes in the Upper San Francisco Bay." *Fisheries* 32 (6):270–277.
- AMS (Applied Marine Sciences Inc.). 2009. Survey of Intertidal Habitat and Marine Biota at Treasure Island and Along the Western Shoreline of Yerba Buena Island. Report prepared for the Treasure Island Redevelopment Project, San Francisco, California, April 2009.
- Anderson, M. H., M. Berggren, D. Wilhelmsson, and M.C. Ohman. 2009. "Epibenthic Colonization of Concrete and Steel Pilings in a Cold-Temperate Embayment: A Filed Experiment." *Helgoland Marine Research* 63(3):249–260.

- ANSI (American National Standards Institute). 2014. Annual Report 2013–2014. Accessed October 28, 2016. https://share.ansi.org/shared%20documents/News%20and% 20Publications/Brochures/Annual%20Report%20Archive/ANSI_2013-14 Annual Report with Roster.pdf.
- Bartling, R. 2006. *Pacific Herring Status of the Fisheries Report.* Prepared for the California Department of Fish and Game. Accessed November 10, 2014. https://www.google.com/search?client=safari&rls=en&q=Pacific+Herring+%E2%80%93+Status+of+the+Fisherie s+Report&ie=UTF-8&oe=UTF-8#rls=en&q=pacific herring %e2%80%93 status of the fisheries report%2c.
- Bay Institute. 2007. Petition to the State of California Fish and Game Commission and Supporting Information for Listing the Delta Smelt (Hypomesus transpacificus) as an Endangered Species under the California Endangered Species Act. Prepared by The Bay Institute, Center for Biological Diversity, and The Natural Resources Defense Council. February 7, 2007.
- BCDC (San Francisco Bay Conservation and Development commission). 1968. San Francisco Bay Plan. Accessed January 5, 2015. http://www.bcdc.ca.gov/laws_plans/plans/sfbay_plan#2.
- BCDC. 2008. San Francisco Bay Plan. Accessed October 27, 2016. http://www.bcdc.ca.gov/ pdf/bayplan/bayplan.pdf.
- BCDC. 2012. San Francisco Waterfront Special Area Plan. April 1975 as amended through April 2012. http://www.bcdc.ca.gov/pdf/sfwsap/SFWSAP_Final_2012.pdf.California Fish and Game Code, Section 2050–2115.5. California Endangered Species Act.
- California Fish and Game Code, Sections 90–99.5, 105, 7050–7090, 8585–8589.7, 8842, and 9001.7. Marine Life Management Act.
- California Public Resources Code, 71200-71271. Marine Invasive Species Act.
- California Water Code, Section 13000–16104. Porter-Cologne Water Quality Control Act, as amended. Prepared by the State Water Resources Control Board, with additions and amendments (shown as tracked changes) effective January 1, 2011. Accessed January 17, 2011. http://www.swrcb.ca.gov/laws_regulations/.
- Caltrans (California Department of Transportation). 2009. *Technical Guidance for Assessment and Mitigation of the Hydroacoustic Effects of Pile Driving on Fish. Final Report.* Prepared for California Department of Transportation by ICF Jones & Stokes and Illingworth and Rodkin, Inc. February 2009. http://www.dot.ca.gov/hq/env/bio/ files/Guidance_Manual_2_09.pdf.

- CDFW (California Department of Fish and Wildlife). 2014a. California Natural Diversity Database (CNDDB). RareFind, Version 4.0 (Commercial Subscription). Sacramento, California: CDFW, Biogeographic Data Branch. Accessed October 15, 2014. http://www.dfg.ca.gov/biogeodata/cnddb/mapsanddata.asp.
- CDFW. 2014b. *Fish Species of Special Concern*. Accessed on June 24, 2014. https://www.dfg.ca.gov/wildlife/nongame/ssc/fish.html.
- City of Vallejo. 1988. City of Vallejo Municipal Code, Chapter 10.12 Trees.
- City of Vallejo. 2017a. *Vallejo General Plan 2040*. Adopted August 29, 2017. Accessed December 12, 2018. http://www.ci.vallejo.ca.us/cms/one.aspx?pageId=25644.
- <u>City of Vallejo. 2017b. "City of Vallejo General Plan 2040 Land Use Map." As amended</u> <u>November 7, 2017. Accessed December 12, 2018.</u> <u>http://www.ci.vallejo.ca.us/cms/one.aspx?pageId=25644.</u>
- Dalen, J. and G.M. Knutsen. 1986. "Scaring Effects of Fish and Harmful Effects on Eggs, Larvae and Fry from Offshore Seismic Explorations." ICA Associated Symposium on Underwater Acoustics, Halifax, Canada, July 16–18, 1986.
- Ducks Unlimited. 2014. *Restoring Tidal Wetlands in American Canyon, CA*. Accessed November 4, 2014. http://www.ducks.org/california/california-bay-area-projects/restoring-tidal-wetlands-in-american-canyon-ca_
- EPA (U.S. Environmental Protection Agency). 2008. Creosote-Preliminary Risk Assessment for the Re-registration Eligibility Decision Document (RED). PC Codes 02203,025003, and 02004.
- Figley, B. 2003. Marine Life Colonization of Experimental Reef Habitat in Temperate Ocean Waters of New Jersey. Prepared for the New Jersey Department of Environmental Protection, Division of Fish and Wildlife and the Bureau of Marine Fisheries. January 2003.
- Fishbase.org. 2014. *Lampetra fluviatilis*, River lamprey. Accessed on November 3, 2014. http://www.fishbase.org/summary/Lampetra-fluviatilis.html
- Ganssle, D. 1966. Fishes and decapods of San Pablo and Suisun Bays, in Ecological studies of the Sacramento-Joaquin Estuary, Part I, compiled by D.W. Kelley. California Department of Fish Game Fish Bulletin 133: 64–94.
- Goals Project (San Francisco Bay Area Wetlands Ecosystem Goals Project). 2000. Baylands Ecosystem Species and Community Profiles: Life Histories and Environmental Requirements of Key Plants, Fish and Wildlife, Edited by P.R. Olofson. Oakland, California: San Francisco Bay Regional Water Quality Control Board.

- Garza, J.C. and E.D. Crandall. 2013. *Genetic Analysis of Chinook Salmon from the Napa River, California*. Fisheries Ecology Division, Southwest Fisheries Science Center, NOAA Fisheries and the Institute of Marine Sciences, University of California, Santa Cruz. http://naparcd.org/documents/NapaRiverChinookReport2013.pdf.
- Hewitt, C. and M. Campbell. 2010. *The Relative Contribution of Vectors to the Introduction and Translocation of Invasive Marine Species*. Final Report prepared for the Australian Department of Agriculture, Fisheries and Forestry.
- Hoover, J.J., K.J. Killgore, D.G. Clarke, H. Smith, A. Turnage, and J. Beard. 2005. "Paddlefish and Sturgeon Entrainment by Dredges: Swimming Performance as an Indicator of Risk," DOER Technical Notes Collection (ERDC TN-DOER-E22), U.S. Army Corps of Engineers Research and Development Center, Vicksburg, Mississippi. http://el.erdc.usace.army.mil/dots/doer/doer.html.
- IEP (Interagency Ecological Program for the San Francisco Bay Estuary). 2010–2012. San Francisco Bay Study, 2010–2012. Unpublished data.
- Kastak, D. and R.J. Schusterman. 1998. "Low-Frequency Amphibious Hearing in Pinnipeds: Methods, Measurements, Noise and Ecology." *Journal of the Acoustical Society of America* 103(4): 2216–2226.
- Kastelein, R.A, S. van der Heul, W.C. Verboom, R.J.V. Triesscheijn, and N.V. Jennings. 2006.
 "The Influence of Underwater Data Transmission Sounds on the Displacement Behaviour of Captive Harbour Seals (*Phoca Vitulina*)." *Marine Environmental Research* 61: 19–39
- Kimmerer, W.J. 2006. "Response of Anchovies Dampens Effects of the Invasive Bivalve Corbula amurensis on the San Francisco Estuary Food Web." *Marine Ecology Progress Series* 324:207–218.
- Kopec, D. and Harvey, J. 1995. Toxic Pollutants, Health Indices, and Population Dynamics of Harbor Seals in San Francisco Bay, 1989-91: A Final Report. Technical publication. Moss Landing, California: Moss Landing Marine Labs.
- Lassuy, D.R. 1989. Species Profiles: Life Histories and Environmental Requirements of Coastal Fishes and Invertebrates (Pacific Northwest)—Pacific Herring. USFWS Biological Report 82(11.126), U.S. Army Corps of Engineers, TR-EL-82-4.
- LSA. 2009. "Sacramento Splittail." In *Administrative Draft Solano HCP*. Solano County Water Agency Natural Community and Species Account. April 2009.

- LTMS Agencies. 2009. Background Information for Dredgers' Assessments of Potential Impacts on the Longfin Smelt in San Francisco Bay. Dated July 13, 2009. Accessed November 14, 2014.http://www.spn.usace.army.mil/Portals/68/docs/Dredging/guidance/Longfin%20and %20Dredging%20Impacts_091409.pdf
- LTMS. 1998. Long-Term Management Strategy (LTMS) for the Placement of Dredged Material in the San Francisco Bay Region, Final Policy Environmental Impact Statement/ Programmatic Environmental Impact Report, Volume I. August 1998. Accessed July 2015. http://www.spn. usace.army.mil/Missions/DredgingWorkPermits/LTMS/Volume1.aspx.
- LTMS. 2004. Environmental Work Windows, Informal Consultation Preparation Packet. April 2004. Prepared by the LTMS Environmental Windows Work Group. Accessed November 14, 2014. http://www.bcdc.ca.gov/pdf/Dredging/Informal_Consult_ Pckt.pdf. LTMS. 2013.
- LTMS. 2014. Long Term Management Strategy, LTMS Update, Draft Programmatic ESA Consultation with NOAA: Proposed Windows Modifications and Additional Measures for Salmonids and Green Sturgeon. Draft. April 7, 2014. Accessed July 2015. http://bayplanningcoalition.org/wp-content/uploads/2014/04/LTMS-update-for-NMFSconsultation-04-7-14.pdf.
- McAllister, D.E. 1963. "A Revision of the Smelt Family, Osmeridae." *National Museum of Canada Bulletin* 191.
- McEwan, D., and T.A. Jackson. 1996. *Steelhead Restoration and Management Plan for California*. California Department of Fish and Game, Inland Fisheries Division, Sacramento, California.
- McGowan, M.F., and M.N. Josselyn. 2008. *Green Sturgeon Critical Habitat Analysis*. Prepared for Bay Planning Coalition, San Francisco, California.
- Merkel & Associates. 2005. Eelgrass Community Pilot Study for San Francisco Bay: Techniques for Examining Invertebrates and Fish Assemblages within Multiple Eelgrass Beds.
 Document EA-012041. San Francisco-Oakland Bay Bridge East Span Seismic Safety Project. Prepared for the California Department of Transportation in cooperation with NOAA Marine Fisheries Service. October 2005.
- Merkel & Associates. 2010. San Francisco Bay Eelgrass Inventory; October-November 2009. Prepared for the California Department of Transportation and NOAA National Marine Fisheries Service. November 2010.

- Merz, J.E., Hamilton, S., Bergman, P.S., Cavallo, B. and T.A. Jackson. 2011. "Spatial Perspective for Delta Smelt: A Summary of Contemporary Survey Data." *California Fish* and Game 97(4): 164–189.
- Miller D.J. and R.N. Lea. 1972. "Guide to the Coastal Marine Fishes of California." California Department of Fish and Game. Fish Bulletin No. 157.
- Moyle, P.B., R.M. Yoshiyama, J.E. Williams and E.D. Wikramanayake. 1995. *Fish Species of Special Concern of California*.2nd ed. Prepared for the California Department of Fish and Game. Rancho Cordova, California: University of California, Davis, Department of Wildlife and Fisheries Biology.
- Moyle, P.B. 2002. *Inland Fishes of California*. Revised and expanded. Berkeley, California: University of California Press.
- NCRCD (Napa Country Resource Conservation District). 2012. Napa River Steelhead and Salmon Monitoring Program, 2011-2012 Season. September 2012.
- NOAA Fisheries (National Oceanic and Atmospheric Administration National Marine Fisheries). 2007. "Report on the Subtidal Habitats and Associated Biological Taxa in San Francisco Bay." June 2007.
- NOAA Fisheries. 2009a. "Harbor Porpoise (*Phocoena phocoena*): San Francisco-Russian River Stock." Accessed October 24, 2014. http://www.nmfs.noaa.gov/pr/sars/2013/ po2013_harborporpoise-sfrussianriver.pdf
- NOAA Fisheries. 2009b. "Designation of Critical Habitat for the Threatened Southern District Population Segment of Green Sturgeon, Final ESA Section 4(b)(2) Report." October 2009.
- NOAA Fisheries. 2013. Draft Guidance for Assessing the Effects of Anthropogenic Sound on Marine Mammals; Acoustic Threshold Levels for Onset of Permanent and Temporary Threshold Shifts. December 23, 2013. Accessed July 2015. http://www.nmfs.noaa.gov/pr/ acoustics/draft_acoustic_guidance_2013.pdf.
- NOAA Fisheries. 2014. "Sea Lion Diet." Accessed November 10, 2014. https://swfsc.noaa.gov/ textblock.aspx?Division=PRD&ParentMenuId=148&id=1252.
- NOAA Fisheries. 2015. "Endangered and Threatened Marine Species under NMFS' Jurisdiction." Updated April 27, 2015. Accessed July 2015. http://www.nmfs.noaa.gov/ pr/species/esa/listed.htm.

- NOAA Fisheries. 2016. Technical Guidance for Assessing the Effects of Anthropogenic Sound on <u>Marine Mammal Hearing: Underwater Acoustic Thresholds for Onset of Permanent and</u> <u>Temporary Threshold Shifts. NOAA Technical Memorandum NMFS-OPR-55. July 2016.</u>
- Pauley, G.B., D.A. Armstrong, R. Van Citter, and G.L. Thomas. 1989. "Species Profiles: Life Histories and Environmental Requirements of Coastal Fishes and Invertebrates (Pacific Southwest) Dungeness Crab." U.S. Fish Wildlife Service Biological Report 82(11.121).
- Ray, G. 2005. *Invasive Marine and Estuarine Animals of California*. ERDC/TC ANSRP-05-2. August 2005.
- Robinson, A. and B.K. Greenfield. 2011. LTMS Longfin Smelt Literature Review and Study Plan.
 Final Technical Report. Prepared for the Long Term Management Strategy for Dredged
 Materials in San Francisco Bay. August 2011. Oakland, California: San Francisco
 Estuary Institute.
- Roman, J. 2010. "Aquatic Invasive Species." The Encyclopedia of Earth. Accessed March 22, 2011. http://www.eoearth.org/article/Aquatic_invasive_species.
- SFBSHGP (San Francisco Bay Subtidal Habitat Goals Project). 2010. "San Francisco Bay Subtidal Habitat Goals Report – Conservation Planning for the Submerged Areas of the Bay; 50-Year Conservation Plan." Accessed October 24, 2014. http://www.sfbaysubtidal.org/report.html.
- San Luis and Delta Mendota Water Authority and C.H. Hanson. 1996. *Georgina Slough Acoustic Barrier Applied Research Project: Results of 1994 Phase II Field Tests*. Interagency Ecological Program for the San Francisco Bay/Delta estuary. Technical Report 44. May 1996
- Sommer, T., and F. Mejia. 2013. A Place to Call Home: A Synthesis of Delta Smelt Habitat in the Upper San Francisco Estuary. San Francisco Estuary and Watershed Science. June 2013. http://www.escholarship.org/uc/item/32c8t244.
- Stadler, J.H., K.F. Griffin, E.J. Chavez, B.C. Soence, P. Roni, C.A. Gavette, B.A. Seekins, and A.K. Obaza. 2011. Pacific Coast Salmon 5-Year Review of Essential Fish Habitat Final Report to the Pacific Fishery Management Council. May 2, 2011.
- Stratus Consulting. 2006. Creosote-Treated Wood in Aquatic Environments: Technical Review and Use Recommendations. Prepared for NOAA Fisheries, Southwest Division, Habitat Conservation Division. Santa Rosa, California: Stratus Consulting. December 31, 2006.
- Tasto, R.N. 1979. "San Francisco Bay: Critical to the Dungeness Crab?" In San Francisco Bay: The Urbanized Estuary, edited by T.J. Conomos, 479–490. San Francisco, California: Pacific Division of the American Association for the Advancement of Science.

- Thompson, J.K. and F. Parchaso. 2003. "The Immigration of an Asian Bivalve Potamocorbula in San Francisco Bay and the Subsequent Environmental Change." Accessed January 5, 2015. Menlo Park, California: U.S. Geological Survey. http://www.kisuiiki.jp/ old/symposium/intsem2/Intnl_Presen(E)/8.Thompson_E_.pdf.
- Thompson, J.K., J.R. Koseff, S.G. Monismith, and L.V. Lucas. 2008. "Shallow Water Processes Govern System-wide Phytoplankton Bloom Dynamics: A Field Study." *Journal of Marine Systems* 74 (2008) 153–166.
- Todorov, K. 2007. "Porpoises Playing in Napa River." *Napa Valley Register*. August 3, 2007. Accessed October 24, 2014. http://napavalleyregister.com/news/local/porpoises-playingin-napa-river/article_9e95d523-26bf-5a37-9d33-182fb5e97d93.html.
- Torok, M.L. 1994. "Movements, Daily Activity Patterns, Dive Behavior, and Food Habits of Harbor Seals (*Phoca Vitulina Richardsi*) in San Francisco Bay, California." Master's thesis; California State University, Stanislaus and Moss Landing Marine Laboratories
- TRAC (Washington State Transportation Commission). 2001. Executive Summary Overwater Structures: Marine Issues. Research Project 1803, Task 35, Overwater Whitepaper. Prepared by B. Nightingale and C.A. Simenstad (University of Washington, Seattle, Washington). Prepared for the TRAC.
- UCSD (University of California, San Diego). 2014. "River Lamprey *Lampetra ayresi* (Gunther)." Accessed October 31, 2014. http://www.sdsc.edu/~sekar/dijlibdemo/ sdsc_sp-river-lamprey.html.
- USACE and EPA (U.S. Army Corps of Engineers and U.S. Environmental Protection Agency). 2009. Programmatic Essential Fish Habitat (EFH) Assessment for the Long Term Management Strategy for the Placement of Dredged Material in the San Francisco Bay Region. July 2009.
- USFWS (U.S. Fish and Wildlife Service). 1989. The Ecology of the Sacramento-San Joaquin Delta: A Community Profile. Biological Report 85(7.22). September 1989.
- USFWS. 2013a. "Anadromous Fish." Updated February 15, 2013. Accessed on June 25, 2014. http://www.fws.gov/stockton/afrp/definitions.cfm?code=2.
- USFWS. 2013b. "Longfin Smelt 12-Month Finding." July 2, 2013. Accessed on October 27, 2014. Bay Delta Fish & Wildlife Office. http://www.fws.gov/sfbaydelta/species/longfin_smelt.cfm.

- USFWS. 2015. "Listed Species Believed to or Known to Occur in California." USFWS Environmental Conservation Online System, Accessed July 2, 2015. http://ecos.fws.gov/ tess_public/reports/species-listed-by-state-report?state=CA&status=listed.
- Vulcan Iron Works Inc. 2014. "Vulcanhammer.info Guide to Pile Driving Equipment." Accessed November 13, 2014. http://www.vulcanhammer.info/.
- Vines, C.A., T. Robbins, F.J. Griffin, and G.N. Cherr, 2000. "The Effects of Creosote derived compounds on the development in Pacific herring *(Clupea pallasii)*." *Aquatic Toxicology* 51: 225–239.
- Weise, M.J. and J.T. Harvey. 1999. Food Habits of California Sea Lions (Zalophus californianus) and Their Impact on Salmonid Fisheries in Monterey Bay, California.
 Report submitted to the Fisherman's Alliance of California, Monterey Chapter. Prepared by Moss Landing Marine Laboratories. MLML Publication No. 99-01. February 3, 1999.
- Zeiner, D.C., W.F. Laudenslayer Jr. K.E. Mayer, and M. White. eds. 1990. California's Wildlife, Vol. 3, Mammals. California State Wildlife Habitat Relationships System. Sacramento, California: State of California Department of Fish and Game. November 1990.

3.4 Cultural Resources

- 14 CCR 15000–15387 and Appendix A–L. Guidelines for Implementation of the California Environmental Quality Act, as amended.
- 16 U.S.C. 470-470x-6. National Historic Preservation Act, of 1966, as amended.
- California Public Resources Code, Sections 5097–5097.6. Archaeological, Paleontological, and Historical Sites.
- California Public Resources Code, Sections 21000–21177. California Environmental Quality Act (CEQA), as amended.
- City of Vallejo. 19992017. Vallejo General Plan 2040. July 1999Adopted August 2017. http://www.ci.vallejo.ca.us/common/ pages/DisplayFile.aspx?itemId=31253 http://www.ci.vallejo.ca.us/cms/One.aspx?pageId=25644.
- CSLC (California State Lands Commission). 2014. *Shipwreck Database*. Accessed October 22, 2014. http://shipwrecks.slc.ca.gov/ShipwrecksDatabase/Shipwrecks_Database.asp ?frmIncrement=&frmStartRow=0&frmQuery=solono&holdSearchValue=ShipName&fr mCounty=x&frmOrderBy=%5Bship%27s+name%5D&frmOrderDirection=ASC.

3.5 Geology and Soils

- 14 CCR 15000–15387 and Appendix A–L. Guidelines for Implementation of the California Environmental Quality Act, as amended.
- Bartugno, E.J. 1986. "Landslide Hazards in the Benicia-Vallejo Area, Solano County, California" [map]. 1:24,000. Landslide Hazard Identification Map No. 8. Open-File Report 86-17.
- California Department of Conservation. 2014a. "California's Alquist–Priolo Earthquake Fault Zones, Regulatory Maps Index." Accessed October 21, 2014. http://www.quake.ca.gov/ gmaps/WH/regulatorymaps.htm.
- California Department of Conservation. 2014b. "PSHA Ground Motion Interpolator: Longitude 122.2454, Latitude 38.0802." Accessed October 21, 2014. http://www.quake.ca.gov/gmaps/PSHA/psha_interpolator.html.
- CGS (California Geological Survey). 2002. "California's Geomorphic Provinces." Based on the Geomorphic Map of California prepared by O.P. Jennings, 1938; revisions of the text by D.L. Wagner, 2002. CGS Note 36. http://www.conservation.ca.gov/cgs/information/publications/cgs_notes/note_36/Documents/note_36.pdf.
- City of Vallejo. 2017a. *Vallejo General Plan 2040*. Adopted August 29, 2017. Accessed December 12, 2018. http://www.ci.vallejo.ca.us/cms/one.aspx?pageId=25644.
- <u>City of Vallejo. 2017b. "City of Vallejo General Plan 2040 Land Use Map." As amended</u> <u>November 7, 2017. Accessed December 12, 2018.</u> <u>http://www.ci.vallejo.ca.us/cms/one.aspx?pageId=25644.</u>
- Dibblee, T.W. and J.A. Minch. 2005. "Geologic Map of the Benicia Quadrangle, Contra Costa and Solano Counties, California" [map]. 1:24,000. Dibblee Foundation Map DF-146. Dibblee Geological Foundation.
- ERRG (Engineering/Remediation Resources Group Inc.). 2007. "Final Backfill Report, Former General Mills Facility, Vallejo, California." Letter report from D. Tang (ERRG) to T. Miller (Malcolm Pirnie Inc.). August 6, 2007.
- ICBO (International Conference of Building Official). 1994. Uniform Building Code.
- USDA (U.S. Department of Agriculture). 2014. *Custom Soil Resource Report for Solano County, California, Vallejo Marine Terminal and Orcem Project.* Accessed October 20, 2014. http://websoilsurvey.nrcs.usda.gov/.

- USGS (U.S. Geological Survey). 1974. "Preliminary Photointerpretation Map of Landslide and Other Surficial Deposits of the Mare Island and Carquinez Strait 15-minute Quadrangles Contra Costa, Marin, Napa, Solano, and Sonoma counties, California." USGS Miscellaneous Field Studies Map: 595.
- USGS. 2014. "Earthquake Summary Map, M6.0 South Napa, California Earthquake of 24 August 2014." Prepared in cooperation with the Global Seismographic Network. August 26, 2014.

3.6 Greenhouse Gas Emissions

- CARB (California Air Resources Board). 2006. Public Workshop to Discuss Establishing the 1990 Emissions Level and the California 2020 Limit and Developing Regulations to Require Reporting of Greenhouse Gas Emissions. Sacramento, California. December 1, 2006. http://www.arb.ca.gov/cc/inventory/meet/2006_12_01_presentation_intro.pdf.
- CARB. 2008. *Climate Change Proposed Scoping Plan: A Framework for Change*. October; approved December 12, 2008. http://www.arb.ca.gov/cc/scopingplan/document/psp.pdf.
- CARB. 2014a. First Update to the Climate Change Scoping Plan Building on the Framework Pursuant to AB 32 – The California Global Warming Solutions Act of 2006. May 2014. http://www.arb.ca.gov/cc/scopingplan/2013_update/first_update_climate_change_ scoping_plan.pdf.
- CARB. 2014b. California Greenhouse Gas Inventory for 2000–2012, by Category as Defined in the 2008 Scoping Plan. Last updated March 24, 2014. http://www.arb.ca.gov/cc/inventory/data/tables/ghg_inventory_scopingplan_00-12_2014-03-24.pdf.
- CAT (California Climate Action Team). 2006. *Final 2006 Climate Action Team Report to the Governor and Legislature*. Sacramento, California: CAT. March 2006.
- CAT. 2010a. *Climate Action Team Biennial Report*. Sacramento, California: CAT. April 2010. Accessed February 2014. http://www.energy.ca.gov/2010publications/ CAT-1000-2010-004/CAT-1000-2010-004.PDF.
- CAT. 2010b. Climate Action Team Report to Governor Schwarzenegger and the California Legislature. Sacramento, California: CAT. December 2010. Accessed February 2014. http://www.energy.ca.gov/2010publications/CAT-1000-2010-005/CAT-1000-2010-005.PDF.
- CEC (California Energy Commission). 2015. Existing Buildings Energy Efficiency Action Plan. Draft. March 2015. Accessed August 3, 2015. http://docketpublic.energy.ca.gov/ PublicDocuments/15-IEPR-05/TN203806_20150310T093903_California%E2%80%99s_ Existing_Buildings_Energy_Efficiency_Action_Plan.pdf.

- City of Vallejo. 2012. *City of Vallejo Climate Action Plan*. Final. Prepared by PMC. Oakland, California: PMC. March 2012. Accessed August 3, 2015. http://www.cityofvallejo.net/ common/pages/DisplayFile.aspx?itemId=30907.
- EPA (U.S. Environmental Protection Agency). 2014. Inventory of U.S. Greenhouse Gas Emissions and Sinks: 1990–2012. April 15, 2014. Accessed April 25, 2014. http://www.epa.gov/climatechange/ghgemissions/usinventoryreport.html.
- IPCC (Intergovernmental Panel on Climate Change). 2014. *Climate Change 2014: Impacts, Adaptation, and Vulnerability – Summary for Policymakers*. http://www.ipcc.ch/pdf/ assessment-report/ar5/wg2/ar5_wgII_spm_en.pdf.
- National Climatic Data Center. 2009. Global Warming Frequently Asked Questions. Asheville, N.C. http://lwf.ncdc.noaa.gov/oa/climate/globalwarming.html.
- NOAA (National Oceanic and Atmospheric Administration). 2015. Vertical Datum. Updated June 2015. http://www.ngs.noaa.gov/datums/vertical/.
- Pyle, T. 2008. "Industry Background and Overview." Presented by CAT Cement Sub-Group Leader T. Pyle at the CARB AB32 meeting, 2008.

3.7 Hazards and Hazardous Materials

- 14 CCR 15000–15387 and Appendix A–L. Guidelines for Implementation of the California Environmental Quality Act, as amended.
- City of Vallejo. 2017a. *Vallejo General Plan 2040*. Adopted August 29, 2017. Accessed December 12, 2018. http://www.ci.vallejo.ca.us/cms/one.aspx?pageId=25644.
- <u>City of Vallejo. 2017b. "City of Vallejo General Plan 2040 Land Use Map." As amended</u> <u>November 7, 2017. Accessed December 12, 2018.</u> <u>http://www.ci.vallejo.ca.us/cms/one.aspx?pageId=25644.</u>
- City of Vallejo and USACE (U.S. Army of Corps of Engineers). 2005. Final Environmental Impact Statement/Environmental Impact Report, Reuse of the Mare Island Dredged Material Disposal Ponds as a Confined Upland Dredged Material Disposal Facility, City of Vallejo, Solano County, California . Prepared by Jones and Stokes, State Clearinghouse Number 2003122039. October 2005.
- DTSC (California Department of Toxic Substances Control). 2007. "Hazardous Waste Transporter Requirements." Fact Sheet. August 2007. Accessed October 29, 2014. http://www.dtsc.ca.gov/HazardousWaste/Transporters/upload/Hazardous-Waste-Transporter-Requirements.pdf.

- EPA (U.S. Environmental Protection Agency). 2014a. "Summary of the Resource Conservation and Recovery Act." Accessed October 29, 2014. http://www2.epa.gov/lawsregulations/summary-resource-conservation-and-recovery-act.
- EPA. 2014b. "Summary of the Comprehensive Environmental Response, Compensation, and Liability Act (Superfund)." Accessed October 29, 2014. http://www2.epa.gov/lawsregulations/summary-comprehensive-environmental-response-compensation-andliability-act.
- HSC (Harbor Safety Committee of the San Francisco Bay Region). 2013. San Francisco, San Pablo and Suisun Bays Harbor Safety Plan. June 13, 2013.
- NIOSH (National Institute for Occupational Safety and Health). 2013. Criteria for a Recommended Standard, Occupational Exposure to Haxavalent Chromium. NIOSH Publication No. 2013-128. September 2013.
- NOAA (National Oceanic and Atmospheric Administration). 1992. "An Evaluation of the Extent and Magnitude of Biological Effects Associated with Chemical Contaminants in San Francisco Bay, California."
- Pinhey, M. 2014. Personal communication (email) from LTJG Matthew Pinhey (Vessel Traffic Service Supervisor). November 18, 2014.
- Solano County. 2014. "Hazardous Materials and Waste." County of Solano, Resource Management. Accessed October 29, 2014. http://www.solanocounty.com/depts/rm/ environmental_health/hazmat/default.asp.
- USCG (U.S. Coast Guard). 2014. Vessel Traffic Service San Francisco User's Manual. Revised July 28, 2014. Accessed October 14, 2014. http://www.uscg.mil/d11/vtssf/vtssfum.asp.
- WETA (Water Emergency Transportation Authority—the San Francisco Bay Region). 2009. Emergency Water Transportation System Management Plan. June 2009.

3.8 Hydrology and Water Quality

- 14 CCR 15000–15387 and Appendix A–L. Guidelines for Implementation of the California Environmental Quality Act, as amended.
- 33 U.S.C. 1251–1387. Federal Water Pollution Control Act, as amended (commonly referred to as the Clean Water Act).
- 40 CFR 411. Cement Manufacturing Point Source Category.

- ABAG (Association of Bay Area Governments). 2014. Flood Hazard Map for the Mare Island Strait. Accessed November 18, 2014. http://gis.abag.ca.gov/website/Hazards/ ?hlyr=femaZones.
- CalEMA (California Emergency Management Agency). 2009a. *Tsunami Inundation Map for Emergency Planning, Benicia Quadrangle*. Prepared in cooperation with the California Geological Survey and University of Southern California. July 15, 2009.
- CalEMA. 2009b. *Tsunami Inundation Map for Emergency Planning, Mare Island Quadrangle*. Prepared in cooperation with the California Geological Survey and University of Southern California. July 15, 2009.
- California Water Code, Section 13000–16104. Porter-Cologne Water Quality Control Act, as amended. Prepared by the State Water Resources Control Board, with additions and amendments (shown as tracked changes) effective January 1, 2011. Accessed January 17, 2011. http://www.swrcb.ca.gov/laws_regulations/.
- City of Vallejo. 2006. City of Vallejo 2005 Urban Water Management Plan. February 2006.
- City of Vallejo. 2014. Vallejo, California Municipal Code, as amended. May 13, 2014. http://www.ci.vallejo.ca.us/cms/one.aspx?portalId=13506&pageId=22432.
- Geosyntec. 2013. *Final Hydromodification Management Plan*. Prepared for the City of Vallejo. April 2013.
- NOAA (National Oceanic and Atmospheric Administration). 2015. *Tides & Currents: Tidal Datum*. Accessed July 13, 2015. http://tidesandcurrents.noaa.gov/datum_options.html.
- San Francisco Bay RWQCB (Regional Water Quality Control Board). 2011. San Francisco Bay Basin (Region 2) Water Quality Control Plan (Basin Plan). Incorporating all amendments approved by the Office of Administrative Law as of December 31, 2011.
- SWRCB (State Water Resources Control Board). 2014. "Industrial General Permit Notice of Intent, Waste Discharge Requirements for Discharges of Storm Water Associated with Industrial Activities."
- SFEI (San Francisco Estuary Institute) 2014. Contaminant Display and Download Tool. Spatial Queries for selected contaminants of concern. Accessed November 18, 2014. http://www.sfei.org/ tools/wqt.

3.9 Land Use and Planning

- ABAG and MTC (Association of Bay Area Governments and Metropolitan Transportation Commission). 2013. *Plan Bay Area: Strategy for A Sustainable Region*. Adopted July 18, 2013.
- BCDC (San Francisco Bay Conservation and Development Commission). 2012. *San Francisco Bay Plan.* Adopted 1965, as amended through 2012.
- BCDC. 2014. "History of the San Francisco Bay Conservation and Development Commission." Accessed May 8, 2014. http://www.bcdc.ca.gov/history.shtml.
- BCDC and MTC (San Francisco Bay Conservation and Development Commission and Metropolitan Transportation Commission). 2012. San Francisco Bay Area Seaport Plan. Adopted April 18, 1996, as amended through January 2012.
- City of Vallejo. 1999. *Vallejo General Plan*. July 1999. http://www.ci.vallejo.ca.us/common/pages/DisplayFile.aspx?itemId=31253.
- City of Vallejo. 2014. Vallejo, California Municipal Code, as amended. May 13, 2014. http://www.ci.vallejo.ca.us/cms/one.aspx?portalId=13506&pageId=22432.
- County of Solano. 2008. *Solano County General Plan*. Adopted August 5, 2008. http://www.solanocounty.com/depts/rm/planning/general plan.asp.
- County of Solano. 2014. Solano County Zoning Map. Map No. 21-S. Accessed May 8, 2014. http://www.co.solano.ca.us/civicax/filebank/blobdload.aspx?BlobID=17736.
- One Bay Area. 2014. "Plan Bay Area." Accessed May 8, 2014. http://onebayarea.org/ plan-bay-area.html.

3.10 Noise

- 14 CCR 15000–15387 and Appendix A–L. Guidelines for Implementation of the California Environmental Quality Act, as amended.
- Caltrans (California Department of Transportation). 2002. *Transportation-Related Earthborne Vibrations*. Report No. TAV-02-01-R9201. California Department of Transportation; Environmental Program; Environmental Engineering; Noise, Air Quality, and Hazardous Waste Management Office. February 20, 2002. http://www.dot.ca.gov/hq/env/noise/pub/ TRANSPORTATION%20RELATED%20EARTHBORNE%20VIBRATIONS.pdf.

- City of Vallejo. 1999. *Vallejo General Plan*. July 1999. http://www.ci.vallejo.ca.us/common/pages/DisplayFile.aspx?itemId=31253.
- City of Vallejo. 2012. *Vallejo General Plan Noise Element Update*. December 26, 2012. Accessed November 20, 2014. http://www.ci.vallejo.ca.us/workspaces/one.aspx? objectid=31257&contextId=31167.
- City of Vallejo. 2014. Vallejo, California Municipal Code, as amended. May 13, 2014. http://www.ci.vallejo.ca.us/cms/one.aspx?portalId=13506&pageId=22432.
- DOT (U.S. Department of Transportation). 1980. Noise Effects Handbook, A Desk Reference to Health and Welfare Effects of Noise, Office of Noise and Abatement Control. July 1981.
- EPA (U.S. Environmental Protection Agency). 1971. Noise from Construction Equipment and Operations, Building Equipment and Home Appliances. Prepared by Bolt, Beranek & Newman, Boston, Massachusetts. Washington, D.C.: EPA.
- FHWA (Federal Highway Administration). 2008. "Roadway Construction Noise Model RCNM." https://www.fhwa.dot.gov/environment/noise/construction_noise/rcnm/.
- FTA (Federal Transit Administration). 2006. Transit Noise & Vibration Impact Assessment. Federal Transit Administration, Office of Planning and Environment. May 2006.

3.11 Public Services and Recreation

- 14 CCR 15000–15387 and Appendix A–L. Guidelines for Implementation of the California Environmental Quality Act, as amended.
- City of Vallejo. 1999. *Vallejo General Plan*. July 1999. http://www.ci.vallejo.ca.us/common/pages/DisplayFile.aspx?itemId=31253.
- City of Vallejo. 2013a. "Vallejo Fire Divisions." Vallejo Fire Department. Accessed May 2, 2014. http://www.ci.vallejo.ca.us/city_hall/departments___divisions/fire/stations____divisions/divisions/.
- City of Vallejo. 2013b. "Fire Stations." Vallejo Fire Department. Accessed May 2, 2014. http://www.ci.vallejo.ca.us/city_hall/departments___divisions/fire/stations____divisions/divisions/.
- City of Vallejo. 2013c. "Police." Vallejo Police Department. Accessed May 2, 2014. http://www.ci.vallejo.ca.us/city_hall/departments___divisions/police/.

- <u>City of Vallejo. 2017. Vallejo General Plan 2040. Adopted August 29, 2017. Accessed</u> December 12, 2018. http://www.ci.vallejo.ca.us/cms/one.aspx?pageId=25644.
- GVRD (Greater Vallejo Recreation District). 2014. "About Us." Greater Vallejo Recreation District. Accessed May 6, 2014. http://www.gvrd.org/?page=About_GVRD.
- GVRD. 2006. *Greater Vallejo Recreation District Park and Recreation Master Plan*. Prepared by Moore Iacofano Goltsman Inc. September 2006. http://www.gvrd.org/files/ Revised_MP_Draft_9-28-06.pdf.
- O'Connell, J. 2014. Vallejo Police Department service capabilities. Email from J. O'Connell (Captain, VPD) to H. Martinelli (Dudek). October 16, 2014.
- Sproete, V. 2014. Vallejo Fire Department service capabilities. Telephone conversation between V. Sproete (VFD) and H. Martinelli (Dudek). October 16, 2014.

3.12 Transportation and Traffic

- 14 CCR 15000–15387 and Appendix A–L. Guidelines for Implementation of the California Environmental Quality Act, as amended.
- Caltrans (California Department of Transportation). 2002. "Guide for the Preparation of Traffic Impact Studies." December 2002. Accessed November 14, 2014. http://www.dot.ca.gov/hq/tpp/offices/ocp/igr ceqa files/tisguide.pdf.
- Caltrans. 2012. *Highway Design Manual*. May 7, 2012. http://www.dot.ca.gov/hq/oppd/hdm/pdf/english/HDM_Complete_01Jul2015.pdf.
- Caltrans. 2014. Freeway Count Data Provided by the District 4 Office of Highway Operations. October 28, 2014.
- CNRR (California Northern Railroad Company). October 2014. Phone conversation between E. Polling (Fehr & Peers) and J. Golian (CNRR).
- City of Vallejo. n.d. "Traffic Impact Analysis/Study Guidelines." City of Vallejo Public Works Department. Accessed November 14, 2014. http://www.ci.vallejo.ca.us/common/ pages/DisplayFile.aspx?itemId=40753.
- City of Vallejo. 1999. Vallejo General Plan. July 1999. http://www.ci.vallejo.ca.us/ common/pages/DisplayFile.aspx?itemId=31253.

- City of Vallejo. 2017a. *Vallejo General Plan 2040*. Adopted August 29, 2017. Accessed December 12, 2018. http://www.ci.vallejo.ca.us/cms/one.aspx?pageId=25644.
- <u>City of Vallejo. 2017b. "City of Vallejo General Plan 2040 Land Use Map." As amended</u> <u>November 7, 2017. Accessed December 12, 2018.</u> <u>http://www.ci.vallejo.ca.us/cms/one.aspx?pageId=25644.</u>
- County of Solano. 2013. Congestion Management Program._Adopted by the Solano Transportation Authority. December 11, 2013. Accessed November 14, 2014. http://www.sta.ca.gov/docManager/1000004382/FINAL%202013%20Solano%20CMP% 2012-11-13.pdf.
- Loewke, D. 2014. "Orcem and VMT truck and rail data confirmation spreadsheet." Email dated September 10, 2014; forwarded by Dudek on September 11, 2014.
- SolTrans (Solano County Transit). 2014. "Routes." Accessed October 20, 2014. http://www.soltransride.com.
- Transportation Research Board. 2010. *HCM2010 Highway Capacity Manual*. http://hcm.trb.org/?qr=1.

3.13 Utilities and Service Systems

- 14 CCR 15000–15387 and Appendix A–L. Guidelines for Implementation of the California Environmental Quality Act, as amended.
- CalRecycle. 2014. "Facility/Site Summary Details: Keller Canyon Landfill (07-AA-0032)." Accessed October 10, 2014. http://www.calrecycle.ca.gov/SWFacilities/Directory/07-AA-0032/Detail/.

City of Vallejo. 1999. Vallejo General Plan. July 1999. http://www.ci.vallejo.ca.us/common/ pages/DisplayFile.aspx?itemId=31253.

- City of Vallejo. 2006. *City of Vallejo 2005 Urban Water Management Plan*. February 28, 2006. http://www.scwa2.com/documents/uwmp/Vallejo%202005%20UWMP.pdf.
- <u>City of Vallejo. 2017. Vallejo General Plan 2040.</u> Adopted August 29, 2017. Accessed December 12, 2018. http://www.ci.vallejo.ca.us/cms/one.aspx?pageId=25644.
- PG&E (Pacific Gas & Electric). 2014. Preliminary Study New Primary Electric Service for Orcem California, Vallejo, CA. January 2014.

- PG&E. 2015. Will Serve Letter for Property Located at 780/790 Derr Street Vallejo, CA. January 16, 2015.
- RWQCB (Regional Water Quality Control Board, San Francisco Bay Region). 2015. San Francisco Bay Basin (Region 2) Water Quality Control Plan (Basin Plan). As amended through March 20, 2015. California Regional Water Quality Control Board, San Francisco Bay Region. http://www.waterboards.ca.gov/sanfranciscobay/water issues/ programs/planningtmdls/basinplan/web/docs/bp ch1withcover.pdf.
- RWQCB. 2012. Order No. R2-2012-0017 (NPDES No. CA0037699. California Regional Water Quality Control Board, San Francisco Bay Region. February 8, 2012.
- VSFCD (Vallejo Sanitation and Flood Control District). 2008. Sanitary Sewer Management Plan. December 12, 2008.

CHAPTER 5 OTHER CEQA CONSIDERATIONS

City of Vallejo. 2012. Economic Development Strategy Plan. Prepared by RDA Global Inc. September 11, 2012.

8301

Report Preparation

This Final EIR was prepared for the City of Vallejo by Dudek, 1630 San Pablo Avenue, Oakland, California, 94612. The following professionals participated in its preparation:

City of Vallejo

Andrea Ouse, Community & Economic Development Director Jim Reese, Contract Community & Economic Development Director Lisa Plowman, Contract Planner Michelle Hightower, Senior Planner Dan Marks, Special Advisor Afshan Hamid, Senior Planner

Dudek

Darcey Rosenblatt, Senior Project Manager Heather Ivey, AICP, Environmental Planner Kara Laurenson-Wright, Environmental Analyst Alejandro Goena, Environmental Analyst Candice Disney Magnus, Environmental Analyst Sara Orofino, Environmental Analyst Lora Granovsky, Air Quality Specialist Jennifer Longabaugh, AICP, Air Quality Analyst Christine Kronenberg, AICP, Environmental Planner Lisa Achter, Wildlife Biologist Adam Giacinto, M.A., RPA, Archaeologist Dylan Duvergé, Hydrogeologist Nicole Peacock, Environmental Engineer/Geologist Jonathan Leech, Acoustician Amna Javed, GIS Specialist Becky Golden-Harrell, Technical Editor Amy Seals, Technical Editor Devin Brookhart, Publications Specialist Lead Taylor Eaton, Publications Specialist Chelsea Ringenback, Publications Specialist
Applied Marine Sciences Inc.

Jay Johnson, Senior Oceanographer, Managing Principal Rebecca DeGabriele, Aquatic/Marine Biologist Doug George, Oceanographer

Carey and Company Inc.

Hisashi (Bill) Sugaya, Principal

Fehr and Peers

Ellen Poling, Senior Associate